UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended April 30, 2024

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to _____

Commission File Number: 001-04604

HEICO CORPORATION

(Exact name of registrant as specified in its charter)

Florida

(State or other jurisdiction of incorporation or organization)

65-0341002

(I.R.S. Employer Identification No.)

33021

(Zip Code)

3000 Taft Street, Hollywood, Florida

(Address of principal executive offices)

(954) 987-4000

(Registrant's telephone number, including area code)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, \$.01 par value per share	HEI	New York Stock Exchange
Class A Common Stock, \$.01 par value per share	HEI.A	New York Stock Exchange

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes \mathbb{Z} No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer \square Accelerated filer \square Non-accelerated filer \square Smaller reporting company \square Emerging growth company \square

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \Box No 🗷

The number of shares outstanding of each of the registrant's classes of common stock as of May 28, 2024 is as follows:

Common Stock, \$.01 par value	54,835,023 shares
Class A Common Stock, \$.01 par value	83,660,813 shares

HEICO CORPORATION

INDEX TO QUARTERLY REPORT ON FORM 10-Q

			Page
Part I.	Financi	al Information	
	Item 1.	Financial Statements:	
		Condensed Consolidated Balance Sheets (unaudited)	
		as of April 30, 2024 and October 31, 2023	2
		Condensed Consolidated Statements of Operations (unaudited)	
		for the six and three months ended April 30, 2024 and 2023	<u>3</u>
		Condensed Consolidated Statements of Comprehensive Income (unaudited)	
		for the six and three months ended April 30, 2024 and 2023	<u>4</u>
		Condensed Consolidated Statements of Shareholders' Equity (unaudited)	
		for the six and three months ended April 30, 2024 and 2023	<u>5</u>
		Condensed Consolidated Statements of Cash Flows (unaudited)	
		for the six months ended April 30, 2024 and 2023	<u>7</u>
		Notes to Condensed Consolidated Financial Statements (unaudited)	8
			_
	Item 2.	Management's Discussion and Analysis of Financial Condition and	
	100111 2.	Results of Operations	<u>24</u>
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk	37
			<u> </u>
	Item 4.	Controls and Procedures	37
	ποπi τ.	<u>Controls and Procedures</u>	<u>51</u>
Dowt II	Othan I	nformation	
1 art II.	other I		
	T . F		20
	Item 5.	Other Events	<u>38</u>
	Item 6.	Exhibits	<u>38</u>
Signatu	ires		<u>39</u>

PART I. FINANCIAL INFORMATION; Item 1. FINANCIAL STATEMENTS HEICO CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS - UNAUDITED (in thousands, except per share data)

Accounts receivable, net 5 Contract assets 1 Inventories, net 1,0 Prepaid expenses and other current assets 1,0 Prepaid expenses and other current assets 1,9 Property, plant and equipment, net 3 Goodwill 3,2 Intangible assets, net 1,3 Other assets \$7,3 Current liabilities: \$7,3 Short-term debt and current maturities of long-term debt 1 Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 1 Total current liabilities 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 3,5 Commitments and contingencies (Note 11) 5	04,161 04,362 10,158 88,101 61,919 68,701 26,740 85,468 33,337 51,599 65,845 \$4,382 94,875 53,874 6,032 59,163	\$171,048 509,075 111,702 1,013,680 49,837 1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101 8,547
Cash and cash equivalents\$2Accounts receivable, net5Contract assets1Inventories, net1,0Prepaid expenses and other current assets1,9Total current assets1,9Property, plant and equipment, net3Goodwill3,2Intangible assets, net1,3Other assets4Total assets\$7,3Current liabilities:\$7,3Short-term debt and current maturities of long-term debt1Trade accounts payable1Accrued expenses and other current liabilities3Income taxes payable1Total current liabilities5Long-term debt, net of current maturities5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3	04,362 10,158 88,101 61,919 68,701 26,740 85,468 33,337 51,599 65,845 \$4,382 94,875 53,874 6,032	509,075 111,702 1,013,680 49,837 1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Accounts receivable, net5Contract assets1Inventories, net1,0Prepaid expenses and other current assets1,0Property, plant and equipment, net3Goodwill3,2Intangible assets, net1,3Other assets4Total assets\$7,3LIABILITIES AND EQUITYCurrent liabilities:Short-term debt and current maturities of long-term debtTrade accounts payable1Accrued expenses and other current liabilities3Income taxes payable5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3	04,362 10,158 88,101 61,919 68,701 26,740 85,468 33,337 51,599 65,845 \$4,382 94,875 53,874 6,032	509,075 111,702 1,013,680 49,837 1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Contract assets1Inventories, net1,0Prepaid expenses and other current assets1,9Total current assets1,9Property, plant and equipment, net3Goodwill3,2Intangible assets, net1,3Other assets4Total assets\$7,3LIABILITIES AND EQUITYCurrent liabilities:Short-term debt and current maturities of long-term debtTrade accounts payable1Accrued expenses and other current liabilities3Income taxes payable5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3	10,158 88,101 61,919 68,701 26,740 85,468 33,337 51,599 65,845 \$4,382 94,875 53,874 6,032	111,702 1,013,680 49,837 1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Inventories, net1,0Prepaid expenses and other current assets1,9Total current assets1,9Property, plant and equipment, net3Goodwill3,2Intangible assets, net1,3Other assets4Total assets\$7,3LIABILITIES AND EQUITYCurrent liabilities:Short-term debt and current maturities of long-term debtTrade accounts payable1Accrued expenses and other current liabilities3Income taxes payable5Long-term debt, net of current maturities5Long-term debt, net of current maturities4Total liabilities4Total liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3	88,101 61,919 68,701 26,740 85,468 33,337 51,599 65,845 \$4,382 94,875 53,874 6,032	1,013,680 49,837 1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Prepaid expenses and other current assets 1,9 Total current assets 1,9 Property, plant and equipment, net 3 Goodwill 3,2 Intangible assets, net 1,3 Other assets 4 Total assets \$7,3 LIABILITIES AND EQUITY \$7,3 Current liabilities: \$7,3 Short-term debt and current maturities of long-term debt 1 Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 2,3 Deferred income taxes 1 Other long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	61,919 68,701 26,740 85,468 33,337 51,599 65,845 \$4,382 94,875 53,874 6,032	49,837 1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Total current assets1,9Property, plant and equipment, net3Goodwill3,2Intangible assets, net1,3Other assets4Total assets\$7,3LIABILITIES AND EQUITYCurrent liabilities:Short-term debt and current maturities of long-term debtTrade accounts payable1Accrued expenses and other current liabilities3Income taxes payable1Total current liabilities5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3	68,701 26,740 85,468 33,337 51,599 65,845 65,845 \$4,382 94,875 53,874 6,032	1,855,342 321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Property, plant and equipment, net 3 Goodwill 3,2 Intangible assets, net 1,3 Other assets 4 Total assets 4 Total assets \$7,3 LIABILITIES AND EQUITY Current liabilities: \$7,3 Short-term debt and current maturities of long-term debt Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 1 Total current liabilities 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	26,740 85,468 33,337 51,599 65,845 65,845 \$4,382 94,875 53,874 6,032	321,848 3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Goodwill3,2Intangible assets, net1,3Other assets4Total assets\$7,3LIABILITIES AND EQUITYCurrent liabilities: Short-term debt and current maturities of long-term debt Trade accounts payableAccrued expenses and other current liabilities3Income taxes payable1Total current liabilities5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3	\$4,382 94,875 53,874 6,032	3,274,327 1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Intangible assets, net 1,3 Other assets 4 Total assets \$7,3 LIABILITIES AND EQUITY Current liabilities: \$7,3 Short-term debt and current maturities of long-term debt 1 Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 1 Total current liabilities 5 Long-term debt, net of current maturities 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	33,337 51,599 65,845 \$4,382 94,875 53,874 6,032	1,357,281 386,265 \$7,195,063 \$17,801 205,893 433,101
Other assets 4 Total assets \$7,3 LIABILITIES AND EQUITY Current liabilities: Short-term debt and current maturities of long-term debt Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	51,599 65,845 \$4,382 94,875 53,874 6,032	386,265 \$7,195,063 \$17,801 205,893 433,101
Total assets\$7,3LIABILITIES AND EQUITYCurrent liabilities: Short-term debt and current maturities of long-term debt Trade accounts payable1Accrued expenses and other current liabilities3Income taxes payable1Total current liabilities5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:1	\$4,382 94,875 53,874 6,032	\$7,195,063 \$17,801 205,893 433,101
LIABILITIES AND EQUITY Current liabilities: Short-term debt and current maturities of long-term debt Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 1 Total current liabilities 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3	\$4,382 94,875 53,874 6,032	\$17,801 205,893 433,101
Current liabilities: Short-term debt and current maturities of long-term debt Trade accounts payable Accrued expenses and other current liabilities Income taxes payable Total current liabilities Total current liabilities Short-term debt, net of current maturities Deferred income taxes 1 Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) Redeemable noncontrolling interests (Note 3) Shareholders' equity:	94,875 53,874 6,032	205,893 433,101
Short-term debt and current maturities of long-term debt 1 Trade accounts payable 1 Accrued expenses and other current liabilities 3 Income taxes payable 2 Total current liabilities 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	94,875 53,874 6,032	205,893 433,101
Trade accounts payable1Accrued expenses and other current liabilities3Income taxes payable5Total current liabilities5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities4Total liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:1	94,875 53,874 6,032	205,893 433,101
Accrued expenses and other current liabilities 3 Income taxes payable	53,874 6,032	433,101
Income taxes payable 5 Total current liabilities 5 Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	6,032	
Total current liabilities5Long-term debt, net of current maturities2,3Deferred income taxes1Other long-term liabilities4Total liabilities3,5Commitments and contingencies (Note 11)3Redeemable noncontrolling interests (Note 3)3Shareholders' equity:3		8,547
Long-term debt, net of current maturities 2,3 Deferred income taxes 1 Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	59 163	
Deferred income taxes 1 Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	59,105	665,342
Other long-term liabilities 4 Total liabilities 3,5 Commitments and contingencies (Note 11) 7 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	85,267	2,460,277
Total liabilities 3,5 Commitments and contingencies (Note 11) 3 Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	19,987	131,846
Commitments and contingencies (Note 11) Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity:	90,253	379,640
Redeemable noncontrolling interests (Note 3) 3 Shareholders' equity: 3	54,670	3,637,105
Shareholders' equity:		
	68,369	364,807
none issued		
Common Stock, \$.01 par value per share; 150,000 shares authorized; 54,828 and 54,721 shares issued and outstanding	548	547
Class A Common Stock, \$.01 par value per share; 150,000 shares authorized; 83,644 and 83,507 shares issued and outstanding	836	835
	98,699	578,809
Deferred compensation obligation	6,318	6,318
HEICO stock held by irrevocable trust	(6,318)	(6,318)
•	35,677)	(40,180)
-		2,605,984
	25,021	3,145,995
	25,021 89,427	47,156
		3,193,151
Total liabilities and equity \$7,3	89,427	

HEICO CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS – UNAUDITED (in thousands, except per share data)

	Six months ended April 30,		Three months en	ded April 30,	
	2024	2023	2024	2023	
Net sales	\$1,851,758	\$1,308,756	\$955,395	\$687,841	
Operating costs and expenses:					
Cost of sales	1,133,194	798,445	583,600	421,329	
Selling, general and administrative expenses	329,201	223,787	162,642	109,422	
Total operating costs and expenses	1,462,395	1,022,232	746,242	530,751	
Operating income	389,363	286,524	209,153	157,090	
Interest expense	(77,119)	(17,441)	(38,512)	(11,373)	
Other income	1,139	982	460	343	
Income before income taxes and noncontrolling interests	313,383	270,065	171,101	146,060	
Income tax expense	53,000	52,000	36,200	31,000	
Net income from consolidated operations	260,383	218,065	134,901	115,060	
Less: Net income attributable to noncontrolling interests	22,539	19,918	11,755	9,940	
Net income attributable to HEICO	\$237,844	\$198,147	\$123,146	\$105,120	
Net income per share attributable to HEICO shareholders:					
Basic	\$1.72	\$1.45	\$.89	\$.77	
Diluted	\$1.70	\$1.43	\$.88	\$.76	
Weighted average number of common shares outstanding:					
Basic	138,325	136,786	138,386	136,916	
Diluted	139,976	138,590	140,059	138,600	

HEICO CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME – UNAUDITED (in thousands)

	Six months en	ded April 30,	Three months ended April 30,			
	2024	2023	2024	2023		
Net income from consolidated operations	\$260,383	\$218,065	\$134,901	\$115,060		
Other comprehensive income (loss):						
Foreign currency translation adjustments	4,618	30,379	(10,143)	1,994		
Amortization of unrealized loss on defined benefit pension plan, net of tax	26	28	13	13		
Total other comprehensive income (loss)	4,644	30,407	(10,130)	2,007		
Comprehensive income from consolidated operations	265,027	248,472	124,771	117,067		
Net income attributable to noncontrolling interests	22,539	19,918	11,755	9,940		
Foreign currency translation adjustments attributable to noncontrolling interests	141	1,534	(415)	275		
Comprehensive income attributable to noncontrolling interests	22,680	21,452	11,340	10,215		
Comprehensive income attributable to HEICO	\$242,347	\$227,020	\$113,431	\$106,852		

HEICO CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY - UNAUDITED For the Six Months Ended April 30, 2024 and 2023 (in thousands, except per share data)

		HEICO Shareholders' Equity								
	Redeemable Noncontrolling Interests	Common Stock	Class A Common Stock	Capital in Excess of Par Value	Deferred Compensation Obligation	HEICO Stock Held by Irrevocable Trust	Accumulated Other Comprehensive Loss	Retained Earnings	Noncontrolling Interests	Total Shareholders' Equity
Balances as of October 31, 2023	\$364,807	\$547	\$835	\$578,809	\$6,318	(\$6,318)	(\$40,180)	\$2,605,984	\$47,156	\$3,193,151
Comprehensive income	15,999	_	—	_	—	—	4,503	237,844	6,681	249,028
Cash dividends (\$.10 per share)	—	_	—	_	—	—	—	(13,831)	—	(13,831)
Issuance of common stock to HEICO Savings and Investment Plan	_	_		9,300	_		_	_	_	9,300
Share-based compensation expense	_	_	_	9,463	_	_	_	_	_	9,463
Proceeds from stock option exercises	_	1	1	4,149	_	_	_	_	_	4,151
Redemptions of common stock related to stock option exercises	_	_	_	(2,352)	_	_	_	_	_	(2,352)
Distributions to noncontrolling interests	(14,967)	_	_	_	_	_	_	_	(458)	(458)
Acquisitions of noncontrolling interests	(3,165)	_	_	_	_	_	_	_	_	_
Adjustments to redemption amount of redeemable noncontrolling interests	4,608	_	_	_	_	_	_	(4,608)	_	(4,608)
Other	1,087			(670)				(368)		(1,038)
Balances as of April 30, 2024	\$368,369	\$548	\$836	\$598,699	\$6,318	(\$6,318)	(\$35,677)	\$2,825,021	\$53,379	\$3,442,806

		HEICO Shareholders' Equity								
	Redeemable Noncontrolling Interests	Common Stock	Class A Common Stock	Capital in Excess of Par Value	Deferred Compensation Obligation	HEICO Stock Held by Irrevocable Trust	Accumulated Other Comprehensive Loss	Retained Earnings	Noncontrolling Interests	Total Shareholders' Equity
Balances as of October 31, 2022	\$327,601	\$545	\$821	\$397,337	\$5,297	(\$5,297)	(\$46,499)	\$2,253,932	\$42,170	\$2,648,306
Comprehensive income	15,356	—	_	—	—	—	28,873	198,147	6,096	233,116
Cash dividends (\$.10 per share)	—	_	_	—	—	—	—	(13,668)	—	(13,668)
Issuance of common stock to HEICO Savings and Investment Plan		_	_	7,760			_		_	7,760
Share-based compensation expense	_	_	_	6,055	_	_	_	_	_	6,055
Proceeds from stock option exercises	—	2	2	4,070	_	—	_	_	—	4,074
Redemptions of common stock related to stock option exercises	_	_	_	(14,811)	_		_	_	_	(14,811)
Noncontrolling interests assumed related to acquisitions	14,642	_	_	_	_	_	_	—	_	_
Distributions to noncontrolling interests	(16,161)	—	_	_	_	_	_	_	(6,489)	(6,489)
Acquisitions of noncontrolling interests	(1,059)	—	—	(1,674)	_	_	_	—	—	(1,674)
Adjustments to redemption amount of redeemable noncontrolling interests	3,103	_	_	_	_	_	_	(3,103)	_	(3,103)
Deferred compensation obligation	—	—	_	_	874	(874)	_	_	_	_
Other	2,351			254				(153)		101
Balances as of April 30, 2023	\$345,833	\$547	\$823	\$398,991	\$6,171	(\$6,171)	(\$17,626)	\$2,435,155	\$41,777	\$2,859,667

HEICO CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY - UNAUDITED For the Three Months Ended April 30, 2024 and 2023 (in thousands, except per share data)

					• •	,				
	Redeemable Noncontrolling Interests	Common Stock	Class A Common Stock	Capital in Excess of Par Value	Deferred Compensation Obligation	HEICO Stock Held by Irrevocable Trust	Accumulated Other Comprehensive Loss	Retained Earnings	Noncontrolling Interests	Total Shareholders' Equity
Balances as of January 31, 2024	\$365,865	\$548	\$836	\$585,888	\$6,318	(\$6,318)	(\$25,962)	\$2,705,128	\$50,201	\$3,316,639
Comprehensive income	8,003	_	_	_	—	_	(9,715)	123,146	3,337	116,768
Issuance of common stock to HEICO Savings and Investment Plan	_	_	_	6,724	_	_	_	_	_	6,724
Share-based compensation expense	_	_	_	4,582	_	_	_	—	_	4,582
Proceeds from stock option exercises	_	_	_	1,897	_	—	_	_	—	1,897
Redemptions of common stock related to stock option exercises	_	_	_	(1,751)	_	_	_	_	_	(1,751)
Distributions to noncontrolling interests	(6,500)	_	_	_	_	_	_	_	(159)	(159)
Acquisitions of noncontrolling interests	(2,109)	—	_	1,156	_	_	_	—	_	1,156
Adjustments to redemption amount of redeemable noncontrolling interests	3,165	_	_	_	_	_	_	(3,165)	_	(3,165)
Other	(55)			203				(88)		115
Balances as of April 30, 2024	\$368,369	\$548	\$836	\$598,699	\$6,318	(\$6,318)	(\$35,677)	\$2,825,021	\$53,379	\$3,442,806

		HEICO Shareholders' Equity								
	Redeemable Noncontrolling Interests	Common Stock	Class A Common Stock	Capital in Excess of Par Value	Deferred Compensation Obligation	HEICO Stock Held by Irrevocable Trust	Accumulated Other Comprehensive Loss	Retained Earnings	Noncontrolling Interests	Total Shareholders' Equity
Balances as of January 31, 2023	\$340,287	\$547	\$822	\$388,603	\$6,171	(\$6,171)	(\$19,358)	\$2,328,523	\$45,037	\$2,744,174
Comprehensive income	7,376	—	—	_	—	—	1,732	105,120	2,839	109,691
Issuance of common stock to HEICO Savings and Investment Plan	_	_	_	5,796	_		_	_	_	5,796
Share-based compensation expense	_	_	_	3,243	_	_	_	_	_	3,243
Proceeds from stock option exercises	_	_	1	1,228	_	_	_	_	_	1,229
Redemptions of common stock related to stock option exercises	_	_	_	(6)	_	_	_	_	_	(6)
Noncontrolling interests assumed related to acquisitions	2,592	_	_	_	_	_	_	_	_	_
Distributions to noncontrolling interests	(5,260)	—	_	_	_	_	_	—	(6,099)	(6,099)
Adjustments to redemption amount of redeemable noncontrolling interests	(1,513)	_	_	_	_	_	_	1,513	_	1,513
Other	2,351			127				(1)		126
Balances as of April 30, 2023	\$345,833	\$547	\$823	\$398,991	\$6,171	(\$6,171)	(\$17,626)	\$2,435,155	\$41,777	\$2,859,667

HEICO CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS - UNAUDITED (in thousands)

-	Six months ende	
	2024	2023
Operating Activities:		
Net income from consolidated operations	\$260,383	\$218,065
Adjustments to reconcile net income from consolidated operations to net cash provided by operating activities:		
Depreciation and amortization	86,336	56,784
Share-based compensation expense	9,463	6,055
Employer contributions to HEICO Savings and Investment Plan	8,802	6,533
Deferred income tax benefit	(11,532)	(9,596
Amendment and termination of contingent consideration agreement		(9,057
Payment of contingent consideration	(6,203)	(6,299
(Decrease) increase in accrued contingent consideration, net	(5,326)	1,842
Changes in operating assets and liabilities, net of acquisitions:		
Decrease (increase) in accounts receivable	5,309	(21,222
Decrease (increase) in contract assets	3,172	(9,267
Increase in inventories	(71,103)	(75,251
(Increase) decrease in prepaid expenses and other current assets	(9,243)	1,738
(Decrease) increase in trade accounts payable	(11,406)	6,797
Decrease in accrued expenses and other current liabilities	(58,102)	(2,671
Decrease in income taxes payable	(6,830)	(13,824
Net changes in other long-term liabilities and assets related to	(0,050)	(15,021
HEICO Leadership Compensation Plan	17,319	10,563
Other	41,753	(6,754
Net cash provided by operating activities	252,792	154,436
	232,192	134,430
Investing Activities:		
Acquisitions, net of cash acquired	(46,208)	(524,231
Capital expenditures	(26,325)	(21,921
Investments related to HEICO Leadership Compensation Plan	(14,410)	(14,000
Other	1,657	362
Net cash used in investing activities	(85,286)	(559,790
Financing Activities:		
Payments on revolving credit facility	(125,000)	(108,000
Borrowings on revolving credit facility	50,000	556,000
Distributions to noncontrolling interests	(15,372)	(22,650
(Payments) borrowings on short-term debt, net	(13,924)	1,672
Cash dividends paid	(13,831)	(13,668
Payment of contingent consideration	(13,797)	(12,610
Acquisitions of noncontrolling interests	(3,165)	(2,733
Redemptions of common stock related to stock option exercises	(2,352)	(14,811
Proceeds from stock option exercises	4,151	4,074
Other	(1,905)	1,491
Net cash (used in) provided by financing activities		
Net cash (used in) provided by financing activities	(135,195)	388,765
Effect of exchange rate changes on cash	802	4,246
Net increase (decrease) in cash and cash equivalents	33,113	(12,343
Cash and cash equivalents at beginning of year	171,048	139,504
Cash and cash equivalents at end of period	\$204,161	\$127,161

HEICO CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS – UNAUDITED

1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Presentation

The accompanying unaudited condensed consolidated financial statements of HEICO Corporation and its subsidiaries (collectively, "HEICO," or the "Company") have been prepared in conformity with accounting principles generally accepted in the United States of America for interim financial information and in accordance with the instructions to Form 10-Q. Therefore, the condensed consolidated financial statements do not include all information and footnotes normally included in annual consolidated financial statements and should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the year ended October 31, 2023. The October 31, 2023 Condensed Consolidated Balance Sheet has been derived from the Company's audited consolidated financial statements. In the opinion of management, the unaudited condensed consolidated financial statements contain all adjustments (consisting principally of normal recurring accruals) necessary for a fair presentation of the condensed consolidated balance sheets, statements of operations, statements of comprehensive income, statements of shareholders' equity and statements of cash flows for such interim periods presented. The results of operations for the six months ended April 30, 2024 are not necessarily indicative of the results which may be expected for the entire fiscal year.

The Company has two operating segments: the Flight Support Group ("FSG"), consisting of HEICO Aerospace Holdings Corp. and HEICO Flight Support Corp. ("HFSC") and their respective subsidiaries; and the Electronic Technologies Group ("ETG"), consisting of HEICO Electronic Technologies Corp. and its subsidiaries.

New Accounting Pronouncements

In November 2023, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2023-07, "Segment Reporting (Topic 280): Improvements to Reportable Segment Disclosures," which expands reportable segment disclosure requirements by requiring disclosures of significant reportable segment expenses that are regularly provided to the Chief Operating Decision Maker ("CODM") and included within each reported measure of a segment's profit or loss. The ASU also requires disclosure of the title and position of the individual identified as the CODM and an explanation of how the CODM uses the reported measures of a segment's profit or loss in assessing segment performance and deciding how to allocate resources. Additionally, ASU 2023-07 requires all segment profit or loss and assets disclosures to be provided on an annual and interim basis. ASU 2023-07 is effective for fiscal years beginning after December 15, 2023, or in fiscal 2025 for HEICO, and interim periods within fiscal years beginning one year later. Early adoption is permitted and the amendments must be applied retrospectively to all prior periods presented. The adoption of this guidance will not affect the Company's consolidated results of operations, financial position or cash flows and the Company is currently evaluating the effect the guidance will have on its disclosures.

In December 2023, the FASB issued ASU 2023-09, "Income Taxes (Topic 740): Improvements to Income Tax Disclosures," which requires disclosure of specific categories in the annual effective tax rate reconciliation table and further disaggregation for reconciling items that meet a quantitative threshold. The ASU also requires the disaggregation of income taxes paid by jurisdiction. ASU 2023-09 may be applied either prospectively or retrospectively and is effective for fiscal years beginning after December 15, 2024, or in fiscal 2026 for HEICO. Early adoption is permitted. The adoption of this guidance will not affect the Company's consolidated results of operations, financial position or cash flows and the Company is currently evaluating the effect the guidance will have on its disclosures.

2. ACQUISITION

In December 2023, the Company, through a subsidiary of HFSC, entered into an exclusive license and acquired certain assets for the capability to support the Boeing 737NG/777 Cockpit Display and Legacy Displays product lines from Honeywell International. The transaction provides the HFSC subsidiary with the exclusive capability to produce, sell, and repair Boeing 737NG/777 Cockpit Displays as well as other Legacy Displays for Boeing 717, ATR, and select business and general aviation aircraft. The purchase price of this acquisition was paid in cash using proceeds from the Company's revolving credit facility and is not material or significant to the Company's condensed consolidated financial statements.

The allocation of the total consideration for this acquisition to the tangible and identifiable intangible assets acquired is preliminary until the Company obtains final information regarding their fair values. However, the Company does not expect any adjustment to such allocation to be material to the Company's consolidated financial statements. This acquisition's operating results were included in the Company's results of operations from the effective acquisition date. The amount of net sales and earnings of this acquisition included in the Condensed Consolidated Statements of Operations for the six and three months ended April 30, 2024 is not material. Had this acquisition occurred as of November 1, 2022, net sales, net income from consolidated operations, net income attributable to HEICO, and basic and diluted net income per share attributable to HEICO shareholders on a pro forma basis for the six and three months ended April 30, 2024 and 2023 would not have been materially different than the reported amounts.

3. SELECTED FINANCIAL STATEMENT INFORMATION

Accounts Receivable

(in thousands)	April 30, 2024	October 31, 2023
Accounts receivable	\$517,633	\$521,696
Less: Allowance for doubtful accounts	(13,271)	(12,621)
Accounts receivable, net	\$504,362	\$509,075

Inventories

(in thousands)	April 30, 2024	October 31, 2023
Finished products	\$659,590	\$622,395
Work in process	91,326	79,789
Materials, parts, assemblies and supplies	337,185	311,496
Inventories, net of valuation reserves	\$1,088,101	\$1,013,680

Property, Plant and Equipment

(in thousands)	April 30, 2024	October 31, 2023
Land	\$19,809	\$19,706
Buildings and improvements	207,253	202,499
Machinery, equipment and tooling	403,944	386,602
Construction in progress	30,518	25,867
	661,524	634,674
Less: Accumulated depreciation and amortization	(334,784)	(312,826)
Property, plant and equipment, net	\$326,740	\$321,848

Accrued Customer Rebates and Credits

The aggregate amount of accrued customer rebates and credits included within accrued expenses and other current liabilities in the accompanying Condensed Consolidated Balance Sheets was \$25.5 million as of April 30, 2024 and \$24.5 million as of October 31, 2023. The total customer rebates and credits deducted within net sales for the six months ended April 30, 2024 and 2023 was \$5.8 million and \$4.2 million, respectively. The total customer rebates and credits deducted within net sales for the three months ended April 30, 2024 and 2023 was \$2.3 million and \$2.0 million, respectively.

Research and Development Expenses

The amount of new product research and development ("R&D") expenses included in cost of sales for the six and three months ended April 30, 2024 and 2023 is as follows (in thousands):

	Six months en	ded April 30,	Three months e	ended April 30,
	2024	2023	2024	2023
R&D expenses	\$53,031	\$43,134	\$27,935	\$22,896

Redeemable Noncontrolling Interests

The holders of equity interests in certain of the Company's subsidiaries have rights ("Put Rights") that may be exercised on varying dates causing the Company to purchase their equity interests through fiscal 2032. The Put Rights, all of which relate either to common shares or membership interests in limited liability companies, provide that the cash consideration to be paid for their equity interests (the "Redemption Amount") be at fair value or a formula that management intended to reasonably approximate fair value based solely on a multiple of future earnings over a measurement period. Management's estimate of the aggregate Redemption Amount of all Put Rights that the Company could be required to pay is as follows (in thousands):

	April 30, 2024	October 31, 2023
Redeemable at fair value	\$309,005	\$308,472
Redeemable based on a multiple of future earnings	59,364	56,335
Redeemable noncontrolling interests	\$368,369	\$364,807

During fiscal 2024, the holders of a 15% noncontrolling equity interest in a subsidiary of the ETG that was acquired in fiscal 2019 exercised their option to cause the Company to purchase their noncontrolling interest over a four-year period ending in fiscal 2027. Accordingly, the Company acquired one-fourth of such interest in March 2024, which increased the Company's ownership interest in the subsidiary to 88.75%.

During fiscal 2022, the holder of a 19.9% noncontrolling equity interest in a subsidiary of the FSG that was acquired in fiscal 2015 exercised their option to cause the Company to purchase their noncontrolling interest over a four-year period ending in fiscal 2026. In December 2023, the Company acquired an additional one-fourth of such interest, which increased the Company's ownership interest in the subsidiary to 90.05%.

Accumulated Other Comprehensive Loss

Changes in the components of accumulated other comprehensive loss for the six months ended April 30, 2024 are as follows (in thousands):

	Foreign Currency Translation	Defined Benefit Pension Plan	Accumulated Other Comprehensive Loss
Balances as of October 31, 2023	(\$39,165)	(\$1,015)	(\$40,180)
Unrealized gain	4,477		4,477
Amortization of unrealized loss		26	26
Balances as of April 30, 2024	(\$34,688)	(\$989)	(\$35,677)

4. GOODWILL AND OTHER INTANGIBLE ASSETS

Changes in the carrying amount of goodwill by operating segment for the six months ended April 30, 2024 are as follows (in thousands):

	Segn	Segment		
	FSG	ETG	Consolidated Totals	
Balances as of October 31, 2023	\$1,824,305	\$1,450,022	\$3,274,327	
Goodwill acquired	7,577		7,577	
Foreign currency translation adjustments	649	2,190	2,839	
Adjustments to goodwill	(88)	813	725	
Balances as of April 30, 2024	\$1,832,443	\$1,453,025	\$3,285,468	

The goodwill acquired pertains to the fiscal 2024 acquisition described in Note 2, Acquisition, and represents the residual value after the allocation of the total consideration to the tangible and identifiable intangible assets acquired. The Company estimates that \$7 million of the goodwill acquired in fiscal 2024 will be deductible for income tax purposes. Foreign currency translation adjustments are included in other comprehensive income (loss) in the Company's Condensed Consolidated Statements of Comprehensive Income. The adjustments to goodwill represent immaterial measurement period adjustments to the allocation of the purchase consideration of certain fiscal 2023 acquisitions.

	As	As of April 30, 2024		As of October 3		023
	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount
Amortizing Assets:						
Customer relationships	\$988,166	(\$271,612)	\$716,554	\$967,090	(\$227,089)	\$740,001
Intellectual property	446,307	(120,027)	326,280	448,336	(121,503)	326,833
Other	8,627	(7,559)	1,068	8,685	(7,404)	1,281
	1,443,100	(399,198)	1,043,902	1,424,111	(355,996)	1,068,115
Non-Amortizing Assets:						
Trade names	289,435		289,435	289,166		289,166
	\$1,732,535	(\$399,198)	\$1,333,337	\$1,713,277	(\$355,996)	\$1,357,281

Identifiable intangible assets consist of the following (in thousands):

Amortization expense related to intangible assets for the six months ended April 30, 2024 and 2023 was \$60.8 million and \$36.9 million, respectively. Amortization expense related to intangible assets for the three months ended April 30, 2024 and 2023 was \$30.6 million and \$19.1 million, respectively. Amortization expense related to intangible assets for the remainder of fiscal 2024 is estimated to be \$60.4 million. Amortization expense for each of the next five fiscal years and thereafter is estimated to be \$116.2 million in fiscal 2025, \$110.5 million in fiscal 2026, \$106.1 million in fiscal 2027, \$100.6 million in fiscal 2028, \$95.1 million in fiscal 2029, and \$455.0 million thereafter.

5. SHORT-TERM AND LONG-TERM DEBT

A subsidiary of the Company acquired in the first quarter of fiscal 2023 ended its shortterm borrowing arrangement in the first quarter of fiscal 2024 during which it made net payments of \$13.9 million.

Long-term debt consists of the following (in thousands):

	April 30, 2024	October 31, 2023
Borrowings under revolving credit facility	\$1,175,000	\$1,250,000
2028 senior unsecured notes	600,000	600,000
2033 senior unsecured notes	600,000	600,000
Finance leases and notes payable	27,280	28,024
Less: Debt discount and debt issuance costs	(12,631)	(13,478)
	2,389,649	2,464,546
Less: Current maturities of long-term debt	(4,382)	(4,269)
	\$2,385,267	\$2,460,277

Revolving Credit Facility

The Company's borrowings under its revolving credit facility mature in fiscal 2028. As of April 30, 2024 and October 31 2023, the weighted average interest rate on borrowings under the Company's revolving credit facility ("Credit Facility") was 6.9% and 6.7%, respectively. The Credit Facility contains both financial and non-financial covenants. As of April 30, 2024, the Company was in compliance with all such covenants.

Senior Unsecured Notes

The Company's senior unsecured notes consist of \$600 million principal amount of 5.25% Senior Notes due August 1, 2028 (the "2028 Notes") and \$600 million principal amount of 5.35% Senior Notes due August 1, 2033 (the "2033 Notes" and, collectively with the 2028 Notes, the "Notes"). Interest on the Notes is payable semi-annually in arrears on February 1 and August 1 of each year, and commenced on February 1, 2024. The 2028 Notes and 2033 Notes each have an effective interest rate of 5.5%. The Notes are fully and unconditionally guaranteed on a senior unsecured basis by all of the Company's existing and future subsidiaries that guarantee the Company's obligations under the Credit Facility (the "Guarantor Group"). As of April 30, 2024, the Company was in compliance with all covenants related to the Notes.

The following table sets forth the carrying value and estimated fair value of the Company's Notes, which are classified as Level 1 financial instruments in the fair value hierarchy (in thousands). The Company estimated the fair value of the Notes by taking the weighted average of market quotes for the exact security that was actively traded on April 30, 2024 and October 31, 2023.

	April 30, 2024		October 31, 2023		
	Carrying Value	Fair Value	Carrying Value	Fair Value	
2028 Notes	\$594,705	\$594,792	\$594,158	\$579,762	
2033 Notes	592,664	590,118	592,364	552,594	
Total	\$1,187,369	\$1,184,910	\$1,186,522	\$1,132,356	

6. REVENUE

Contract Balances

Contract assets (unbilled receivables) represent revenue recognized on contracts using an over-time recognition model in excess of amounts invoiced to the customer. Contract liabilities (deferred revenue) represent customer advances and billings in excess of revenue recognized and are included within accrued expenses and other current liabilities and other long-term liabilities in the Company's Condensed Consolidated Balance Sheets.

Changes in the Company's contract assets and liabilities for the six months ended April 30, 2024 are as follows (in thousands):

	April 30, 2024	October 31, 2023	Change
Contract assets, current	\$110,158	\$111,702	(\$1,544)
Contract liabilities, current	76,650	87,556	(10,906)
Contract liabilities, long-term	52,186	_	52,186
Total contract liabilities	128,836	87,556	41,280
Net contract (liabilities) assets	(\$18,678)	\$24,146	(\$42,824)

The increase in the Company's total contract liabilities during the first six months of fiscal 2024 principally reflects the receipt of advance deposits on certain customer contracts, mainly at the FSG.

The amount of revenue that the Company recognized during the six and three months ended April 30, 2024 that was included in contract liabilities as of the beginning of fiscal 2024 was \$42.9 million and \$16.2 million, respectively.

Remaining Performance Obligations

Backlog, which the Company believes to be the equivalent of its remaining performance obligations, represents contractually committed or firm customer orders. As of April 30, 2024, the Company had \$1,846.2 million of remaining performance obligations associated with firm contracts pertaining to the majority of the products offered by the ETG and FSG. The Company will recognize net sales as these obligations are satisfied. The Company expects to recognize \$820.5 million of this amount during the remainder of fiscal 2024 and \$1,025.7 million thereafter, of which approximately half is expected to occur in fiscal 2025.

Disaggregation of Revenue

The following table summarizes the Company's net sales by product line for each operating segment (in thousands):

	Six months ended April 30,		Three months en	ded April 30,
	2024	2023	2024	2023
Flight Support Group:				
Aftermarket replacement parts (1)	\$798,879	\$426,986	\$403,725	\$218,343
Repair and overhaul parts and services (2)	283,763	149,001	148,181	77,851
Specialty products ⁽³⁾	183,306	187,493	95,326	96,008
Total net sales	1,265,948	763,480	647,232	392,202
Electronic Technologies Group:				
Electronic component parts primarily for defense, space and aerospace equipment ⁽⁴⁾	474,404	395,320	253,758	220,742
Electronic component parts for equipment in various other industries ⁽⁵⁾	130,860	161,498	65,564	81,017
Total net sales	605,264	556,818	319,322	301,759
Intersegment sales	(19,454)	(11,542)	(11,159)	(6,120)
Total consolidated net sales	\$1,851,758	\$1,308,756	\$955,395	\$687,841

⁽¹⁾ Includes various jet engine and aircraft component replacement parts.

⁽²⁾ Includes primarily the sale of parts consumed in various repair and overhaul services on selected jet engine and aircraft components, avionics, instruments, composites and flight surfaces of commercial and military aircraft.

⁽³⁾ Includes primarily the sale of specialty components such as thermal insulation blankets, renewable/ reusable insulation systems, advanced niche components, complex composite assemblies, and expanded foil mesh as well as machining, brazing, fabricating and welding services generally to original equipment manufacturers.

⁽⁴⁾ Includes various component parts such as electro-optical infrared simulation and test equipment, electro-optical laser products, electro-optical, microwave and other power equipment, high-speed interface products, power conversion products, underwater locator beacons, emergency locator transmission beacons, traveling wave tube amplifiers, microwave power modules, a wide variety of memory products and radio frequency (RF) and microwave products, crashworthy and ballistically self-sealing auxiliary fuel systems, high performance communications and electronic intercept receivers and tuners, high performance active antenna systems and airborne antennas, technical surveillance countermeasures (TSCM) equipment, custom high power filters and filter assemblies, radiation assurance services and products, and high-reliability, complex, passive electronic components and rotary joint assemblies.

⁽⁵⁾ Includes various component parts such as electromagnetic and radio frequency interference shielding, high voltage interconnection devices, high voltage advanced power electronics, harsh environment

connectivity products, custom molded cable assemblies, silicone material for a variety of demanding applications, and rugged small form-factor embedded computing solutions, and high performance test sockets and adaptors.

The following table summarizes the Company's net sales by industry for each operating segment (in thousands):

	Six months end	Six months ended April 30,		ded April 30,
	2024	2023	2024	2023
Flight Support Group:				
Aerospace	\$939,590	\$523,893	\$478,349	\$269,353
Defense and Space	288,678	196,909	149,906	101,267
Other ⁽¹⁾	37,680	42,678	18,977	21,582
Total net sales	1,265,948	763,480	647,232	392,202
Electronic Technologies Group:				
Defense and Space	300,757	260,571	164,981	138,609
Other ⁽²⁾	200,442	215,794	99,832	118,024
Aerospace	104,065	80,453	54,509	45,126
Total net sales	605,264	556,818	319,322	301,759
Intersegment sales	(19,454)	(11,542)	(11,159)	(6,120)
Total consolidated net sales	\$1,851,758	\$1,308,756	\$955,395	\$687,841

⁽¹⁾ Principally industrial products.

⁽²⁾ Principally other electronics and medical products.

7. INCOME TAXES

The Company's effective tax rate decreased to 16.9% in the first six months of fiscal 2024, down from 19.3% in the first six months of fiscal 2023. The decrease in the Company's effective tax rate principally reflects a larger tax benefit from stock option exercises recognized in the first quarter of fiscal 2024. The Company recognized a discrete tax benefit from stock option exercises in both the first quarter of fiscal 2024 and 2023 of \$13.6 million and \$6.2 million, respectively.

The Company's effective tax rate was 21.2% in both the second quarter of fiscal 2024 and 2023.

8. FAIR VALUE MEASUREMENTS

The Company's assets and liabilities that were measured at fair value on a recurring basis are set forth by level within the fair value hierarchy in the following tables (in thousands):

		As of April 30,	2024	
	Quoted Prices in Active Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)	Total
Assets:				
Deferred compensation plan:				
Corporate-owned life insurance	\$—	\$275,579	\$—	\$275,579
Money market fund	10,460	_	—	10,460
Total assets	\$10,460	\$275,579	\$—	\$286,039
Liabilities:				
Contingent consideration	\$—	\$—	\$45,917	\$45,917
	As of October 31, 2023			
		As of October 31	, 2023	
	Quoted Prices in Active Markets for Identical Assets (Level 1)	As of October 31 Significant Other Observable Inputs (Level 2)	, 2023 Significant Unobservable Inputs (Level 3)	Total
Assets:	in Active Markets for Identical Assets	Significant Other Observable Inputs	Significant Unobservable Inputs	Total
Assets: Deferred compensation plan:	in Active Markets for Identical Assets	Significant Other Observable Inputs	Significant Unobservable Inputs	Total
	in Active Markets for Identical Assets	Significant Other Observable Inputs	Significant Unobservable Inputs	Total \$227,710
Deferred compensation plan:	in Active Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)	
Deferred compensation plan: Corporate-owned life insurance	in Active Markets for Identical Assets (Level 1) \$—	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)	\$227,710
Deferred compensation plan: Corporate-owned life insurance Money market fund	in Active Markets for Identical Assets (Level 1) \$	Significant Other Observable Inputs (Level 2) \$227,710	Significant Unobservable Inputs (Level 3) \$	\$227,710 5,829
Deferred compensation plan: Corporate-owned life insurance Money market fund	in Active Markets for Identical Assets (Level 1) \$	Significant Other Observable Inputs (Level 2) \$227,710	Significant Unobservable Inputs (Level 3) \$	\$227,710 5,829

The Company maintains the HEICO Corporation Leadership Compensation Plan (the "LCP"), which is a non-qualified deferred compensation plan. The assets of the LCP principally represent cash surrender values of life insurance policies, which derive their fair values from investments in mutual funds that are managed by an insurance company, and are classified within Level 2 and valued using a market approach. Certain other assets of the LCP represent an investment in a money market fund that is classified within Level 1. The assets of the LCP are held within an irrevocable trust and classified within other assets in the Company's Condensed Consolidated Balance Sheets. The related liabilities of the LCP are included within other long-term liabilities and accrued expenses and other current liabilities in the Company's Condensed Consolidated Balance Sheets and have an aggregate value of \$284.0 million as of April 30, 2024 and \$226.2 million as of October 31, 2023.

In connection with a fiscal 2023 acquisition that is part of the FSG, the Company assumed an agreement which may have obligated it to pay contingent consideration of \$17.5 million if certain operating entities of the acquired company met a calendar year 2023 earnings objective and obtained a certain level of new orders with deliveries scheduled in calendar year 2024, of which both targets were tied to a specific customer contract. Both requirements were met as of October 31,2023. However, payment of the earnout is also predicated on no indication of a significant change with respect to the underlying customer agreement. In the second quarter of fiscal 2024, the customer notified the Company that it intends to reduce its future orders. As a result, the parties to this agreement agreed to settle on a specific contingent consideration amount of \$11.0 million, which is expected to be paid in the third quarter of fiscal 2024. Accordingly, the \$17.3 million estimated fair value of contingent consideration as of October 31, 2023 was reduced to \$11.0 million as of April 30, 2024.

As part of the agreement to acquire 80.36% of the stock of a subsidiary by the ETG in fiscal 2022, the Company may be obligated to pay contingent consideration of up to \$12.1 million in fiscal 2027 based on the earnings of the acquired entity during fiscal years 2025 and 2026 provided the entity meets a certain earnings objective during each of fiscal years 2024 to 2026. As of April 30, 2024, the estimated fair value of the contingent consideration was \$5.6 million.

As part of the agreement to acquire 96% of the stock of a subsidiary by the FSG in fiscal 2022, the Company may be obligated to pay contingent consideration of up to \$27.4 million in fiscal 2027 based on the earnings of the acquired entity during fiscal years 2025 and 2026 provided the entity meets certain earnings objectives during each of fiscal years 2022 to 2024. As of April 30, 2024, the estimated fair value of the contingent consideration was \$20.4 million.

As part of the agreement to acquire 74% of the membership interests of a subsidiary by the FSG in fiscal 2022, the Company would be obligated to pay contingent consideration of \$14.1 million in fiscal 2027 only if the acquired entity met a certain earnings objective during the five-year period following the acquisition. Based on the actual earnings of the acquired entity subsequent to the acquisition and forecasted earnings over the remainder of the earnout period, the Company does not expect that the required earnings objective will be met. Accordingly, as of April 30, 2024 and October 31, 2023, the Company did not accrue any contingent consideration for this agreement.

As part of the agreement to acquire 89.99% of the equity interests of a subsidiary by the ETG in fiscal 2020, the Company may be obligated to pay contingent consideration of up to CAD \$13.5 million, or \$9.9 million, in fiscal 2025 should the acquired entity meet certain earnings objectives during fiscal 2023 and 2024. As of April 30, 2024, the estimated fair value of the contingent consideration was CAD \$12.3 million, or \$9.0 million.

As part of the agreement to acquire a subsidiary by the ETG in fiscal 2017, the Company paid contingent consideration of \$20.0 million in December 2023 as the acquired entity met a certain earnings objective during the first six years following the acquisition.

Acquisition Date	Fair Value	Unobservable Input	Range	Weighted Average ⁽¹⁾
8-4-2023	\$11,000	—	—	—
9-1-2022	\$5,577	Compound annual revenue growth rate	9% - 26%	18%
		Discount rate	9.3% - 9.3%	9.3%
7-18-2022	20,370	Compound annual revenue growth rate	1% - 11%	6%
		Discount rate	9.3% - 9.3%	9.3%
8-18-2020	8,970	Compound annual revenue growth rate	10% - 15%	14%
		Discount rate	9.9% - 9.9%	9.9%

The following unobservable inputs were used to derive the estimated fair value of the Company's Level 3 contingent consideration liabilities as of April 30, 2024 (\$ in thousands):

⁽¹⁾ Unobservable inputs were weighted by the relative fair value of the contingent consideration liability.

Changes in the Company's contingent consideration liabilities measured at fair value on a recurring basis using unobservable inputs (Level 3) for the six months ended April 30, 2024 are as follows (in thousands):

	Liabilities
Balance as of October 31, 2023	\$71,136
Payment of contingent consideration	(20,000)
Decrease in accrued contingent consideration, net	(5,326)
Foreign currency transaction adjustments	107
	\$45,917
Included in the accompanying Condensed Consolidated Balance Sheet under the following captions:	
Accrued expenses and other current liabilities	\$19,970
Other long-term liabilities	25,947
	\$45,917

The Company records changes in accrued contingent consideration and foreign currency transaction adjustments within SG&A expenses in its Condensed Consolidated Statements of Operations.

The carrying amounts of the Company's cash and cash equivalents, accounts receivable, trade accounts payable and accrued expenses and other current liabilities approximate fair value as of April 30, 2024 due to the relatively short maturity of the respective instruments. The carrying amount of borrowings under the Company's credit facility approximates fair value due to its variable interest rate. See Note 5, Short-Term and Long-Term Debt, for the estimated fair value of the Company's senior unsecured notes.

9. NET INCOME PER SHARE ATTRIBUTABLE TO HEICO SHAREHOLDERS

The computation of basic and diluted net income per share attributable to HEICO shareholders is as follows (in thousands, except per share data):

	Six months ended April 30,		Three months ended April 30,	
	2024	2023	2024	2023
Numerator:				
Net income attributable to HEICO	\$237,844	\$198,147	\$123,146	\$105,120
Denominator:				
Weighted average common shares outstanding - basic	138,325	136,786	138,386	136,916
Effect of dilutive stock options	1,651	1,804	1,673	1,684
Weighted average common shares outstanding - diluted	139,976	138,590	140,059	138,600
Net income per share attributable to HEICO shareholders:				
Basic	\$1.72	\$1.45	\$.89	\$.77
Diluted	\$1.70	\$1.43	\$.88	\$.76
Anti-dilutive stock options excluded	1,215	1,045	1,009	1,340

10. OPERATING SEGMENTS

Information on the Company's two operating segments, the FSG and the ETG, for the six and three months ended April 30, 2024 and 2023, respectively, is as follows (in thousands):

	Segment		Other, Primarily Corporate and	Consolidated	
	FSG	ETG	Intersegment ⁽¹⁾	Totals	
Six months ended April 30, 2024:					
Net sales	\$1,265,948	\$605,264	(\$19,454)	\$1,851,758	
Depreciation	11,929	11,061	609	23,599	
Amortization	36,304	25,649	784	62,737	
Operating income	284,967	130,591	(26,195)	389,363	
Capital expenditures	12,714	13,028	583	26,325	
Six months ended April 30, 2023:					
Net sales	\$763,480	\$556,818	(\$11,542)	\$1,308,756	
Depreciation	8,152	9,461	535	18,148	
Amortization	13,286	24,802	548	38,636	
Operating income	183,521	124,516	(21,513)	286,524	
Capital expenditures	10,643	11,058	220	21,921	
Three months ended April 30, 2024:					
Net sales	\$647,232	\$319,322	(\$11,159)	\$955,395	
Depreciation	5,442	5,522	305	11,269	
Amortization	18,447	12,723	392	31,562	
Operating income	148,876	75,263	(14,986)	209,153	
Capital expenditures	5,982	6,854	112	12,948	
Three months ended April 30, 2023:					
Net sales	\$392,202	\$301,759	(\$6,120)	\$687,841	
Depreciation	3,974	5,523	265	9,762	
Amortization	6,555	13,133	274	19,962	
Operating income	99,912	67,979	(10,801)	157,090	
Capital expenditures	3,990	6,969	116	11,075	

⁽¹⁾ Intersegment activity principally consists of net sales from the ETG to the FSG.

Total assets by operating segment are as follows (in thousands):

	Segment		ent Other, Primarily	
	FSG	ETG	Corporate	Consolidated Totals
Total assets as of April 30, 2024	\$4,104,662	\$2,901,556	\$359,627	\$7,365,845
Total assets as of October 31, 2023	4,006,748	2,915,300	273,015	7,195,063

11. COMMITMENTS AND CONTINGENCIES

Guarantees

As of April 30, 2024, the Company has arranged for standby letters of credit aggregating \$11.2 million, which are supported by its revolving credit facility and principally pertain to performance guarantees related to customer contracts entered into by certain of the Company's subsidiaries as well as payment guarantees related to potential workers' compensation claims.

Product Warranty

Changes in the Company's product warranty liability for the six months ended April 30, 2024 and 2023, respectively, are as follows (in thousands):

	Six months ende	Six months ended April 30,		
	2024	2023		
Balances as of beginning of fiscal year	\$3,847	\$3,296		
Accruals for warranties	1,425	1,222		
Warranty claims settled	(1,459)	(1,074)		
Balances as of April 30	\$3,813	\$3,444		

Litigation

The Company is involved in various legal actions arising in the normal course of business. Based upon the Company's and its legal counsel's evaluations of any claims or assessments, management is of the opinion that the outcome of these matters will not have a material adverse effect on the Company's results of operations, financial position or cash flows.

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Overview

This discussion of our financial condition and results of operations should be read in conjunction with our condensed consolidated financial statements and notes thereto included herein. The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ materially from those estimates if different assumptions were used or different events ultimately transpire.

Our critical accounting estimates, which require management to make judgments about matters that are inherently uncertain, are described in Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," under the heading "Critical Accounting Estimates" in our Annual Report on Form 10-K for the year ended October 31, 2023. Based on our recent assessment, we no longer deem revenue recognition to be a critical accounting estimate. Historically, the majority of our revenue was recognized at a point-in-time and involved minimal estimates to determine when control had transferred. We now recognize an even greater portion of our revenue at a point-in-time as a result of the acquisitions of Wencor Group and Exxelia International SAS in fiscal 2023. Additionally, we do not believe that the factors involving estimation uncertainty that are used when we recognize revenue using an over-time recognition model for certain contracts are reasonably likely to have a material impact on our financial position or results of operations. Other than the removal of revenue recognition, there have been no material changes to our critical accounting estimates during the six months ended April 30, 2024.

Our business is comprised of two operating segments: the Flight Support Group ("FSG"), consisting of HEICO Aerospace Holdings Corp. and HEICO Flight Support Corp. and their respective subsidiaries; and the Electronic Technologies Group ("ETG"), consisting of HEICO Electronic Technologies Corp. and its subsidiaries.

Our results of operations for the six and three months ended April 30, 2024 have been affected by the fiscal 2023 acquisitions as further detailed in Note 2, Acquisitions, of the Notes to Consolidated Financial Statements of our Annual Report on Form 10-K for the year ended October 31, 2023.

Results of Operations

The following table sets forth the results of our operations, net sales and operating income by segment and the percentage of net sales represented by the respective items in our Condensed Consolidated Statements of Operations (in thousands):

	Six months ended April 30,		Three months ended April 30,	
	2024	2023	2024	2023
Net sales	\$1,851,758	\$1,308,756	\$955,395	\$687,841
Cost of sales	1,133,194	798,445	583,600	421,329
Selling, general and administrative expenses	329,201	223,787	162,642	109,422
Total operating costs and expenses	1,462,395	1,022,232	746,242	530,751
Operating income	\$389,363	\$286,524	\$209,153	\$157,090
Net sales by segment:				
Flight Support Group	\$1,265,948	\$763,480	\$647,232	\$392,202
Electronic Technologies Group	605,264	556,818	319,322	301,759
Intersegment sales	(19,454)	(11,542)	(11,159)	(6,120)
	\$1,851,758	\$1,308,756	\$955,395	\$687,841
Operating income by segment:				
Flight Support Group	\$284,967	\$183,521	\$148,876	\$99,912
Electronic Technologies Group	130,591	124,516	75,263	67,979
Other, primarily corporate	(26,195)	(21,513)	(14,986)	(10,801)
	\$389,363	\$286,524	\$209,153	\$157,090
N7 1	100.00/	100.00/	100.00/	100.00/
Net sales	100.0%	100.0%	100.0%	100.0%
Gross profit	38.8%	39.0%	38.9%	38.7%
Selling, general and administrative expenses	17.8%	17.1%	17.0%	15.9%
Operating income	21.0%	21.9%	21.9%	22.8%
Interest expense	(4.2%)	(1.3%)	(4.0%)	(1.7%)
Other income	.1%	.1%	%	<u> %</u>
Income tax expense	2.9%	4.0%	3.8%	4.5%
Net income attributable to noncontrolling interests	1.2%	1.5%	1.2%	1.4%
Net income attributable to HEICO	12.8%	15.1%	12.9%	15.3%

Comparison of First Six Months of Fiscal 2024 to First Six Months of Fiscal 2023

Net Sales

Our consolidated net sales in the first six months of fiscal 2024 increased by 41% to a record \$1,851.8 million, up from net sales of \$1,308.8 million in the first six months of fiscal 2023. The increase in consolidated net sales principally reflects an increase of \$502.5 million (a 66% increase) to a record \$1,265.9 million in net sales of the FSG and an increase of \$48.4 million (a 9% increase) to a record \$605.3 million in net sales of the ETG. The net sales increase in the FSG reflects \$408.7 million contributed by fiscal 2023 and 2024 acquisitions as well as strong organic growth of 12%. The FSG's organic net sales growth reflects increased demand within its aftermarket replacement parts and repair and overhaul parts and services product lines resulting in net sales increases of \$82.9 million and \$15.1 million, respectively. The net sales increase in the ETG includes \$39.4 million contributed by a fiscal 2023 acquisition. The ETG's net sales increase also includes organic net sales growth of \$26.6 million and \$10.1 million for its defense and aerospace products, respectively, partially offset by organic net sales decreases of \$19.7 million and \$9.1 million for its other electronics and medical products, respectively. Although sales price changes were not a significant contributing factor to the change in net sales of the FSG and ETG in the first six months of fiscal 2024, continued cost inflation may lead to higher sales prices during the remainder of fiscal 2024.

Gross Profit and Operating Expenses

Our consolidated gross profit margin was 38.8% in the first six months of fiscal 2024, as compared to 39.0% in the first six months of fiscal 2023 and is inclusive of a .8% increase in the ETG's gross profit margin and no change in the FSG's gross profit margin. The increase in the ETG's gross profit margin principally reflects the previously mentioned higher net sales of defense products, partially offset by the previously mentioned decrease in net sales of other electronics products. Total new product research and development expenses included within our consolidated cost of sales were \$53.0 million in the first six months of fiscal 2023.

Our consolidated selling, general and administrative ("SG&A") expenses were \$329.2 million in the first six months of fiscal 2024, as compared to \$223.8 million in the first six months of fiscal 2023. The increase in consolidated SG&A expenses principally reflects \$79.9 million attributable to our fiscal 2023 and 2024 acquisitions, costs incurred to support the previously mentioned net sales growth resulting in increases of \$16.2 million and \$5.6 million in other general and administrative expenses and other selling expenses, respectively, and a \$9.1 million prior year impact from the amendment and termination of a contingent consideration agreement pertaining to a fiscal 2021 acquisition, partially offset by a \$5.3 million decrease in acquisition costs.

Our consolidated SG&A expenses as a percentage of net sales were 17.8% in the first six months of fiscal 2024, as compared to 17.1% in the first six months of fiscal 2023. The increase in consolidated SG&A expenses as a percentage of net sales principally reflects a .7% impact

from the previously mentioned amendment and termination of a contingent consideration agreement and a .5% impact from higher intangible asset amortization expense, partially offset by a .4% impact from the previously mentioned lower acquisition costs.

Operating Income

Our consolidated operating income increased by 36% to a record \$389.4 million in the first six months of fiscal 2024, up from \$286.5 million in the first six months of fiscal 2023. The increase in consolidated operating income principally reflects a \$101.4 million increase (a 55% increase) to a record \$285.0 million in operating income of the FSG and a \$6.1 million increase (a 5% increase) to \$130.6 million in operating income of the ETG. The increase in operating income of the FSG principally reflects the previously mentioned net sales growth, partially offset by a \$23.0 million increase in intangible asset amortization expense, a \$9.8 million increase in inventory obsolescence expense, and a \$9.1 million prior year impact from the previously mentioned termination of a contingent consideration agreement. The increase in operating income of the ETG principally reflects the previously mentioned net sales increase, a \$5.1 million decrease in acquisition costs and the previously mentioned favorable gross profit margin, partially offset by a lower level of SG&A efficiencies.

Our consolidated operating income as a percentage of net sales was 21.0% in the first six months of fiscal 2024, as compared to 21.9% in the first six months of fiscal 2023. The decrease in consolidated operating income as a percentage of net sales principally reflects a decrease in the FSG's operating income as a percentage of net sales to 22.5% in the first six months of fiscal 2024, as compared to 24.0% in the first six months of fiscal 2023 and a decrease in the ETG's operating income as a percentage of net sales to 21.6% in the first six months of fiscal 2024, as compared to 22.4% in the first six months of fiscal 2023. The decrease in the FSG's operating income as a percentage of net sales principally reflects a 1.2% prior year impact from the previously mentioned amendment and termination of a contingent consideration agreement and a 1.2% impact from the previously mentioned higher intangible asset amortization expense. partially offset by a .9% impact from lower performance-based compensation expense as a percentage of net sales. The decrease in the ETG's operating income as a percentage of net sales principally reflects a 1.6% impact from an increase in SG&A expenses as a percentage of net sales, partially offset by the previously mentioned increase in gross profit margin. The increase in the ETG's SG&A expenses as a percentage of net sales mainly reflects the previously mentioned lower level of efficiencies, partially offset by a .9% impact from the previously mentioned lower acquisition costs.

Interest Expense

Interest expense increased to \$77.1 million in the first six months of fiscal 2024, as compared to \$17.4 million in the first six months of fiscal 2023. The increase in interest expense was principally due to an increase in the amount of outstanding debt related to fiscal 2023 acquisitions.

Other Income

Other income in the first six months of fiscal 2024 and 2023 was not material.

Income Tax Expense

Our effective tax rate decreased to 16.9% in the first six months of fiscal 2024, down from 19.3% in the first six months of fiscal 2023. The decrease in our effective tax rate principally reflects a larger tax benefit from stock option exercises recognized in the first quarter of fiscal 2024. We recognized a discrete tax benefit from stock option exercises in both the first quarter of fiscal 2024 and 2023 of \$13.6 million and \$6.2 million, respectively.

Net Income Attributable to Noncontrolling Interests

Net income attributable to noncontrolling interests relates to the 20% noncontrolling interest held by Lufthansa Technik AG in HEICO Aerospace Holdings Corp. and the noncontrolling interests held by others in certain subsidiaries of the FSG and ETG. Net income attributable to noncontrolling interests was \$22.5 million in the first six months of fiscal 2024, as compared to \$19.9 million in the first six months of fiscal 2023. The increase in net income attributable to noncontrolling interests principally reflects improved operating results of certain subsidiaries of the FSG and ETG in which noncontrolling interests are held.

Net Income Attributable to HEICO

Net income attributable to HEICO increased by 20% to a record \$237.8 million, or \$1.70 per diluted share, in the first six months of fiscal 2024, up from \$198.1 million, or \$1.43 per diluted share, in the first six months of fiscal 2023 principally reflecting the previously mentioned higher consolidated operating income, partially offset by the previously mentioned higher interest expense.

Comparison of Second Quarter of Fiscal 2024 to Second Quarter of Fiscal 2023

Net Sales

Our consolidated net sales in the second quarter of fiscal 2024 increased by 39% to a record \$955.4 million, up from net sales of \$687.8 million in the second quarter of fiscal 2023. The increase in consolidated net sales principally reflects an increase of \$255.0 million (a 65% increase) to a record \$647.2 million in net sales of the FSG and an increase of \$17.6 million (a 6% increase) to \$319.3 million in net sales of the ETG. The net sales increase in the FSG reflects \$206.6 million contributed by fiscal 2023 and 2024 acquisitions as well as strong organic growth of 12%. The FSG's organic net sales growth reflects increased demand within its aftermarket replacement parts and repair and overhaul parts and services product lines resulting in net sales increases of \$45.0 million and \$4.1 million, respectively. The net sales increase in the ETG principally reflects organic growth of 4%. The ETG's organic net sales growth is mainly attributable to increased demand for its defense and aerospace products resulting in net sales increases of \$26.4 million and \$6.8 million, respectively, partially offset by decreased demand for its other electronics and medical products resulting in net sales decreases of \$12.4 million and \$4.3 million, respectively. Although sales price changes were not a significant contributing factor to the change in net sales of the FSG and ETG in the second quarter of fiscal 2024, continued cost inflation may lead to higher sales prices during the remainder of fiscal 2024.

Gross Profit and Operating Expenses

Our consolidated gross profit margin improved to 38.9% in the second quarter of fiscal 2024, up from 38.7% in the second quarter of fiscal 2023, principally reflecting increases of 1.7% and .3% in the ETG's and FSG's gross profit margin, respectively. The increase in the ETG's gross profit margin principally reflects the previously mentioned increase in net sales of defense products, partially offset by the previously mentioned decrease in net sales of other electronics products. The increase in the FSG's gross profit margin principally reflects the previously mentioned higher net sales, partially offset by a .5% impact from higher inventory obsolescence expense. Total new product research and development expenses included within our consolidated cost of sales were \$27.9 million in the second quarter of fiscal 2023.

Our consolidated SG&A expenses were \$162.6 million in the second quarter of fiscal 2024, as compared to \$109.4 million in the second quarter of fiscal 2023. The increase in consolidated SG&A expenses principally reflects \$30.8 million attributable to our fiscal 2023 and 2024 acquisitions, costs incurred to support the previously mentioned net sales growth resulting in increases of \$10.4 million and \$2.9 million in other general and administrative expenses and other selling expenses, respectively, and a \$9.1 million prior year impact from the amendment and termination of a contingent consideration agreement pertaining to a fiscal 2021 acquisition.

Our consolidated SG&A expenses as a percentage of net sales were 17.0% in the second quarter of fiscal 2024, as compared to 15.9% in the second quarter of fiscal 2023. The increase in consolidated SG&A expenses as a percentage of net sales principally reflects a 1.3% impact from the previously mentioned amendment and termination of a contingent consideration agreement.

Operating Income

Our consolidated operating income increased by 33% to a record \$209.2 million in the second quarter of fiscal 2024, up from \$157.1 million in the second quarter of fiscal 2023. The increase in consolidated operating income principally reflects a \$49.0 million increase (a 49% increase) to a record \$148.9 million in operating income of the FSG and a \$7.3 million increase (an 11% increase) to \$75.3 million in operating income of the ETG. The increase in operating income of the FSG principally reflects the previously mentioned net sales growth, partially offset by an \$11.9 million increase in intangible asset amortization expense, a \$9.1 million prior year impact from the previously mentioned amendment and termination of a contingent consideration agreement, and a \$4.8 million increase in inventory obsolescence expense. The increase in operating income of the ETG principally reflects the previously mentioned improved gross profit margin and net sales growth, partially offset by a lower level of SG&A efficiencies.

Our consolidated operating income as a percentage of net sales was 21.9% in the second quarter of fiscal 2024, as compared to 22.8% in the second quarter of fiscal 2023. The decrease in consolidated operating income as a percentage of net sales principally reflects a decrease in the FSG's operating income as a percentage of net sales to 23.0% in the second quarter of fiscal 2024, as compared to 25.5% in the second quarter of fiscal 2023, partially offset by an increase in the ETG's operating income as a percentage of net sales to 23.6% in the second quarter of fiscal 2024, up from 22.5% in the second quarter of fiscal 2023. The decrease in the FSG's operating income as a percentage of net sales principally reflects a 2.3% prior year impact from the previously mentioned amendment and termination of a contingent consideration agreement and a 1.2% impact from the previously mentioned increase in intangible asset amortization expense, partially offset by a .9% impact from lower performance-based compensation expense as a percentage of net sales. The increase in the ETG's operating income as a percentage of net sales principally reflects the previously mentioned improved gross profit margin and net sales growth, partially offset by a lower level of SG&A efficiencies.

Interest Expense

Interest expense increased to \$38.5 million in the second quarter of fiscal 2024, as compared to \$11.4 million in the second quarter of fiscal 2023. The increase in interest expense was principally due to an increase in the amount of outstanding debt related to fiscal 2023 acquisitions.

Other Income

Other income in the second quarter of fiscal 2024 and 2023 was not material.

Income Tax Expense

Our effective tax rate in both the second quarter of fiscal 2024 and 2023 was 21.2%.

Net Income Attributable to Noncontrolling Interests

Net income attributable to noncontrolling interests relates to the 20% noncontrolling interest held by Lufthansa Technik AG in HEICO Aerospace Holdings Corp. and the noncontrolling interests held by others in certain subsidiaries of the FSG and ETG. Net income attributable to noncontrolling interests was \$11.8 million in the second quarter of fiscal 2024, as compared to \$9.9 million in the second quarter of fiscal 2023. The increase in net income attributable to noncontrolling interests principally reflects improved operating results of certain subsidiaries of the FSG and ETG in which noncontrolling interests are held.

Net Income Attributable to HEICO

Net income attributable to HEICO increased by 17% to \$123.1 million, or \$.88 per diluted share, in the second quarter of fiscal 2024, up from \$105.1 million, or \$.76 per diluted share, in the second quarter of fiscal 2023 principally reflecting the previously mentioned higher consolidated operating income, partially offset by the previously mentioned higher interest expense.

Outlook

As we look ahead to the remainder of fiscal 2024, we continue to anticipate net sales growth in both the FSG and ETG, principally driven by contributions from our fiscal 2023 acquisitions and demand for the majority of our products. Additionally, we plan to continue our commitment to developing new products and services and further market penetration, while maintaining our financial strength and flexibility.

Liquidity and Capital Resources

Our principal uses of cash include acquisitions, capital expenditures, interest payments, cash dividends, distributions to noncontrolling interests and working capital needs. We now anticipate fiscal 2024 capital expenditures to be approximately \$60 to \$65 million. We finance our activities primarily from our operating and financing activities, including borrowings under our revolving credit facility. The revolving credit facility and senior unsecured notes contain both financial and non-financial covenants. As of April 30, 2024, we were in compliance with all such covenants and our total debt to shareholders' equity ratio was 69.4%.

Based on our current outlook, we believe that net cash provided by operating activities and available borrowings under our revolving credit facility will be sufficient to fund our cash requirements for at least the next twelve months.

Operating Activities

Net cash provided by operating activities was \$252.8 million in the first six months of fiscal 2024 and consisted primarily of net income from consolidated operations of \$260.4 million, depreciation and amortization expense of \$86.3 million (a non-cash item), net changes of \$41.8 million included in the "Other" caption (principally the receipt of advance deposits on certain long-term customer contracts), and net changes in other long-term liabilities and assets related to the HEICO LCP of \$17.3 million (principally participant deferrals and employer contributions), partially offset by a \$148.2 million increase in net working capital. The increase in net working capital is inclusive of a \$71.1 million increase in accrued expenses and other current liabilities mainly reflecting the payment of fiscal 2023 accrued performance-based compensation, partially offset by accrued fiscal 2024 performance-based compensation, as well as an \$11.4 million decrease in trade accounts payable and a \$9.2 million increase in prepaid expenses and other current assets.

Net cash provided by operating activities increased by \$98.4 million in the first six months of fiscal 2024, up from \$154.4 million in the first six months of fiscal 2023. The increase is principally attributable to a \$42.3 million increase in net income from consolidated operations, a \$48.5 million increase in the "Other" caption mainly from the previously mentioned receipt of advance long-term customer deposits in fiscal 2024, a \$29.6 million increase in depreciation and amortization expense, and a \$9.1 million prior year impact from the amendment and termination of a contingent consideration agreement, partially offset by a \$34.5 million increase and other current liabilities partially offset by a \$26.5 million decrease in accounts receivable.

Investing Activities

Net cash used in investing activities totaled \$85.3 million in the first six months of fiscal 2024 and related primarily to acquisitions of \$46.2 million, capital expenditures of \$26.3 million and investments related to the HEICO LCP of \$14.4 million. Further details regarding our fiscal 2024 acquisition may be found in Note 2, Acquisition, of the Notes to Condensed Consolidated Financial Statements.

Financing Activities

Net cash used in financing activities in the first six months of fiscal 2024 totaled \$135.2 million. During the first six months of fiscal 2024, we made \$125.0 million of payments on our revolving credit facility, \$15.4 million of distributions to noncontrolling interests, \$13.9 million of net payments on short-term debt, \$13.8 million of cash dividends paid on our common stock, and \$13.8 million of contingent consideration payments, partially offset by \$50.0 million of borrowings on our revolving credit facility and \$4.2 million of proceeds from stock option exercises.

Other Obligations and Commitments

There have not been any material changes to our other obligations and commitments that were included in our Annual Report on Form 10-K for the year ended October 31, 2023.

New Accounting Pronouncements

See Note 1, Summary of Significant Accounting Policies - New Accounting Pronouncements, of the Notes to Condensed Consolidated Financial Statements for additional information.

Guarantor Group Summarized Financial Information

On July 27, 2023, we completed the public offer and sale of senior unsecured notes, which consisted of \$600 million principal amount of 5.25% Senior Notes due August 1, 2028 (the "2028 Notes") and \$600 million principal amount of 5.35% Senior Notes due August 1, 2033 (the "2033 Notes" and, collectively with the 2028 Notes, the "Notes"). The Notes are fully and unconditionally guaranteed on a senior unsecured basis by all of our existing and future subsidiaries that guarantee our obligations under our revolving credit facility ("Credit Facility") (the "Guarantor Group").

The Notes were issued pursuant to an Indenture, dated as of July 27, 2023 (the "Base Indenture"), between HEICO and certain of its subsidiaries (collectively, the "Subsidiary Guarantors") and Truist Bank, as trustee (the "Trustee"), as supplemented by a First Supplemental Indenture, dated as of July 27, 2023 (the "First Supplemental Indenture" and, together with the Base Indenture, the "Indenture"), between us, the Subsidiary Guarantors and the Trustee. The Notes are direct, unsecured senior obligations of HEICO and rank equally in right of payment with all of our existing and future senior unsecured indebtedness. Each Subsidiary Guarantor is owned either directly or indirectly by the Company and jointly and severally guarantee our obligations under the Notes. None of the Subsidiary Guarantors are organized outside of the U.S.

Under the Indenture, holders of the Notes will be deemed to have consented to the release of a subsidiary guarantee provided by a subsidiary guarantor, without any action required on the part of the Trustee or any holder of the Notes, upon such subsidiary guarantor ceasing to guarantee or to be an obligor with respect to the Credit Facility. Accordingly, if the lenders under the Credit Facility release a subsidiary guarantor from its guarantee of, or obligations as a borrower under, the Credit Facility, the obligations of the subsidiary guarantors to guarantee the Notes will immediately terminate. If any of our future subsidiaries incur obligations under the Credit Facility while the Notes are outstanding, then such subsidiary will be required to guarantee the Notes. In addition, a subsidiary guarantor will be released and relieved from all its obligations under its subsidiary guarantee in the following circumstances, each of which is permitted by the indenture:

- upon the sale or other disposition (including by way of consolidation or merger), in one transaction or a series of related transactions, of a majority of the total voting stock of such subsidiary guarantor (other than to us or any of our affiliates); or
- upon the sale or disposition of all or substantially all the property of such subsidiary guarantor (other than to any of our affiliates or another subsidiary guarantor);

provided, however, that, in each case, such transaction is permitted by the Credit Facility and after giving effect to such transaction, such subsidiary guarantor is no longer liable for any subsidiary guarantee or other obligations in respect of the Credit Facility. The subsidiary guarantee of a subsidiary guarantor also will be released if we exercise our legal defeasance, covenant defeasance option or discharge the Indenture.

We conduct our operations almost entirely through our subsidiaries. Accordingly, the Guarantor Group's cash flow and ability to service any guaranteed registered debt securities will depend on the earnings of our subsidiaries and the distribution of those earnings to the Guarantor Group, including the earnings of the non-guarantor subsidiaries, whether by dividends, loans or otherwise. Holders of the guaranteed registered debt securities will have a direct claim only against the Guarantor Group.

The following tables include summarized financial information for the Guarantor Group (in thousands). The information for the Guarantor Group is presented on a combined basis, excluding intercompany balances and transactions between us and the Guarantor Group and excluding investments in and equity in the earnings of non-guarantor subsidiaries. The Guarantor Group's amounts due from, amounts due to, and transactions with non-guarantor subsidiaries have been presented in separate line items. The consolidating schedules are provided in accordance with the reporting requirements of Rule 13-01 under SEC Regulation S-X for the issuer and guarantor subsidiaries.

	As of April 30, 2024	As of October 31, 2023
Current assets (excluding net intercompany receivable from non-guarantor subsidiaries)	\$1,551,209	\$1,440,062
Noncurrent assets	4,536,099	4,490,490
Net intercompany receivable from/ (payable to) non- guarantor subsidiaries	211,323	182,795
Current liabilities (excluding net intercompany payable to non-guarantor subsidiaries)	451,187	531,466
Noncurrent liabilities	2,921,281	2,895,592
Redeemable noncontrolling interests	253,342	252,013
Noncontrolling interests	43,417	37,786

	Six months ended April 30, 2024
Net sales	\$1,519,035
Gross profit	574,891
Operating income	325,100
Net income from consolidated operations	234,932
Net income attributable to HEICO	219,165
	Six months ended April 30, 2024
Intercompany net sales	\$1,249
Intercompany management fee	1,290
Intercompany interest income	4,329
Intercompany dividends	21,945

Forward-Looking Statements

Certain statements in this report constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. All statements contained herein that are not clearly historical in nature may be forward-looking and the words "anticipate," "believe," "expect," "estimate" and similar expressions are generally intended to identify forward-looking statements. Any forward-looking statement contained herein, in press releases, written statements or other documents filed with the Securities and Exchange Commission or in communications and discussions with investors and analysts in the normal course of business through meetings, phone calls and conference calls, concerning our operations, economic performance and financial condition are subject to risks, uncertainties and contingencies. We have based these forward-looking statements on our current expectations and projections about future events. All forward-looking statements involve risks and uncertainties, many of which are beyond our control, which may cause actual results, performance or achievements to differ materially from anticipated results, performance or achievements. Also, forward-looking statements are based upon management's estimates of fair values and of future costs, using currently available information. Therefore, actual results may differ materially from those expressed in or implied by those forward-looking statements. Factors that could cause such differences include:

- The severity, magnitude and duration of public health threats, such as the COVID-19 pandemic;
- Our liquidity and the amount and timing of cash generation;
- Lower commercial air travel, airline fleet changes or airline purchasing decisions, which could cause lower demand for our goods and services;

- Product specification costs and requirements, which could cause an increase to our costs to complete contracts;
- Governmental and regulatory demands, export policies and restrictions, reductions in defense, space or homeland security spending by U.S. and/or foreign customers or competition from existing and new competitors, which could reduce our sales;
- Our ability to introduce new products and services at profitable pricing levels, which could reduce our sales or sales growth;
- Product development or manufacturing difficulties, which could increase our product development and manufacturing costs and delay sales;
- Cyber security events or other disruptions of our information technology systems could adversely affect our business; and
- Our ability to make acquisitions, including obtaining any applicable domestic and/or foreign governmental approvals, and achieve operating synergies from acquired businesses; customer credit risk; interest, foreign currency exchange and income tax rates; and economic conditions, including the effects of inflation, within and outside of the aviation, defense, space, medical, telecommunications and electronics industries, which could negatively impact our costs and revenues.

For further information on these and other factors that potentially could materially affect our financial results, see Item 1A, *Risk Factors*, of our Annual Report on Form 10-K for the year ended October 31, 2023. We undertake no obligation to publicly update or revise any forward-looking statement, whether as a result of new information, future events or otherwise, except to the extent required by applicable law.

Item 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

There have not been any material changes in our assessment of HEICO's sensitivity to market risk that was disclosed in Item 7A, "Quantitative and Qualitative Disclosures About Market Risk," in our Annual Report on Form 10-K for the year ended October 31, 2023.

Item 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

Our management, with the participation of our Chief Executive Officer and our Chief Financial Officer, evaluated the effectiveness of our disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) as of the end of the period covered by this quarterly report. Based upon that evaluation, our Chief Executive Officer and our Chief Financial Officer concluded that HEICO's disclosure controls and procedures are effective as of the end of the period covered by this quarterly report.

Changes in Internal Control Over Financial Reporting

There have been no changes in our internal control over financial reporting during the second quarter ended April 30, 2024 that have materially affected, or are reasonably likely to materially affect, HEICO's internal control over financial reporting.

PART II. OTHER INFORMATION

Item 5. Other Events.

None of our directors or officers adopted, modified or terminated a "Rule 10b5-1 trading arrangement" or "non-Rule 10b5-1 trading arrangement," as each term is defined in Item 408(a) of Regulation S-K, during the second quarter ended April 30, 2024.

Item 6. EXHIBITS

Exhibit Description

- 10.1 <u>HEICO Savings and Investment Plan, as amended and restated effective as of January 1, 2024.</u> *
- 22.1 <u>Subsidiary Guarantors and Issuers of Guaranteed Securities, is incorporated by</u> reference to Exhibit 22.1 to the Form 10-K for the year ended October 31, 2023. ***
- 31.1 <u>Rule 13a-14(a)/15d-14(a) Certification of Chief Executive Officer.</u> *
- 31.2 <u>Rule 13a-14(a)/15d-14(a) Certification of Chief Financial Officer.</u> *
- 32.1 <u>Section 1350 Certification of Chief Executive Officer.</u> **
- 32.2 <u>Section 1350 Certification of Chief Financial Officer.</u> **
- 101.INS Inline XBRL Instance Document The instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL Document. *
- 101.SCH Inline XBRL Taxonomy Extension Schema Document. *
- 101.CAL Inline XBRL Taxonomy Extension Calculation Linkbase Document. *
- 101.DEF Inline XBRL Taxonomy Extension Definition Linkbase Document. *
- 101.LAB Inline XBRL Taxonomy Extension Labels Linkbase Document. *
- 101.PRE Inline XBRL Taxonomy Extension Presentation Linkbase Document. *
- 104 Cover Page Interactive Data File (formatted as inline XBRL and contained in Exhibit 101). *

- ** Furnished herewith.
- *** Previously filed.

^{*} Filed herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

HEICO CORPORATION

Date: May 30, 2024

- By: <u>/s/ CARLOS L. MACAU, JR.</u> Carlos L. Macau, Jr. Executive Vice President - Chief Financial Officer and Treasurer (Principal Financial Officer)
- By: <u>/s/ STEVEN M. WALKER</u> Steven M. Walker Chief Accounting Officer and Assistant Treasurer (Principal Accounting Officer)

Exhibit 10.1

HEICO Savings and Investment Plan

As Amended and Restated Effective January 1, 2024

Table of Contents

Article I -	Establishment of Plan	1
1.1	Establishment and History	1
1.2	Qualified Plan	1
1.3	Employee Stock Ownership Plan (ESOP)	1
Article II -	Definitions	2
Article III	- Eligibility and Participation	
3.1	Participation in the Plan	
3.2	Eligibility Upon Reemployment	
3.3	Termination of Participation	
3.4	Beneficiary Designation	
Article IV - Contributions and Allocations		
4.1	Employee Contributions	
4.2	Employer Contributions	
4.3	Rollover Contributions	
4.4	Actual Deferral Percentage Test	
4.5	Reductions During Plan Year	
4.6	Return of Excess Contributions After End of Plan Year	
4.7	Actual Contribution Percentage Test	
4.8	Return of Excess Aggregate Contributions	
4.9	Distribution of Excess Deferrals	
4.10	Maximum Annual Additions	
4.11	Return of Contributions to Employer	
4.12	Rights of Reemployed Veterans	
Article V	- Maintenance and Valuation of Accounts	
5.1	Establishment of Accounts	
5.2	Valuation of Accounts	
	- ESOP Provisions and Investment of Contributions	
6.1	Purpose and Nature of the ESOP	
6.2	Allocations and Accounting	
6.3	Company Stock Cash Dividends	
6.4	Diversification Election	
6.5	Voting and Tendering of Company Stock	
6.6	Investment Funds	
6.7	Investment of Participant Accounts	
6.8	Changing Investment Elections – Future Contributions	
6.9	Transfer Among Investment Funds	
	I - Vesting	
7.1	Full Vesting in Designated Accounts	
7.2	Vesting in Matching Contributions and Equity Builder Contributions	
7.3	Forfeiture of Nonvested Interest	
7.4	Restoration of Forfeitures and Service	
Article VIII - Withdrawals and Loans During Employment		
8.1	General Rules Applicable to All In-Service Withdrawals	
8.2	Hardship Withdrawals	
8.3	Withdrawals of Rollover Contributions from the ACT Plan	
8.4	Withdrawals Upon Attainment of Age 59 ¹ / ₂	
8.5	Qualified Reservist Withdrawals	41

8.6	Loans to Participants	41
Article IX	- Distributions Upon Severance from Employment	45
9.1	Eligibility for Distribution	45
9.2	Forms of Payment	45
9.3	Timing of Payment	45
9.4	Minimum Distribution Requirements	46
9.5	Special Timing Rules	49
9.6	Proof of Death	49
9.7	Direct Rollovers	49
9.8	Ordering Rules for Distributions	51
Article X - Top Heavy Provisions		
10.1	When Applicable	52
10.2	Top Heavy Determination	52
10.3	Minimum Contribution	53
10.4	Vesting Rules	54
10.5	Dual Plan Special Limitations	54
10.6	Aggregation Groups	54
Article XI	- Administration of Plan	56
11.1	Committee	56
11.2	Powers and Duties	56
11.3	Compensation and Expenses	
11.4	Company Approval of Equity Builder Contributions	
11.5	Claims Procedure	
11.6	Bonding of Fiduciaries	
11.7	Standard of Conduct	
	I - Management of Funds	
12.1	Appointment of Trustees	
12.2	Investment of Trust Fund by Trustees	
12.3	Investment of Trust Fund by Investment Manager	
12.4	Exclusive Benefit Rule	
	II - Amendment, Merger, Termination of Plan	
13.1	Amendment of Plan	
13.2	Merger or Consolidation	
13.3	Participating Employers	
13.4	Termination of Plan	
	V - Miscellaneous Provisions	
14.1	Limitation of Liability	
14.2		
14.3	Compliance with ERISA	
14.4	Nonalienation of Benefits	
14.5	Employment Not Guaranteed By Plan	
14.6	Form of Communication	
14.7	Facility of Payment	
14.8	Reduction for Overpayment	
14.9	Unclaimed Benefits	
14.10	Payments to Minors and Incompetents	
14.11	Reliance on Information Provided to the Plan	
14.12	Service in More Than One Fiduciary Capacity	
14.13	Binding Effect of Company's Actions	72

14.14	Military Service	73
14.15	Limitation of Rights	73
14.16	Limitation of Third-Party Rights	73
14.17	Invalid Provisions	73
14.18	One Plan	73
14.19	Use and Form of Words	73
14.20	Headings	74
14.21	Authorization	74
14.22	Governing Law	74
14.23	Protected Benefits	74
14.24	Receipt and Release	74
14.25	Forum and Venue	74
Exhibit 1		76
Participating Employers		76

Article I - Establishment of Plan

1.1 Establishment and History

The HEICO Savings and Investment Plan (the "Plan") was established by HEICO Corporation ("HEICO") as of January 1, 1985, for the benefit of employees of HEICO and participating Affiliates. The Plan was last restated in its entirety effective as of March 30, 2022, and is now being further amended and restated effective as of January 1, 2024, (except for those sections of the Plan that have an alternate effective date).

1.2 Qualified Plan

The Plan is intended to be a qualified plan pursuant to the provisions of Section 401(a) of the Internal Revenue Code ("Code") and the related Trust shall be tax-exempt pursuant to the provisions of Section 501(a) of the Code.

1.3 Employee Stock Ownership Plan (ESOP)

In addition, the Plan is intended to be an employee stock ownership plan within the meaning of Section 4975(e)(7) of the Code and is designed to invest primarily in Company Stock and is a stock bonus plan within the meaning of Regulation Section 1.401-1(b)(1)(iii). The portion of the Plan that is an ESOP is no longer a leveraged ESOP.

Article II - Definitions

When used herein the following terms shall have the following meanings:

"Account" means the bookkeeping account or accounts, including any subaccounts, established to reflect contributions made to the Plan by or on behalf of a Participant, and adjustments made in accordance with Article V.

"**ACT Plan**" means the Aerospace & Commercial Technologies LLC Retirement Plan that was merged into the Plan on March 30, 2022.

"Actual Contribution Percentage" means for a specified group of Participants (either Highly Compensated Employees or Non-highly Compensated Employees) for a Plan Year, the average of the Contribution Percentages of the Eligible Participants in the group. For this purpose, Eligible Participants means any employee who is eligible to make Elective Deferrals (if the Employer takes such contributions into account in the calculation of the Contribution Percentage), or to receive a Matching Contribution (including forfeitures) or a QMAC.

"Actual Deferral Percentage" means for a specified group of Participants (either Highly Compensated Employees or Non-highly Compensated Employees) for a Plan Year, the average of the ratios (calculated separately for each Participant in the group) of (a) the amount of Employer contributions actually paid to the Trust on behalf of such Participant for the Plan Year to (b) the Participant's compensation for the Plan Year. Compensation, for purposes of this definition shall mean compensation within the meaning of Section 414(s) of the Code for the Plan Year. Employer contributions on behalf of any Participant shall include: (a) any Elective Deferrals (other than Catch- Up Contributions) made pursuant to the Participant's deferral election (including Excess Deferrals of Highly Compensated Employees), but excluding (i) Excess Deferrals of Non-highly Compensated Employees that arise solely from Elective Deferrals made under the Plan or plans of the Employer and

(ii) Elective Deferrals that are taken into account in the Contribution Percentage test (provided the Actual Deferral Percentage test is satisfied both with and without these Deferral Contributions), and
(b) any QNECs or QMACs that are taken into account in the Actual Deferral Percentage Test. For purposes of calculating Actual Deferral Percentages, an Eligible Employee who would be a Participant but for the failure to make Elective Deferrals shall be treated as a Participant on whose behalf no Elective Deferrals are made.

"Adjustment Factor" means the cost-of-living adjustment factor prescribed by the Secretary of the Treasury under Section 415(d) of the Code, as applied to such items and in such manner as the Secretary shall provide.

"Administrator" means the Committee, or such other persons or entities designated by the Committee to administer the Plan pursuant to Article XI.

"Affiliate" means an employer who is a corporation, trade, or business (whether incorporated or not), or organization under common control with the Employer under Section 414(b), (c), (m), or (o) of the Code and determined in accordance with the applicable guidance under Treasury Regulation Section 1.414(c)-5 and any other applicable guidance issued by the Internal Revenue Service.

"Annual Addition" means, for any Limitation Year, the sum of the following amounts credited to a Participant under the Plan:

- (a) Employer Contributions;
- (b) Elective Deferrals (other than Catch-Up Contributions as defined Section 414(v) of the Code and contributions that have been distributed to the Participant as Excess Deferrals);
- (c) forfeitures allocated to the Participant's Account;
- (d) amounts allocated to an individual medical account, as defined in Section 415(I)(2) of the Code, which is part of a pension or annuity plan, and amounts derived from contributions paid or accrued which are attributable to post-retirement medical benefits, allocated to the separate account of a key employee, as defined in Section 419A(d)(3) of the Code, under a welfare benefit fund, as defined in Section 419(e) of the Code; and
- (e) allocations under a simplified employee pension.

Amounts described in (a), (b), (c), and (e) are annual additions for purposes of both the dollar limitation under Section 4.10 and the percentage of compensation limitation under Section 4.10. Amounts described in (d) are annual additions solely for purposes of the dollar limitation under Section 4.10.

"Authorized Leave of Absence" means the period during which a Participant is absent without compensation and for which the Committee, in its sole discretion, has determined that the Participant is on a leave of absence rather than having incurred a Severance from Employment. The discretion of the Committee shall be exercised in a nondiscriminatory manner. In all events, a leave of absence by reason of service in the armed forces of the United States shall end no later than the time at which a Participant's re-employment rights as a member of the armed forces cease to be protected by law. A leave of absence for any other reason shall end after six months, except that if the Participant has resumed employment prior thereto, the leave of absence shall end on the date on which employment is resumed. If the Participant does not resume employment by the date on which the leave of absence ends, such date shall be considered the date on which the Participant incurred a Severance from Employment. In determining a Year of Service for Accrual of Benefits, all such leaves of absence shall be considered periods when the Employee is a Participant.

"Average Contribution Percentage" means the average (expressed as a percentage) of the Contribution Percentages of the Eligible Employees in a group.

"**Beneficiary**" means the beneficiary or beneficiaries designated under to Section 3.4 to receive the amount, if any, payable under the Plan upon the death of a Participant.

"Board of Directors" means the Board of Directors of HEICO Corporation.

"**Break in Service**" means a Plan Year during which the Participant does not complete more than 500 Hours of Service.

"**Catch-Up Contribution**" means an Elective Deferral made to the Plan that is not includible in the gross income of the Participant under Section 4.1(c) and earnings or losses on those contributions.

"Code" means the Internal Revenue Code of 1986, as now in effect or hereafter amended.

"Committee" means the Committee appointed by the Board of Directors under Article XI to administer the Plan. The Committee shall be a named fiduciary (as defined in ERISA Section 402(a)(2)).

"**Company**" means HEICO Corporation or any successor by merger, consolidation, purchase of substantially all of its assets or otherwise.

"Company Stock" means the common stock of HEICO Corporation that also meets the requirements to be Qualifying Employer Securities. Both HEICO Corporation Common Stock and HEICO Corporation Class A Common Stock shall be considered Company Stock for purposes of this Plan.

"Company Stock Fund" means the fund established by the Trustee to hold Company Stock.

"415 Compensation" means wages within the meaning of Section 3401(a) of the Code and all other payments of compensation to an Employee by the Aggregated Employer (in the course of Employer's trade or business) for which the Employer is required to furnish the Employee a written statement under Sections 6041(d), 6051(a)(3), and 6052 of the Code. Compensation must be determined without regard to any rules under Section 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in Section 3401(a)(2) of the Code).

- (a) Except as otherwise provided in Regulation Section 1.415(c)-2(e), compensation for a Limitation Year is the compensation actually paid or made available during such Limitation Year. Compensation paid or made available during a Limitation Year shall include any Elective Deferral as defined in Section 402(g)(3) of the Code and any amount which is contributed or deferred by the Employer at the election of the Participant and which is not includible in the gross income of the Participant by reason of Section 125, 132(f)(4), or 457 of the Code. Any differential wage payment within the meaning of Section 3401(h) of the Code shall be included as compensation.
- (b) Any payments made after Severance from Employment will not be considered compensation, unless the payment may be treated as such in accordance with Regulation Section 1.415(c)-2(e)(3).

"Contribution Percentage" means the ratio (expressed as a percentage) of the Matching Contributions and After-Tax Contributions made under the Plan on behalf of the Participant for the Plan Year to the Participant's compensation for the Plan Year. Compensation, for purposes of this definition, shall mean compensation within the meaning of Section 414(s) of the Code for the Plan Year received from the Employer. For this purpose, Matching Contributions that are forfeited because the contributions to which they relate are Excess Deferrals or Excess Aggregate Contributions shall not be included.

"**Disability**" means the Participant has been determined to be totally disabled by the Social Security Administration.

"Early Retirement Date" means the later of (1) an Employee's 55th birthday or (2) the date on which he or she completes 10 Years of Vesting Service.

"Effective Date" means January 1, 2024, the date as of which the Plan, as amended and restated, is effective, except as otherwise noted herein. The Plan's original effective date is January 1, 1985.

"Elective Deferrals" means a Participant's Pre-Tax Contributions (including automatic contributions under Section 4.1(a)), Roth Contributions, Catch-Up Contributions, and Roth Catch-Up Contributions which the Employer contributes to the Plan at the Participant's election in lieu of cash Eligible Compensation.

"**Eligible Compensation**" means with respect to each Participant for purposes of calculating and allocating contributions to the Plan the total amount of a Participant's base salary or wages, including overtime pay, paid to him or her by his or her Employer.

Eligible Compensation shall include Employee Contributions described in Section 414(h)(2) of the Code, elective deferrals and any amount which is contributed by the Employer pursuant to a salary reduction agreement which is not includable in the gross income of the Employee under Sections 125, 402(g)(3), 403(b), 132(f)(4), or 457(b) of the Code.

Eligible Compensation shall not include commissions, bonuses, incentive compensation, any amount paid in lieu of vacation days and all other items of extraordinary compensation reportable as taxable wage.

The annual Eligible Compensation of each Participant taken into account for all Plan purposes shall not exceed \$345,000, as adjusted by the Secretary of the Treasury for increases in the cost of living in accordance with Section 401(a)(17)(B) of the Code. The cost-of-living adjustment in effect for a calendar year applies to any period, not exceeding 12 months, over which Eligible Compensation is determined (the "determination period") beginning in such calendar year. If a determination period consists of fewer than 12 months, the limit referred to above will be multiplied by a fraction, the numerator of which is the number of months in the determination period and the denominator of which is 12.

For other specific purposes described in the Plan, "Compensation" or "compensation" shall have the meanings set forth in the respective sections in which the term is referenced.

"Eligible Employee" means any Employee of the Employer other than: (a) an Employee who is employed by an Affiliate that is not a Participating Employer; (b) an Employee who is a nonresident alien deriving no earned income from the Employer which constitutes income from sources within the United States; (c) any person who is included in a unit of employees covered by an agreement recognized for purposes of collective bargaining with the Employer, provided retirement benefits have been the subject of good faith bargaining and such bargaining does not provide for coverage under the Plan; and (d) an Employee classified by the Employer as a temporary, seasonal or occasional employee, provided, however, that a temporary, seasonal or occasional employee who completes a Year of Vesting Service shall be an Eligible Employee. An Employee who completes three consecutive 12-month periods, beginning on or after January 1, 2021, during each of which the Employee completes 500 Hours of Service, shall be an Eligible Employee for purposes of making Elective Deferral Contributions. Each such Employee must otherwise be an Eligible Employee in order receive Employer Contributions. Each 12-month period for which an Employee defined in this subsection (e) has at least 500 Hours of Service shall be treated as a Year of Vesting Service.

Notwithstanding any other provision of the Plan, the term 'Eligible Employee' shall not include any Employee or other individual unless such individual is contemporaneously treated by the Employer as an Employee for purposes of the Plan (without regard to any subsequent recharacterization or inconsistent determination made by any person or entity or by any court, agency, or other authority with respect to such individual whenever effective).

"**Employee**" means any person employed by the Employer (including any Affiliate) as a common law employee. An Employee shall not include an Independent Contractor or a Leased Employee within the meaning of Section 414(n) of the Code. Employee shall also mean a person who (a) is an individual treated as an employee within the meaning of Section 414(o) of the Code and the Regulations thereunder; (b) is an Employee on an Authorized Leave of Absence; or (c) is receiving differential wage payments as defined in Section 3401(h) of the Code from an Employer.

"Employer" means the Company or any Participating Employer.

"Employer Contributions" means the amounts contributed by the Employer under Section 4.2.

"**Employment Commencement Date**" means the date on which an Employee is first credited with an Hour of Service.

"Entry Date" means each day of the Plan Year.

"Equity Builder Contributions" means the contributions that are made by an Employer on behalf of a Participant and are designated as such pursuant to Section 4.2(b).

"Equity Builder Contributions Account" means the Account to which are credited any Equity Builder Contributions made on behalf of the Participant and earnings or losses on those contributions. "ERISA" means the Employee Retirement Income Security Act of 1974, as now in effect or as hereafter amended.

"ESOP" means the Employee Stock Ownership Plan established as part of Article VI.

"Excess Aggregate Contributions" means Matching Contributions in excess of the Contribution Percentage limit, as described in Section 401(m)(6)(B) of the Code.

"Excess Contributions" means, for any Plan Year, the excess of:

- (a) The aggregate amount of Employer Contributions actually taken into account in computing the Actual Deferral Percentage of Highly Compensated Employees for the Plan Year, over
- (b) The maximum amount of the contribution permitted by the Actual Deferral Percentage test (determined by hypothetically reducing contributions made on behalf of Highly Compensated Employees in order or the Actual Deferral Percentages, beginning with the highest of the percentages.

"**Excess Deferrals**" means Elective Deferral contributions of a Participant that exceed the dollar limits imposed by Section 402(g) of the Code.

"Highly Compensated Employee" means any Employee who performs services for the Employer, including any Affiliates, during the determination year and who was a 5% owner, as defined in Section 416(i)(1) of the Code during the determination year or look-back year, or during the look-back year received Compensation from the Employer, including any Affiliates, in excess of the limit in Section 414(q)(1)(B) of the Code (\$150,000 for the 2023 look-back year), multiplied by the Adjustment Factor for the look-back year.

- (a) For purposes of this definition, the determination year is the Plan Year; the look-back year is the 12 month period preceding the Plan Year.
- (b) A highly compensated former Employee shall be treated as a Highly Compensated Employee if he or she separated from service (or is deemed to have separated) prior to the determination year, performs no service for the Employer, including any Affiliates, during the determination year and was a highly compensated active Employee for either the separation year or any determination year ending on or after the Employee's 55th birthday.
- (c) The determination of who is a Highly Compensated Employee, including the determination of the Compensation that is considered, will be made in accordance with Section 414(q) of the Code and the Regulations thereunder.
- (d) For purposes of the definition of Highly Compensated Employee, "Compensation" shall mean all Compensation within the meaning of Section 414(s) of the Code, including Elective Deferrals that are not includible in the gross income of the Employee under Sections 125, 132(f)(4), 402(e)(3), 402(h), or 403(b) of the Code.

"Hour of Service" means the hours of employment with the Employer, as follows:

- (a) <u>Hours Paid and Worked.</u> Each hour for which an Employee is directly or indirectly paid or entitled to payment for the performance of duties for the Employer.
- (b) <u>Hours Directly or Indirectly Paid but not Worked.</u> Each hour for which an Employee is directly or indirectly paid or entitled to payment by the Employer on account of a period during which no duties are performed, whether or not the employment relationship has terminated, due to vacation, holiday, illness, incapacity (including disability), layoff, jury duty, military duty, or leave of absence, but not more than 501 such hours on account of any single continuous period during which no duties are performed.
- (c) <u>Maternity or Paternity Leave.</u> Each hour which would otherwise have been credited to an Employee but for the Employee's absence due to maternity or paternity leave, or in any case in which such hours cannot be determined, eight Hours of Service per day of such absence, to a maximum of 501 Hours of Service. For purposes of this paragraph, "maternity or paternity leave" means a leave of absence (i) by reason of the pregnancy of the individual, (ii) by reason of the birth of a child of the individual, (iii) by reason of the placement of a child with the individual in connection with the adoption of such child by the individual, or (iv) for purposes of caring for a child for the period beginning immediately following such birth or placement.
- (d) <u>Back Pay.</u> Each hour for which back pay, irrespective of mitigation of damages, has been awarded or agreed to by the Employer. Any such hours shall be credited to the Employee for the computation period to which the award or agreement pertains and not the computation period in which the award, agreement or payment is made.
- (e) <u>Unemployment Compensation.</u> No hours shall be credited on account of any period during which an Employee performs no duties and receives payment solely for the purpose of reimbursement for medical or medically related expenses incurred by the Employee for the purpose of complying with applicable worker's compensation, unemployment compensation or disability insurance laws.
- (f) <u>Unpaid Leave of Absence.</u> Solely for purposes of determining whether an Employee has Break in Service, an Employee shall be credited with Hours of Service while on a leave of absence protected under the Family Medical Leave Act of 1993 (FMLA), provided the Employee returns to work.
- (g) The same Hours of Service shall not be credited under more than one of the above clauses (a), (b) or (c); and each hour credited to an Employee under clause (a), (b) or (c) above shall be credited in accordance with Section 2530.200b-2(b) and (c) of the U.S. Department of Labor's Regulations, which hereby are incorporated by reference.

(h) Hours of Service shall be credited for any individual considered an Employee under Section 414(n) or Section 414(o) of the Code and the related Regulations. Pursuant to Section 414(n)(4)(B) of the Code, Hours of Service shall be determined by taking into account any period for which an Employee would have been a Leased Employee but for the fact that the Employee failed to perform services for the Employer on a substantially full-time basis for a period of at least one year.

"Independent Contractor" means any individual who is a signatory to a contract, letter of agreement, or other document that acknowledges his or her status as an independent contractor not entitled to benefits under the Plan or any individual who is not otherwise classified by the Employer as a common law employee, even if such independent contractor or other individual is later determined to be a common law employee.

"Investment Fund" means the investment funds established under Article VI, or any of them individually.

"Leased Employee" means an individual who provides services to the Employer (the "recipient") but is not otherwise an employee of the recipient if (a) such services are provided pursuant to an agreement between the recipient and any other person (the "leasing organization"), (b) such individual has performed services for the recipient on a substantially full-time basis for at least one year, and (c) such services are performed under the primary direction of or control by the recipient. The determination of who is a Leased Employee shall be made in accordance with the Regulations.

"Limitation Year" means the calendar year, unless otherwise selected by the Company in a manner consistent with that described in Section 1.415-2(b)(2) of the Regulations.

"Matching Contributions" means Employer contributions made to the Plan under Section 4.2(a) on account of a Participant's Elective Deferrals.

"Matching Contributions Account" means the Account to which are credited any Matching Contributions made on behalf of the Participant and earnings or losses on those contributions.

"**Merged Plan**" means a qualified defined contribution plan under Section 401(a) of the Code that has been merged into the Plan.

"Non-highly Compensated Employee" means an Employee who is not a Highly Compensated Employee.

"Normal Retirement Age" means the attainment of age 65.

"Participant" means any Eligible Employee participating in the Plan as provided in Article III.

"**Participating Employer**" means an Affiliate that with the approval of the Committee adopts this Plan pursuant to Section 13.3, as listed on the attached Exhibit 1, as it may be amended from time to time.

"Plan" means the HEICO Savings and Investment Plan, as set forth herein and as amended from time to time.

"**Plan Year**" means the 12-month period commencing on each January 1 and ending on the next following December 31.

"**Pre-Tax Contributions**" means an Elective Deferral contribution made by the Employer on behalf of the Participant that is not subject to income tax when made under Sections 4.1(a) or 4.1(b).

"**Pre-Tax Contributions Account**" means the Account to which are credited Pre-Tax Contributions and Catch-Up Contributions made on behalf of a Participant and the earnings or losses on those contributions.

"QMAC" means a qualified matching contribution made to the Plan under Section 4.4(f).

"QMAC Account" means the Account to which QMACs and any earnings or losses on those contributions are credited on behalf of a Participant under Section 4.4(f).

"QNEC" means a qualified non-elective contribution as defined in Regulation Section 1.401(m)-5 that is made to the Plan under Section 4.4(f).

"QNEC Account" means the Account to which QNECS and any earnings or losses on those contributions are credited on behalf of a Participant under Section 4.4(f).

"Qualifying Employer Securities" means the common stock issued by the Employer (or by a corporation which is a member of the same controlled group) which is readily tradable on an established securities market in accordance with, or otherwise qualifies under, Section 409(I) of the Code. Noncallable preferred stock shall be treated as Qualifying Employer Securities if such stock is convertible at any time into stock which meets the requirements of the previous sentence and if such conversion is at a conversion price which (as of the date of the acquisition by the tax credit employee stock ownership plan) is reasonable under Section 409(I)(3) of the code. A security is "readily tradable on an established securities market" if the security is traded on a national securities exchange that is registered under section 6 of the Securities Exchange Act of 1934 (15 U.S.C. 78f) or the security is traded on a foreign national securities exchange that is officially recognized, sanctioned, or supervised by a governmental authority and the security is deemed by the Securities and Exchange Commission as having a "ready market" under SEC Rule 15e3-1 (17 CFR 240.15e3-1).

"Reemployment Commencement Date" means the first date following a Break in Service on which the Eligible Employee again performs an Hour of Service.

"**Regulations**" means the Treasury regulations issued under the Code or any other applicable law by the Internal Revenue Service and any proposed or temporary regulations or rules pending the issuance of such regulations. "**Required Beginning Date Age**" means age 70½ if the Participant was born prior to July 1, 1949; age 72 if the participant was born on or after July 1, 1949, but prior to January 1, 1951; age 73 if born on or after January 1, 1951, but before January 1, 1960; and age 75 if born on or after January 1, 1960.

"Rollover Contribution" means a contribution made by a Participant under Section 4.3(a).

"Rollover Contributions Account" means the Participant's Account to which is credited any Rollover Contribution made by the Participant and earnings or losses on that contribution.

"**Roth Catch-Up Contributions**" means an Elective Deferral made to the Plan that is includible in the gross income of the Participant under Section 4.1(d) and earnings or losses on those contributions.

"**Roth Contributions**" means an Elective Deferral made to the Plan that is includible in the gross income of the Participant under Section 4.1(d).

"**Roth Contributions Account**" means the Participant's Account to which are credited any Roth Contributions and Roth Catch-Up Contributions made by the Participant and earnings or losses on those contributions.

"Roth Rollover Contributions" means the contributions made by a Participant under Section 4.3(b).

"Roth Rollover Contributions Account" means the Participant's Account to which are credited any Roth Rollover Contributions made by the Participant and earnings or losses on those contributions.

"Severance from Employment" means the termination of the Employee's employment relationship with the Employer for any reason, including death, Disability, or retirement. The transfer of employment among Affiliates shall not be considered a Severance from Employment for any reason. An Employee receiving differential wage payments within the meaning of Section 3401(h) of the Code shall be treated as not having incurred a Severance from Employment during the period the payments are being made.

"Spouse" means the person to whom a Participant is legally married.

"Trust Agreement" means the agreement entered into between the Company and the Trustee to carry out the purposes of the Plan, as amended from time to time.

"Trust Fund" means the assets of the Plan held in trust by the Trustee in accordance with the Trust Agreement.

"Trustee" means the trustee or trustees by whom the assets of the Plan are held in accordance with the Trust Agreement.

"Valuation Date" means any business day on which the New York Stock Exchange is open for and conducting business, or any more frequent date designated by the Committee.

"Year of Vesting Service" means a Plan Year for which an Employee is credited with at least 1,000 Hours of Service with the Employer or any Affiliate. If so, provided in a purchase agreement, an Employee will also be credited with Hours of Service or Years of Vesting Service for the period of service with the entity whose assets are being acquired by the Company or any Affiliate. In no circumstances shall an Employee be credited with more than one Year of Vesting Service in any calendar year.

Wherever used herein, the singular includes the plural, and the masculine includes the feminine, unless the context clearly requires otherwise.

Article III - Eligibility and Participation

3.1 Participation in the Plan

- (a) An Eligible Employee shall become a Participant on the Entry Date he or she first performs an Hour of Service. With respect to a Participant in the Plan on January 1, 2024, such Participant shall remain a Participant in the Plan on the Effective Date.
- (b) With respect to individuals who become an Employee of an Employer as a result of an acquisition of another business, an Employee who is an otherwise Eligible Employee shall be treated as having performed an Hour of Service as of the date of the Employer's acquisition of such business unless a different dated is required by law, provided for by the Committee or provided for in any agreement related to the acquisition of such business.

3.2 Eligibility Upon Reemployment

Any person reemployed as an Eligible Employee shall become a Participant on his or her Reemployment Commencement Date.

3.3 Termination of Participation

A Participant who incurs a Severance from Employment or ceases to be an Eligible Employee without incurring a Severance from Employment shall continue to be a Participant in the Plan but shall not be eligible to make Elective Deferrals to the Plan or receive Employer Contributions. A Participant's participation shall cease upon distribution to him or her of his or her entire vested Account or upon his or her death prior to such distribution.

3.4 Beneficiary Designation

Each Participant may designate as a primary and/or contingent Beneficiary any person, persons, or entity to receive benefits payable upon his or her death. A Participant may also at any time revoke or change his or her Beneficiary designation by filing a new designation. Any such designation, revocation, or change in Beneficiary shall be made in in writing or electronically in accordance with the procedures of this Section 3.4 and those procedures established by the Committee.

- (a) Any Beneficiary designation, revocation, or change thereof naming as a primary Beneficiary a person, persons or entity other than the Participant's Spouse must be made with the written consent of the Participant's Spouse acknowledging the effect of such designation, revocation or change and witnessed by a notary public. Written consent of the Participant's Spouse shall not be required if it is established to the satisfaction of the Committee, and that such provision is applied on a uniform and nondiscriminatory basis, that there is no Spouse, the Spouse cannot be located or under other circumstances as may be prescribed in the Regulations.
- (b) If at the date of death of the Participant, there is no valid and current Beneficiary designation on file, or if no Beneficiary is living, then the death benefit which would have been payable to the Beneficiary shall be paid to the Participant's surviving Spouse, or if none, to the Participant's estate.
- (c) The interpretation of the Committee with respect to any Beneficiary designation, subject to applicable law, shall be binding and conclusive upon all parties, and no person who claims to be a Beneficiary, or any other person, shall have the right to question any action of the Committee, which in the judgment of the Committee fulfills the intent of the Participant who filed such designation.

Article IV - Contributions and Allocations

4.1 Employee Contributions

- (a) Automatic Enrollment of Participants.
 - (i) <u>Automatic Enrollment of New Employees.</u> Each Eligible Employee who is eligible to make a Deferral Contribution and does not elect to defer a percentage of his or her Eligible Compensation under Section 4.1(b) or (d), shall be deemed to have elected to have 3% of his or her subsequent Eligible Compensation reduced by means of payroll deduction and contributed to the Plan as a Pre-Tax Contribution effective as of the first payroll period that is (A) administratively practicable and (B) is at least 30 days after he or she is given notice of the deemed election as described in Section 4.1(a)(iii). A Participant who affirmatively elects a deferral percentage of 0% shall be deemed to have made an election and not be subject to the provisions of this subparagraph (a)(i).
 - (ii) <u>Withdrawal of Contributions.</u> A Participant who is automatically enrolled under the provisions of (i) above, shall be provided with a 90-day period, commencing on the first day on which his or her participation begins, to elect out of the Plan and in accordance with the provisions of Section 414(w) of the Code, withdraw such contributions made on his or her behalf as well as the earnings thereon.
 - (iii) <u>Notices.</u> Prior to the date that an Eligible Employee is actually enrolled under Section 4.1(a)(i), he or she shall be provided with (A) an effective opportunity to elect to have a different percentage, including 0% of his or her Eligible Compensation, contributed to the Plan and (B) a notice of the deemed election and his or her ability to elect to have a different percentage, including 0% of his or her Eligible Compensation contributed to the Plan.
- (b) Pre-Tax Contributions. Alternatively, an Eligible Employee who meets the requirements of Sections 3.1 or 3.2 may, by advance notice in accordance with applicable administrative procedures, elect to have his or her subsequent Eligible Compensation reduced by means of payroll deduction as of any subsequent payroll and to have an equal amount contributed to the Plan on his or her behalf as Pre-Tax Contributions. The deduction shall commence effective with the first payroll period that begins as soon as administratively practicable thereafter. Except for occasional, bona fide administrative considerations, a Pre-Tax Contribution made in accordance with such an election cannot precede the earlier of: (a) the performance of services relating to the Pre-Tax Contribution, and (b) the date the Eligible Compensation that is the subject of the election would be currently available to the Eligible Employee in the absence of the election to defer. A Participant may elect to make Pre-Tax Contributions in .10% increments of up to 80% of his or her Eligible Compensation, such percentage to be reduced by any amount contributed under Section 4.1(c), (d), or (e). With respect to an Eligible Employee who is a Participant on the Effective Date, such Participant's deferral election in effect prior to January 1, 2024, shall continue to apply until changed by a Participant in accordance with the provisions of this Section 4.1(b).

- (c) <u>Catch-Up Contributions.</u> Each Participant who has attained age 50 before the close of the Plan Year shall be eligible to make Catch-Up Contributions of up to 80% of his or her Eligible Compensation in accordance with, and subject to, the limitations of Section 414(v) of the Code, such percentage to be reduced by any amount contributed under Section 4.1(a), (b), or (d). Such Catch-Up Contributions shall not be taken into account for purposes of the provisions of the Plan implementing the required limitations of Sections 402(g) and 415 of the Code. Catch-Up Contributions shall be made in accordance with procedures established by the Committee and shall constitute Elective Deferrals.
- (d) <u>Roth Contributions.</u> Unless a Participant makes an election under the provisions of this paragraph (d), contributions made under paragraphs (a), (b) and (c) above shall be made on a pre-tax basis.

In addition to and/or instead of making such contributions on a pre-tax basis under paragraphs (b) and (c) above, a Participant may elect in accordance with applicable administrative procedures to have some, or all, of the contributions that would otherwise be contributed to the Plan on a pre-tax basis, designated on an irrevocable basis at the time of deferral, as Roth Contributions or Roth Catch-Up Contributions and be included in the Participant's gross income at the time of deferral. Such election, once made, shall only be revoked with respect to contributions to be contributed after the effective date of the revocation election. Roth Contributions and Roth Catch-Up Contributions shall be treated in the same manner as those made on a pre-tax basis for all Plan purposes, except as provided herein. A Roth Contribution or a Roth Catch-Up Contribution shall be separately accounted for, as well as any gains or losses, in a Roth Contributions Account. No contributions other than Roth Contributions or Roth Catch-Up Contributions, including forfeitures, may be allocated to such Roth Contributions Account. A Participant may elect to make Roth Contributions and/or Roth Catch-Up Contributions in .10% increments of up to 80% of his or her Eligible Compensation, such percentage to be reduced by any amount contributed under Section 4.1(a), (b), or (c).

- (e) <u>After-Tax Contributions.</u> A Participant is not permitted to make after-tax contributions.
- (f) <u>Aggregate Limitation.</u> In no event will the aggregate of a Participant's Elective Deferrals exceed 80% of a Participant's Eligible Compensation.
- (g) <u>402(g) Limitation.</u> A Participant's Elective Deferrals made under this Plan or any other plan or arrangement maintained by the Employer, during any calendar year will not exceed the dollar limit in effect under Section 402(g) of the Code.
- (h) <u>Changes in a Participant's Eligible Compensation.</u> The percentage of Elective Deferrals designated under Section 4.1(a), (b), (c), or (d) shall automatically apply to increases and decreases in his or her Eligible Compensation. A Participant may, in accordance with applicable administrative procedures, change the percentage of his or her Eligible Compensation to be contributed to the Plan under Section 4.1(a), (b), (c), or (d). The change shall commence effective with the first payroll period that begins as soon as administratively practicable thereafter. The changed percentage shall remain in effect until subsequently changed.

 (i) <u>Suspension</u>. A Participant may, by advance notice in accordance with applicable administrative procedures, elect to suspend his or her Elective Deferrals at any time. The suspension shall commence effective with the next payroll period that begins as soon as administratively practicable thereafter.

A Participant who has suspended his or her Elective Deferrals may, by giving advance notice in accordance with applicable administrative procedures, elect to resume making such contributions as of any payroll period thereafter. The resumption shall commence effective with the first payroll period that begins as soon as administratively practicable thereafter.

- (j) <u>Return of Contributions.</u> If Elective Deferrals are returned to the Employer under Section 4.8, the elections to reduce Eligible Compensation that were made by Participants on whose behalf those contributions were made shall be void retroactively to the beginning of the period for which returned contributions were made.
- (k) <u>Remittance of Contributions.</u> The Employer of the Participant shall forward Elective Deferrals to the Trustee as soon as practicable after the date the compensation otherwise would have been paid to the Participant in conformance with Section 2510.3-102 of the Department of Labor Regulations.
- Ordering Rules for Withdrawals. The Committee may operationally implement an ordering rule procedure for withdrawals from a Participant's Account attributable to Pre-Tax Contributions and Roth Contributions. Such ordering rules may specify which type of Pre-Tax Contributions and Roth Contributions.

4.2 Employer Contributions

- (a) <u>Matching Contributions.</u> For each calendar quarter, the Employer, in its sole discretion, may make a Matching Contribution to a Participant's Matching Contribution Account based on percentage of the Participant's Elective Deferrals. Matching Contributions shall be contributed by the Employer as soon as administratively feasible following the end of each calendar quarter for a Participant who makes Elective Deferrals during that calendar quarter and who is employed on the last day of such calendar quarter.
- (b) Equity Builder Contributions. For each Plan Year, the Employer may, in the sole discretion of the Board of Directors, contribute Equity Builder Contributions. Equity Builder Contributions shall be allocated to each Participant's Account based on the ratio that each Participant's Eligible Compensation for the Plan Year bears to the total Eligible Compensation paid to all Participants eligible for an Equity Builder Contribution for the Plan Year. Only Eligible Compensation paid by the Employer on account of service while an Eligible Employee shall be taken into account for purposes of this subsection (b). In order to receive an Equity Builder Contribution, the Eligible Employee must have completed a 1,000 Hours of Service during the Plan Year as an Employee of an Employer and be employed on the last day of the Plan Year. An Equity Builder Contribution may be made without regard to whether the Company has current or accumulated earnings or profits for the taxable year ending with or within the Plan Year.

(c) <u>Remittance of Employer Contributions.</u> Employer Contributions may be made in cash, Company Stock, or such other property as the Employer shall determine. Except as otherwise provided for in this Section 4.2, Employer Contributions for a Plan Year, including Company Stock shall be forwarded to the Trustee at the discretion of the Employer but not later than the date required by applicable law.

4.3 Rollover Contributions

(a) <u>Rollover Contributions in General.</u> An Eligible Employee who has become a Participant may, by notice received by the Committee and under such terms and conditions as the Committee shall determine, make a Rollover Contribution to the Plan and Trust Fund. The Committee may require the individual to submit such evidence and documentation as the Committee determines necessary to be assured that the proposed contribution qualifies as a Rollover Contribution.

A Rollover Contribution is (a) a distribution that is an "eligible rollover distribution" (as defined in Section 402(c)(4) of the Code) from (i) a qualified plan described in Section 401(a) or 403(a) of the Code, (ii) an annuity contract described in Section 403(b) of the Code, (iii) an eligible plan under Section 457(b) of the Code which is maintained by a state political subdivision of a state, or an agency or instrumentality of a state or political subdivision of a state, or (iv) an individual retirement account or annuity described in Section 408(a) or (b) of the Code, or (b) a direct rollover of an eligible rollover distribution from (i) a qualified plan described in Section 401(a) or 403(a) of the Code including after- tax employee contributions, (ii) an annuity contract described in Section 403(b) of the Code including after-tax employee contributions, (iii) an eligible plan under Section 457(b) of the Code which is maintained by a state. political subdivision of a state, or an agency or instrumentality of a state of political subdivision of a state, or (iv) an individual retirement account or annuity described in Section 408(a) or (b) of the Code. Section 402(c)(9) of the Code shall apply in determining whether a distribution is a Rollover Contribution for purposes of this Section 4.3(a).

- (b) <u>Roth Rollover.</u> In accordance with applicable administrative procedures, an Eligible Employee may request that the Plan accept a Roth Rollover but only in the case it is a direct rollover from another Roth Account of an applicable retirement plan as described in Section 402A(e)(1) of the Code and only to the extent the rollover is permitted under the rules of Section 402(c) of the Code. Any such Roth Rollover accepted under this paragraph (b) shall be accounted for in the Eligible Employee's Roth Rollover Contributions Account.
- (c) <u>Prohibited Rollover Contributions.</u> The Plan will not accept rollover contributions of accumulated deductible employee contributions from a simplified employee pension plan, after tax contributions from an arrangement described in (a) above, including Roth IRAs described in Section 408A(b) of the Code.
- (d) <u>Timing.</u> In the case of a distribution described in (a) or (b) that is not a direct rollover, such distribution must be received by the Plan on or before the 60th day following the Employee's receipt of the distribution from the distributing plan or contract.

(e) <u>Accounting for Rollover Contributions.</u> The amount received under this Section 4.3 shall be transferred to the Plan and Trust Fund and credited to the Participant's Rollover Contributions Account or Roth Rollover Contribution Account, as applicable, in accordance with Article V.

4.4 Actual Deferral Percentage Test

- (a) For the Plan Year, the Average Deferral Percentage for the group of all Highly Compensated Employees who are Eligible Employees must satisfy at least one of the following tests:
 - (i) The Actual Deferral Percentage for said group of Highly Compensated Employees for that Plan Year shall not be more than the Actual Deferral Percentage for the group of Non-highly Compensated Employees who are Eligible Employees for the Plan Year multiplied by 1.25; or
 - (ii) The Actual Deferral Percentage for said group of Highly Compensated Employees for that Plan Year shall not exceed two percentage points more than the Actual Deferral Percentage for the group of Non-highly Compensated Employees who are Eligible Employees for the Plan Year, and the Actual Deferral Percentage for said group of Highly Compensated Employees for that Plan Year shall not be more than the Actual Deferral Percentage for the group of Non-highly Compensated Employees who are Eligible Employees for the Plan Year multiplied by two.
- (b) If Deferral Contributions are made to the Plan for a Plan Year for a Highly Compensated Employee who also is eligible to have salary reduction contributions allocated to his or her account under another plan maintained by the Employer that provides a cash or deferred arrangement described in Section 401(k) of the Code, the "actual deferral ratio" within the meaning of Regulation Section 1.401(k)–2(a)(3) for that Highly Compensated Employee shall be calculated as if all such other plans are part of the Plan. If a Highly Compensated Employee participates in two or more cash or deferred arrangements that are parts of plans that have different plan years, all Deferral Contributions made during the Plan Year being tested shall be aggregated, without regard to the Plan Years of the other plans. Notwithstanding the foregoing, certain plans shall be treated as separate if mandatorily disaggregated under Regulation Section 1.401(k)-1(b)(4).
- (c) If the Plan satisfies the requirements of Section 401(k), 401(a)(4), or 410(b) of the Code only if aggregated with one or more other plans, or if one or more plans satisfy the requirements of such sections of the Code only if aggregated with the Plan, this Section 4.4 shall be applied by determining the Actual Deferral Percentages of Employees as if all such plans were a single plan. However, plans may be aggregated in order to satisfy Section 401(k) of the Code only if they have the same plan year.
- (d) For the purposes of satisfying the requirements of Section 401(k), 401(a)(4), or 410(b) of the Code, the Plan may be disaggregated into two or more plans or the Plan may be aggregated with one or more other plans, to the extent permitted by Sections 401(k), 401(a)(4), and 410(b) of the Code and the Regulations thereunder.

- (e) For purposes of determining the Actual Deferral Percentage, Elective Deferrals (excluding Catch-Up Contributions) must be made before the last day of the 12 consecutive month period immediately following the Plan Year to which those contributions relate.
- (f) To enable the Plan to satisfy the test described in Section 4.4(a), the Employer may elect to make a QNEC within the meaning of Regulation Section 1.401(k)-6 to the Plan for the Plan Year. The Employer shall determine the amount, if any, of the QNEC in its sole discretion, except that any QNEC must satisfy the requirements of this Section 4.4(f). QNECs will be allocated to the Participant's QNEC Account as of the end of the Plan Year with respect to which the QNEC is made. QNECs will be paid to the Trustee after such contribution is authorized by the Employer but no later than 12 months after the end of the Plan Year in which such contribution is allocated.

A QNEC cannot be taken into account in determining the actual deferral ratio within the meaning of Regulation Section 1.401(k)-2(a)(3) of an Eligible Employee who is a Non- highly Compensated Employee for the Plan Year to the extent that the QNEC exceeds the product of the Eligible Employee's Eligible Compensation and the greater of 5% or two times the Plan's "representative contribution rate." Any QNEC taken into account under an actual contribution percentage test under Regulation Section 1.401(m)-2(a)(6) (including the determination of the representative contribution rate for purposes of Regulation 1.401(m)-2(a)(6)(v)(B)) is not permitted to be taken into account for purposes of this Section 4.4 including the determination of the "representative contribution rate" for purposes of subsection (i) below. For purposes of this Section 4.4(f):

- (i) The Plan's "representative contribution rate" is the lowest "applicable contribution rate" of any Eligible Employee who is a Non-highly Compensated Employee among a group of eligible Non-highly Compensated Employees that consists of half of all eligible Non-highly Compensated Employees for the Plan Year (or, if greater, the lowest "applicable contribution rate" of any eligible Non-highly Compensated or of any eligible Non-highly Compensated Employee who is employed by the Employer on the last day of the Plan Year), and
- (ii) The "applicable contribution rate" for a Non-highly Compensated Employee who is an Eligible Employee is the sum of the QMACs taken into account in determining the actual deferral ratio within the meaning of Regulation Section 1.401(k)-2(a)(3) for the Eligible Employee for the Plan Year and the QNECS for the Eligible Employee for the Plan Year, divided by the Eligible Employee's Eligible Compensation for the same period.

a QMAC may only be used to calculate an actual deferral ratio to the extent that such QMAC is a matching contribution that is not precluded from being taken into account under Section 4.7 for Plan Year under the rules of Regulation Section 1.401(m)-2(a)(5)(ii) and Section 4.7.

(g) QNECs and QMACS cannot be taken into account to determine an actual deferral ratio within the meaning of Regulation Section 1.401(k)-2(a)(3) to extent such contributions are taken into account for purposes of satisfying any other actual deferral percentage test, actual contribution percentage test, or the requirements of Regulation Section 1.401(k)-3, 1.401(m)-3, or 1.401(k)-4.

4.5 Reductions During Plan Year

If, during the Plan Year, the Committee determines that the Actual Deferral Percentage test provided in Section 4.4 is not met at the time of its review or would not be met if part or all of Pre-Tax Contributions continue to be made on behalf of Participants who are Highly Compensated Employees, the Committee, in its sole discretion, may reduce the rate (to zero if necessary) of Elective Deferrals (excluding Catch-Up Contributions) that would have been made during the remainder of the Plan Year. The Committee may, in its sole and absolute discretion, limit or discontinue the Deferral Contributions of Highly Compensated Employees to comply with the contribution limits set forth herein.

4.6 Return of Excess Contributions After End of Plan Year

(a) If, after the last day of the Plan Year, the Committee determines that the Average Actual Deferral Percentage requirements of Section 4.4 have not been satisfied, the Committee, within 2½ months after the end of the Plan Year or as soon as administratively practicable thereafter (but not later than the last day of the next Plan Year), shall distribute the Excess Contributions, adjusted for any income or loss, to all affected Participants who are Highly Compensated Employees.

The Committee shall calculate any Excess Contributions after determining the amount of Excess Deferrals under Section 4.9. The amount of Excess Contributions to be distributed shall be reduced by any Excess Deferrals previously distributed to the Participant for the tax year ending with or within the Plan Year. The amount of Excess Deferrals to be distributed for a tax year shall be reduced by any Excess Contributions previously distributed for the Plan Year beginning with or within the Participant's tax year.

- (b) Distributions of Excess Contributions must be adjusted for income (gain or loss), but will not include an adjustment for income for the period between the end of the Plan Year and the date of the distribution (the "gap period"). The Committee has the discretion to determine and allocate income using any of the methods set forth below:
 - (i) <u>Reasonable method of allocating income.</u> The Committee may use any reasonable method for computing the income allocable to Excess Contributions, provided that the method does not violate Section 401(a)(4) of the Code, is used consistently for all Participants and for all corrective distributions under the Plan for the Plan Year and is used by the Plan for allocating income to Participant Accounts.

- (ii) <u>Alternative method of allocating income.</u> The Committee may allocate income to Excess Contributions for the Plan Year allocable to the Deferral Contributions and other amounts taken into account for purposes of Section 4.4 (including contributions made for the Plan Year) by a fraction, the numerator of which is the Excess Contributions for the Employee for the Plan Year, and the denominator of which is the sum of the:
 - (A) Accounts attributable to Elective Deferrals (excluding Catch-Up Contributions) and other amounts taken into account for purposes of Section 4.4 as of the beginning of the Plan Year, and
 - (B) Any additional amount of such contributions made for the Plan Year.
- (iii) <u>Safe harbor method of allocating gap period income.</u> The Committee may use the safe harbor method in this paragraph to determine income on Excess Contributions for the gap period.

Under this safe harbor method, income on Excess Contributions for the gap period is equal to 10% of the income allocable to Excess Contributions for the Plan Year that would be determined under Section 4.6(b)(ii), multiplied by the number of calendar months that have elapsed since the end of the Plan Year. For purposes of calculating the number of calendar months that have elapsed under the safe harbor method, a corrective distribution that is made on or before the 15th day of a month is treated as made on the last day of the preceding month and a distribution made after the 15th day of a month is treated as made on the last day of the month.

(iv) <u>Alternative method for allocation Plan Year and gap period income.</u> The Committee may determine the income for the aggregate of the Plan Year and the gap period, by applying the alternative method provided by Section 4.6(b)(ii) to this aggregate period. This is accomplished by (1) substituting the income for the Plan Year and the gap period, for the income for the Plan Year, and (2) substituting the amounts taken into account for purposes of Section 4.4 for the Plan Year and the gap period, for the amounts taken into account for the Plan Year in determining the fraction that is multiplied by that income.

(c) The amount of Excess Contributions for Highly Compensated Employees shall be determined as provided in this paragraph. First, the Actual Deferral Percentage of the Highly Compensated Employee with the highest such Percentage will be reduced to the extent necessary to satisfy the Actual Deferral Percentage test or cause the percentage for that Highly Compensated Employee to equal the percentage for the Highly Compensated Employee with the next highest such Percentage. Second, this process will be repeated until the Actual Deferral Percentage test is satisfied. The total of such Excess Contributions shall then be distributed to Highly Compensated Employees in descending order commencing with the Highly Compensated Employee with the highest dollar amount of Deferral Contributions and other contributions to be distributed in order to satisfy the Actual Deferral Percentage test. Excess Contributions distributed to Participants in accordance with this Section 4.6 shall be distributed in the following order:

(a) from unmatched Pre-Tax Contributions, (b) unmatched Roth Contributions, (c) matched Pre-Tax Contributions, and (d) from matched Roth Contributions.

If Excess Contributions are distributed to Participants in accordance with this Section 4.6, the Participant shall immediately forfeit all Matching Contributions that were made to match such distributed Excess Contributions.

(d) For purposes of this Section 4.6, Excess Contributions means Pre-Tax Contributions in excess of the Actual Deferral Percentage limit as described in Section 401(k)(8)(B) of the Code.

4.7 Actual Contribution Percentage Test

- (a) For the Plan Year, the Average Contribution Percentage for the group of all Highly Compensated Employees who are Eligible Employees must satisfy at least one of the following tests:
 - (i) The Average Contribution Percentage for said group of Highly Compensated Employees for that Plan Year shall not be more than the Average Contribution Percentage for the group of Non-highly Compensated Employees who are Eligible Employees for the Plan Year multiplied by 1.25; or
 - (ii) The Average Contribution Percentage for said group of Highly Compensated Employees for that Plan Year shall not exceed two percentage points more than the Average Contribution Percentage for the group of Non-highly Compensated Employees who are Eligible Employees for the Plan Year, and the Average Contribution Percentage for said group of Highly Compensated Employees for that Plan Year shall not be more than the Average Contribution Percentage for the group of Non-highly Compensated Employees who are Eligible Employees for the Plan Year multiplied by two.

- (b) If Matching Contributions are made to the Plan for a Plan Year for a Highly Compensated Employee who also is eligible to have after-tax contributions and/or matching contributions allocated to his or her account under another plan maintained by the Employer that is qualified under Section 401(a) of the Code, the Contribution Percentage for that Highly Compensated Employee shall be calculated as if all such other plans are part of the Plan. If a Highly Compensated Employee participates in two or more plans that have different plan years, all matching contributions and all after-tax contributions made during the Plan Year being tested shall be aggregated without regard to the Plan Year of the other plans. Notwithstanding the foregoing, certain plans shall be treated as separate if mandatorily disaggregated under Regulation Section 1.401(m)-1(b)(4).
- (c) If the Plan satisfies the requirements of Sections 401(m), 401(a)(4), and 410(b) of the Code only if aggregated with one or more other plans, or if one or more other plans satisfy the requirements of such sections of the Code only if aggregated with the Plan, this Section shall be applied by determining the Contribution Percentages of Employees as if all such plans were a single plan. However, plans may be aggregated in order to satisfy Section 401(m) of the Code only if they have the same plan years.
- (d) For the purposes of satisfying the requirements of Section 401(m), 401(a)(4), or 410(b) of the Code, the Plan may be disaggregated into two or more plans or the Plan may be aggregated with one or more other plans, to the extent permitted by Sections 401(m), 401(a)(4), and 410(b) of the Code and the Regulations thereunder.
- (e) For purposes of determining the Contribution Percentage, Matching Contributions will be considered made for a Plan Year if made before the last day of the 12 consecutive month period immediately following the Plan Year to which those contributions relate.
- (f) To enable the Plan to satisfy the test described in Section 4.7(a), the Employer may elect to make a QMAC within the meaning of Regulation Section 1.401(m)-5 to the Plan for the Plan Year. The Employer shall determine the amount, if any, of the QMAC in its sole discretion, except that any QMAC must satisfy the requirements of this Section 4.7(f). QMACs will be allocated to the Participant's QMAC Account as of the end of the Plan Year with respect to which the QMAC is made. QMACs will be paid to the Trustee after such contribution is authorized by the Employer, but no later than 12 months after the end of the Plan Year in which such contribution is allocated.

A QNEC cannot be taken into account under this Section 4.7 for a Plan Year for any Eligible Employee who is a Non-highly Compensated Employee to the extent such contributions exceed the product of that Employee's Compensation and the greater of 5% or two times the Plan's "representative contribution rate." Any QNEC taken into account under Section 4.4 (including the determination of the "representative contribution rate" for purposes of Regulation Section 1.401(k)-2(a)(6)(iv)(B)) is not permitted to be taken into account for purposes of this Section 4.7, including the determination of the "representative contribution rate" for purposes for subsection (i) below. For purposes of this Section 4.7:

- (i) The Plan's "representative contribution rate" is the lowest "applicable contribution rate" of any Eligible Employee who is a Non-highly Compensated Employee among a group of eligible Non-highly Compensated Employees that consists of half of all eligible Non-highly Compensated Employees for the Plan Year (or, if greater, the lowest "applicable contribution rate" of any eligible Non-highly Compensated Employee who is in the group of all Eligible Non-highly Compensated Employees for the Plan Year and who is employed by the Employer on the last day of the Plan Year), and
- (ii) The "applicable contribution rate" for any Eligible Employee who is a Non-highly Compensated Employee is the sum of the matching contributions (as defined in Regulation Section 1.401(m)-1(a)(2)) taken into account in determining the Contribution Percentage for the eligible Non-highly Compensated Employee for the Plan Year and the QNECs made for that Employee for the Plan Year, divided by the Employee's Compensation for the Plan Year.
- (g) A Matching Contribution cannot be taken into account under this Section 4.7 for any Eligible Employee who is a Non-highly Compensated Employee to the extent it exceeds the greatest of:
 - (i) five percent of the Eligible Employee's Compensation for the Plan Year;
 - (ii) the sum of the Eligible Employee's Deferral Contributions for the Plan Year; and
 - (iii) the product of two times the Plan's "representative matching rate" and the Eligible Employee's total Deferral Contributions for the plan Year.

The Plan's "representative matching rate" is the lowest "matching rate" for any Eligible Employee who is an Non-highly Compensated Employee among a group of eligible Non-highly Compensated Employees that consists of half of all the eligible Non-highly Compensated Employees for the Plan Year who make Deferral Contributions (or, if greater, the lowest "matching rate" for all eligible Non-highly Compensated Employees who are employed by the Employer on the last day of the Plan Year and make Elective Deferrals for the Plan Year).

The "matching rate" for an Employee generally is the amount of Matching Contributions made for such Employee divided by the sum of the Participant's Elective Deferrals for the Plan Year. If the matching rate is not the same for all levels of Deferral Contributions made by a Participant, the Participant's "matching rate" is determined assuming that an Employee's Deferral Contributions equals 6% of Compensation.

4.8 Return of Excess Aggregate Contributions

- (a) If, after the last day of the Plan Year, the Committee determines that the Average Contribution Percentage requirements of Section 4.7 have not been satisfied, the Committee, within 2½ months after the end of the Plan Year or as soon as administratively practicable thereafter (but not later than the last day of the next Plan Year), shall first cause to be forfeited, if forfeitable, or if not forfeitable, distribute the Excess Aggregate Contributions, adjusted for any income or loss, to all affected Participants who are Highly Compensated Employees. The Committee shall calculate any Excess Aggregate Contributions after determining the amount of Excess Deferrals under Section 4.9 and the amount of Excess Contributions under Section 4.6.
- (b) Distributions of Excess Aggregate Contributions must be adjusted for income (gain or loss), but will not include an adjustment for income for the period between the end of the Plan Year and the date of the distribution (the "gap period").

For purposes of this Section 4.8(b), "income" shall be determined and allocated in accordance with the methods described in Section 4.6(b), except that such section shall be applied by substituting "Excess Aggregate Contributions" for "Excess Contributions" in each place that it appears and by substituting amounts taken into account under Section

4.7 for amounts taken into account under Section 4.4.

- (c) The amount of Excess Aggregate Contributions for Highly Compensated Employees shall be determined as provided in this Section 4.8. First, the Contribution Percentage of the Highly Compensated Employee with the highest such Percentage will be reduced to the extent necessary to satisfy the Contribution Percentage test or cause the percentage for that Highly Compensated Employee to equal the percentage for the Highly Compensated Employee with the next highest such Percentage. Second, this process will be repeated until the Contribution Percentage test is satisfied. The total of such Excess Aggregate Contributions to be distributed shall then be distributed to Highly Compensated Employees in descending order commencing with the Highly Compensated Employee with the highest dollar amount of Excess Aggregate Contributions and other contributions to be distributed in order to satisfy the Contribution Percentage test.
- (d) Excess Aggregate Contributions forfeited in accordance with this Section 4.8 shall be treated as Annual Additions under Section 4.10 and shall be applied to reduce subsequent Matching Contributions.
- (e) Excess Aggregate Contributions distributed to Participants in accordance with this Section
 4.8 shall be distributed first from the Participant's After-Tax Contributions Account and then from his or her Matching Contributions Account.

4.9 Distribution of Excess Deferrals

- (a) A Participant may state a claim for the return of Excess Deferrals and such Excess Deferrals, adjusted for any income or loss, shall be distributed if administratively practicable no later than the April 15 following the calendar year for which such allocable Excess Deferrals are made. The Participant's claim shall be made in accordance with applicable administrative procedures, shall be submitted to the Committee no later than March 1 (or as late as April 14 if allowed by the Committee), shall specify the Participant's Excess Deferrals for the preceding calendar year, and shall be accompanied by the Participant's statement that such amounts, if not distributed, will constitute Excess Deferrals.
- (b) The income or loss allocable to Excess Deferrals for the Plan Year shall be determined by multiplying the income or loss allocable to the Participant's Pre-Tax Contributions for the Plan Year by a fraction, the numerator of which is the Excess Deferrals on behalf of the Participant for the Plan Year and the denominator of which is the Participant's Account attributable to Pre-Tax Contributions on the last day of the Plan Year, without regard to any income or loss during the Plan Year. No income or loss shall be attributable to the period between the end of the Plan Year and the date of the distribution.
- (c) If Excess Deferrals have previously been distributed within the Plan Year, the Plan shall offset such distribution from the amount of the Participant's Excess Contributions, if any, to be distributed for such Plan Year. In addition, the amount of Excess Deferrals that may be distributed for a Participant by the Plan for a Plan Year shall be reduced by the amount of Excess Contributions previously distributed.
- (d) Excess Deferrals are Pre-Tax Contributions in excess of the limit imposed by Section 402(g) of the Code.
- (e) Excess Deferrals shall be taken first from unmatched Pre-Tax Contributions, then form unmatched Roth Contributions, then from matched Pre-Tax Contributions and finally from matched Roth Contributions. Any Matching Contributions attributable to refunded Excess Deferrals, adjusted for investment gain or loss, shall be forfeited and used in the manner described in Section 7.3.

4.10 Maximum Annual Additions

The Annual Addition to a Participant's Accounts for any Limitation Year, when added to the annual additions for such year under any other defined contribution plans (including voluntary employee contribution accounts in a defined benefit plan and key employee accounts under a welfare benefit plan described in Section 419 of the Code as well as employer contributions allocated to an IRA) maintained by the Employer, shall not exceed the lesser of (a) 100% of the Participant's 415 Compensation as defined in Section 415(c)(3) of the Code for that Limitation Year, or (b) \$69,000 multiplied by the Adjustment Factor.

The compensation limitation referred to in (a) shall not apply to any contribution for medical benefits after separation from service (within the meaning of Section 401(h) or 419A(f)(2) of the Code) which is otherwise treated as an Annual Addition.

In the event that an excess over the amount of Annual Additions permitted for a Limitation Year under Section 415(c) of the Code is caused by the allocation of forfeitures, a reasonable error in estimating a Participant's 415 Compensation or in determining the amount of Deferral Contributions that may be made with respect to the Participant for the Plan Year or other circumstances approved by the Commissioner of Internal Revenue, the Committee shall take whatever action is necessary to eliminate such excess, all in accordance with Section 415 of the Code and the Regulations thereunder.

If the limitations of this Section 4.10 are exceeded because a Participant is also participating in another plan required to be aggregated with this Plan for purposes of Section 415 of the Code, then the extent to which annual contribution under this Plan will be reduced, as compared with the extent to which annual benefits or contributions under any other plans will be reduced, will be determined by the Committee in a manner as to maximize the aggregate benefits payable to the Participant from all plans. If the reduction is under this Plan, the Committee will advise affected Participants of any additional limitation on their annual contribution required by this paragraph.

For purposes of this Section 4.10, the definition of Employer shall be modified as provided in Section 415(h) of the Code.

The Committee may operationally implement an ordering rule procedure for distribution of Deferral Contributions in order to meet the provisions of this Section 4.10.

4.11 Return of Contributions to Employer

If a contribution or portion of a contribution made by the Employer is determined to be nondeductible under Section 404 of the Code or is made due to a mistake of fact, the Employer may request that the Trustee return the contribution, without interest but reduced by any investment loss allocable to the contribution. The return shall be made as soon as practicable within one year after the date the contribution was made.

4.12 Rights of Reemployed Veterans

A Reemployed Veteran shall be entitled to the restoration of certain benefits under the Plan that would have accrued, or that he or she would have received, under the Plan but for his or her absence from the employ of the Employer due to Qualified Military Service. A Reemployed Veteran is defined as an Employee who left the employ of the Employer to perform service in the Armed Services of the United States, and subsequently was reemployed by the Employer pursuant to the Uniformed Services Employment and Reemployment Rights Act of 1994 (USERRA). 'Qualified Military Service' is defined as service in the uniformed services (as defined in chapter 43 of title 38, United States Code) performed by the Reemployed Veteran whose entitlement to reemployment rights pursuant to USERRA arose with respect to such service.

- (a) <u>Crediting of Period of Qualified Military Service.</u> To the extent required by USERRA and Section 414(u) of the Code, the Reemployed Veteran, for all purposes under the Plan, shall be credited Hours of Service for his or her absence from the employ of the Employer due to Qualified Military Service, in accordance with the regulations or other rules provided by the Internal Revenue Service.
- (b) <u>"Make-Up" Contributions.</u> To the extent required by USERRA and Section 414(u) of the Code, the Reemployed Veteran shall be permitted to make additional contributions and receive related Matching Contributions during the period which (a) begins on the Reemployed Veteran's date of reemployment with the Employer, and (b) has the same length as the lesser of: (i) the period of Qualified Military Service multiplied by three, or (ii) five years.

The maximum amount of additional contributions that the Reemployed Veteran is permitted to make is the maximum amount of such contributions that the Reemployed Veteran would have been permitted to make had he or she continued to be employed by the Employer during the period of Qualified Military Service and received compensation. Compensation for purposes of this Section 4.9 shall be based on the rate of pay that the Reemployed Veteran would have received during the period of Qualified Military Service had he or she remained employed by the Employer. If such rate of pay was not reasonably certain, such compensation shall be based on the Reemployed Veteran's average Eligible Compensation from the Employer during (a) the 12 month period immediately before the Qualified Military Service, or (b) if shorter, the period of employment immediately before the Qualified Military Service.

If any contribution is made by an Employer or an Employee under the Plan with respect to an Employee whose reemployment is governed by USERRA:

- (i) Such contribution shall not be subject to any otherwise applicable limitation contained in Section 402(g), 402(h), 403(b), 404(a), 404(h), 408, 415, or 457 of the Code and shall not be taken into account in applying such limitations to other contributions or benefits under such plan or any other plan, with respect to the year in which the contribution is made;
- (ii) Such contribution shall be subject to the limitations referred to in (i) above with respect to the year to which the contribution relates (in accordance with rules prescribed by the Secretary of the Treasury); and
- (iii) The Plan shall not be treated as failing to meet the requirements of Section 401(a)(4), 401(a)(26), 401(k)(3), 401(k)(11), 401(k)(12), 401(m), 403(b)(12), 408(k)(3), 408(k)(6), 408(p), 410(b), or 416 of the Code by reason of the making of (or the right to make) such contribution.

Article V - Maintenance and Valuation of Accounts

5.1 Establishment of Accounts

The Committee shall maintain an Account for each Participant or Beneficiary for the purpose of accounting for the beneficial interest of the Participant or Beneficiary in the Plan. Each Account shall be adjusted to reflect administrative expenses and any distributions. Each Account consists of any additional subaccounts that the Committee deems appropriate. Separate subaccounts shall be maintained for the accounts merged into the Plan from the ACT Plan or a Merged Plan as determined by the Committee.

5.2 Valuation of Accounts

As of each Valuation Date, the Accounts of each Participant shall be adjusted to reflect contributions, withdrawals, distributions, or income earned or accrued paid from the assets of the Plan, any administrative fee assessed, and any increase or decrease in the fair market value of the assets of the Plan since the preceding Valuation Date.

Article VI - ESOP Provisions and Investment of Contributions

6.1 Purpose and Nature of the ESOP

The Company has established the ESOP as part of the Plan, the primary purpose of which is to enable Participants to share in the growth and prosperity of the Company and its Affiliates by enabling Participants to acquire Company Stock. The ESOP is intended to be a stock bonus plan with the meaning of Regulation 1.401-1(b)(1)(iii) that is qualified under Section 401(a) and is also designed to meet the requirements of Section 4975(e)(7) of the Code. The Company may, in its sole discretion, contribute Company Stock to the Plan in lieu of cash as its Matching Contribution and/or Equity Builder Contribution under Section 4.2.

Except as permitted by law, Company Stock acquired with the proceeds of an exempt loan (as such loan may have been provided for by the Plan prior to January 1, 2024) may not at that time or at any time thereafter by subject to any put, call, option, buy-sell or other similar arrangement when held and distributed from the Plan, whether or not the Plan is then an employee stock ownership plan. The rights and protections granted by this paragraph are non- terminable and will continue to exist as long as any Company Stock acquired with the proceeds of an exempt loan (as such loan may have been provided for by the Plan prior to January 1, 2024), is held by the Plan or by any Participant or any other person for whose benefit such protections and rights have been created, and neither the repayment of such loan nor the failure of the Plan to be an employee stock ownership plan, nor any amendment of the Plan, will cause a termination of such protections and rights.

6.2 Allocations and Accounting

Each Participant's Matching Contribution Account and/or Equity Builder Contribution Account shall (a) reflect such Participant's interest in the ESOP; if any, (b) be credited with its allocable share of Company Stock contributed to the Plan by the Company; and (c) be credited with its allocable share of cash dividends on Company Stock allocated to the Participant's Employer Contribution Account and proceeds from the sale of any shares of Company Stock.

6.3 Company Stock Cash Dividends

(a) Cash Dividends Paid.

(i) Notwithstanding any other provision of this Article VI to the contrary, each Participant may elect to (A) receive a distribution in cash equal to the value of an cash dividends paid by the Company and received by the Trust with respect to shares of Company Stock allocated to his or her Matching Contribution Account and/or Equity Builder Account at the close of business on the ex-dividend date established for the payment of such cash dividends; or (B) reinvest in the Company Stock Fund any cash dividends paid by the Company and received by the Trust with respect to shares of Company Stock allocated to his or her Matching Contribution Account and/or Equity Builder Account at the close of business on the ex-dividend date established for the payment of such cash dividends.

- (ii) Any distribution pursuant to (a)(i)(A) above shall be made as soon as is administratively feasible following the receipt of the cash dividends by the Trust, but in no event later than 90 days after the close of the Plan Year in which such cash dividends were paid by the Company.
- (iii) A Participant shall be deemed to have elected to receive payment of a dividend pursuant to (a)(i)(B) unless he or she affirmatively elects to receive it in cash pursuant to (a)(i)(A).
- (b) <u>Elections.</u> The Company shall specify the manner in which Participants will be required to make their elections subject to the following conditions:
 - (i) The Committee shall provide no less than annually each participant an opportunity to make an election.
 - (ii) A Participant's election shall take effect immediately following receipt by the Committee and shall remain in effect until an election to the contrary is filed by the Participant.
 - (iii) A Participant's election shall become irrevocable at the later of (A) the date on which the cash dividends attributable to such election are paid; or (B) the date established by the Committee for revoking such election.
 - (iv) The rules established by the Committee for making an election shall be applied in a uniform and nondiscriminatory manner.

6.4 Diversification Election

A Participant, alternate payee or Beneficiary may elect to direct the Plan to divest any shares of Company Stock that have been allocated to his or her Account (including, but not limited to, the Matching Contribution Account and/or Equity Builder Account) and to reinvest an equivalent amount in an Investment Fund other than the Company Stock Fund. Such election may be made at any time during the Plan Year in accordance with procedures established by the Committee. The Committee may impose restrictions on the divesture of Company Stock that the Committee is either required to do so in order to ensure compliance with applicable securities laws or is reasonably designed to ensure compliance with applicable securities laws. The Administrator shall direct the Trustee in writing to liquidate, including any specific directions as to the manner in which to liquidate, the shares to Company Stock for which a divesture election has been made and to transfer the cash proceeds to the Investment Fund elected by the Participant as soon as administratively feasible. A Participant may not subsequently elect to again invest the cash proceeds attributable to a divesture election in Company Stock.

6.5 Voting and Tendering of Company Stock

The Trustee shall vote each share of Company Stock held by the Plan. Each Participant shall be entitled to direct the Trustee as to the manner in which the voting rights attributable to the shares of Company Stock allocated to his or her Matching Contribution Account and/or Equity Builder Account to be exercised in accordance with procedures established by the Committee.

The Trustee shall vote all shares of Company Stock as which it receives timely voting instruction solely in accordance with those instructions, provided, however, that the Trustee may vote the shares as it determines is reasonably necessary to fulfill its fiduciary duties under ERISA. If a Participant does not, with respect to any manner, give instructions concerning the voting of Company Stock allocated to his or her Account, the Trustee shall vote the Company Stock attributable to that Participant's Account in the same proportion as Company Stock for which instructions have been received, provided, however, that the Trustee may vote the shares as it determined is reasonably necessary to fulfill its fiduciary duties under ERISA.

6.6 Investment Funds

Contributions under the Plan made by a Participant to the Plan and the proceeds from a Participant's election pursuant to Section 6.4 above, may at the election of the Participant, be invested in one or more Investment Funds as selected by the Committee and made available to a Participant under the Plan. The Committee shall monitor the Investment Funds and may modify or eliminate such Investment Funds as it determines necessary or appropriate in the capacity of a fiduciary at any time. It is intended that the Committee make Investment Funds available in accordance with the provisions of Section 404(c) of ERISA.

- (a) Pending the investment of any amounts in an Investment Fund, the Trustee may invest assets of the Trust Fund temporarily in interest-bearing accounts, certificates of deposit, Treasury bills, commercial paper, money market funds, short-term obligations of the United States Government, short-term investment funds, or other short-term obligations selected by the Trustee. The Trustee may keep such amounts of cash, as it, in its sole discretion, shall deem necessary or advisable as part of such funds, all within the limitations specified in the Trust Agreement or other agreements pertaining to recordkeeping and/or trust agreements entered into for the Plan.
- (b) Other than as provided in Section 6.3, all interest, dividends and proceeds from the disposition of and other income received with respect to assets held with respect to each of Investment Fund shall be reinvested in the respective Investment Fund and all expenses of the Trust that are properly allocable to a particular Investment Fund shall be so allocated and charged.
- (c) The Committee may elect to terminate an Investment Fund with respect to receiving future contributions, but may elect to keep existing amounts currently invested in such Investment Fund. Such amounts shall continue to be invested with such Investment Fund or, alternatively, the Committee may elect to transfer all amounts to be invested in a new Investment Fund. Upon election by the Committee to transfer all investments to a new Investment Fund, the existing Investment Fund will value all accounts on the selected date of transfer and make such transfers as directed by the Committee.

6.7 Investment of Participant Accounts

A Participant may, electronically or telephonically, in accordance with applicable administrative procedures established by the Committee, specify the percentages of his or her Accounts that shall be invested in each Investment Fund maintained under the Plan. If a Participant fails to make an election under this Section 6.7, the portion of his or her Account not attributable to Employer Contributions shall be invested in an Investment Fund that the Committee determines, in its sole discretion, is consistent with the prudent discharge of its fiduciary duties.

6.8 Changing Investment Elections – Future Contributions

A Participant may, electronically or telephonically, in accordance with applicable administrative procedures established by the Committee, change his or her investment election as to subsequent contributions, subject to the limitations of Section 6.7, as of any Valuation Date.

6.9 Transfer Among Investment Funds

A Participant may, electronically or telephonically, in accordance with applicable administrative procedures established by the Committee and subject to any restrictions that may be imposed by particular Investment Funds, elect to transfer all or a portion of the balance in all of his or her Accounts between and among Investment Funds as of any Valuation Date. The Committee, however, reserves the right, in its sole discretion, to implement reasonable restrictions on a Participant's right to transfer among Investment Funds.

Article VII - Vesting

7.1 Full Vesting in Designated Accounts

A Participant shall at all times have a 100% nonforfeitable vested right to the value of all his or her Accounts, except the Matching Contributions Account and the Equity Builder Contributions Account. A Participant shall also be 100% vested in all cash dividends received by the trust for which an election is offered pursuant to Section 6.3.

7.2 Vesting in Matching Contributions and Equity Builder Contributions

(a) A Participant shall have a nonforfeitable vested right to the value of his or her Matching Contributions Account and his or her Equity Builder Account determined in accordance with the following schedule:

Years of Vesting Service	Vested Percentage
Less than 2 years	0%
2 but less than 3 years	20%
3 but less than 4 years	40%
4 but less than 5 years	60%
5 but less than 6 years	80%
6 or more years	100%

- (b) Notwithstanding the provisions of (a) above, a Participant shall have a 100% nonforfeitable vested right to the value of his or her Matching Contributions Account and his or her Equity Builder Account upon the occurrence of any of the following events prior to his or her Severance from Employment:
 - (i) attainment of his or her Early Retirement Date or Normal Retirement Age, if he or she is then an Employee,
 - (ii) Disability,
 - (iii) death, if he or she was an Employee immediately before his or her death, or
 - (iv) in accordance with Section 4.9 if the Participant dies while performing qualified military service within the meaning of Section 414(u) of the Code, the survivors of the Participant shall be entitled to any additional benefits (other than benefit accruals related to the period of qualified military service) provided under the Plan had the Participant resumed and then terminated employment because of death.

7.3 Forfeiture of Nonvested Interest

Upon a Participant's Severance from Employment, the nonvested portion of his or her Equity Builder Account and Matching Contributions Account shall be forfeited at the earlier of the date he or she receives a distribution of the vested portion of his or her Account or the date he or she incurs five consecutive Breaks in Service. The Committee shall apply all forfeitures by first restoring the amount of previous forfeitures in accordance with Section 7.4, by paying Plan expenses and then by either reducing the amount of Employer contributions required under the Plan in accordance with Section 4.2 for the then current Plan Year. Notwithstanding the foregoing, if the vested portion of a Participant's Account is \$0.00 at the time of his or her Severance from Employment, the Participant shall be deemed to have received a distribution of his or her vested Account at his or her Severance from Employment.

7.4 Restoration of Forfeitures and Service

- (a) If a former Participant whose Severance from Employment resulted in a forfeiture of the entire nonvested portion of his or her Matching Contributions Account and/or Equity Builder Account under Section 7.3 resumes employment with the Employer after at least five consecutive Breaks in Service, he or she shall have no right to restoration of any previously forfeited portion of his or her Account. Such Participant's Years of Vesting Service or period of employment after the Break in Service shall not be taken into account in determining the Participant's vested nonforfeitable right to the value of his or her Matching Contributions Account and Equity Builder Account attributable to contributions made by the Employer before he or she resumed participation in the Plan.
- (b) If a former Participant who did not receive a distribution resumes employment with the Employer prior to incurring five consecutive Breaks in Service, his or her Years of Vesting Service before and after the break in service shall be taken into account in determining his or her nonforfeitable right to the value of his or her Account attributable to Employer contributions described in Section 4.2.
- (c) If a former Participant whose Severance from Employment resulted in a forfeiture of the entire nonvested portion of his or her Matching Contributions Account and/or his or her Equity Builder Account under Section 7.3 resumes employment with the Employer prior to incurring five consecutive Breaks in Service, the previously forfeited portion of his or her Account shall be restored as soon as administratively practicable after the Participant's Reemployment Commencement Date. Such Participant's Years of Vesting Service before and after the Break in Service shall be taken into account in determining the Participant's vested nonforfeitable right to the value of his or her Account attributable to Employer contributions described in Section 4.2.

- (d) If a distribution or withdrawal is made at a time when a Participant has a less than 100% nonforfeitable right to the value of his or her Matching Contributions Account and/or Equity Builder Account and may increase his or her vested percentage in such portion of his or her Account after the distribution or withdrawal:
 - (i) A separate account will be established for the Participant's interest in the Plan as of the time of the distribution or withdrawal, and
 - (ii) At any relevant time, the Participant's nonforfeitable portion of the separate account will be equal to an amount ("X") determined by the formula: X = P (AB + (RxD)) (RxD).

For purposes of applying the formula, P is the nonforfeitable percentage at the relevant time, AB is the account balance at the relevant time, D is the amount of the distribution or withdrawal, and R is the ratio of the account balance at the relevant time to the account balance after the distribution or withdrawal.

Article VIII - Withdrawals and Loans During Employment

8.1 General Rules Applicable to All In-Service Withdrawals

The only in-service withdrawals permitted by the Plan are those described in this Article VIII.

Each in-service withdrawal request must be filed in accordance with applicable administrative procedures established by the Committee. Each withdrawal shall be determined as of the Valuation Date as soon as practicable after the withdrawal request is approved and shall be drawn, to the extent available, pro rata from the Investment Funds in which the Account is invested or in which the Accounts are invested if the withdrawal is taken from multiple Accounts. Only a Participant who has not incurred a Severance from Employment may request an in-service withdrawal.

8.2 Hardship Withdrawals

- (a) A Participant, in the event of hardship, may, at any time, be permitted to make a withdrawal from his or her Pre-Tax Contributions Account or his or her Roth Contributions Account. In no event may a Participant be permitted to receive a distribution for a hardship from an amount attributable to earnings on his or her Pre-Tax Contributions Account or Roth Contributions Account. A hardship for purposes of this Section 8.2 is defined as an immediate and heavy financial need. The following are the only financial needs considered, for purposes of the Plan, to be immediate and heavy:
 - Unreimbursed expenses incurred or necessary to obtain medical care that would be deductible under Section 213(d) of the Code (determined without regard to whether the expenses exceed 7.5% of adjusted gross income), incurred by the Participant, the Participant's spouse, children, dependent or Beneficiary;
 - (ii) Costs (excluding mortgage payments) directly related to the purchase or construction of the Participant's principal residence;
 - (iii) Payment of tuition and related educational fees for the next 12 months of postsecondary education for the Participant, his or her Spouse, children, dependents (as defined in Code Section 152 but without regard to Code Section 152(b)(1), (b)(2) and (d)(1)(B) of the Code) or Beneficiary;
 - (iv) Payments necessary to prevent the eviction of the Participant from his or her principal residence or foreclosure on the mortgage of the Participant's principal residence;
 - (v) Burial or funeral expenses for the Participant's deceased parent, spouse, children, or dependents (as defined in Section 152 of the Code but without regard to Section 152(d)(1)(B) of the Code) or Beneficiary;
 - (vi) Expenses for the repair of damage to the Participant's principal residence that would qualify for the casualty deduction under Section 165 of the Code (determined without regard to whether the loss exceeds 10% of adjusted gross income);

- (vii) Expenses and losses (including loss of income) incurred by the Participant on account of a disaster declared by the Federal Emergency Management Agency (FEMA) under the Robert T. Stafford Disaster Relief and Emergency Assistance Act, Public Law 100-707, provided that the Participant's principal residence or principal place of employment at the time of the disaster was located in an area designated by FEMA for individual assistance with respect to the disaster; and
- (viii) Other events which are adopted by the Committee, and which are deemed immediate and heavy financial needs by the Commissioner of the Internal Revenue Service through the publication of revenue rulings, notices, and other documents of general.
- (b) A distribution will be considered necessary to satisfy an immediate and heavy financial need of the Participant only if:
 - (i) The Employee has obtained all distributions, other than hardship distributions under all Plans maintained by the Employer;
 - (ii) The distribution is not in excess of the amount of an immediate and heavy financial need (including amounts necessary to pay any federal, state, or local income taxes or penalties reasonably anticipated to result from the distribution); and
 - (iii) The Participant represents (in writing, electronically, or in another form permitted by the Commissioner of the Internal Revenue Service) that he or she has insufficient cash or other liquid assets to satisfy the immediate and financial need for which the hardship is being sought.
- (c) The distribution will be made in one lump sum payment.
- (d) The Committee may operationally establish an ordering rule procedure for hardship withdrawals which may specify the order in which the Accounts referenced above in Section 8.2(a) shall be distributed as a hardship withdrawal.
- (e) A request for a withdrawal under this Section 8.2 shall be made on forms prescribed by the Committee. The Committee shall establish a uniform and nondiscriminatory policy for reviewing withdrawal applications and any determination made by the Committee shall be final but subject to appeal under Section 11.5.
- (f) For purposes of this Section 8.2, a primary beneficiary means a Beneficiary under the Plan who has an unconditional right to all or a portion of the Participant's Account upon the death of the Participant.

8.3 Withdrawals of Rollover Contributions from the ACT Plan

A Participant may at any time withdraw amounts that are attributable to his or her Rollover Contributions Account from the ACT Plan.

8.4 Withdrawals Upon Attainment of Age 59¹/₂

Upon a Participant's attainment of age 59½, the Participant may, at any time, withdrawal all or any vested amounts credited to his or her Accounts. A Participant who has attained age 59½ may not receive more than two withdrawals under this Section 8.4 in any Plan Year.

8.5 Qualified Reservist Withdrawals

A Participant may, at any time, withdraw any or all amounts from his or her Pre-tax Contributions Account or Roth Contributions Accounts as qualified reservist distribution. A qualified reservist distribution for purposes of this Section 8.5 is any distribution to a Participant if (a) the Participant is a member of a military reserve unit ordered or called to active duty after September 11, 2001 for a period in excess of 179 days or for an indefinite period; and (b) the distribution is made during the period beginning on the date of the order or call to duty and ending at the close of the active duty period.

8.6 Loans to Participants

- (a) Loans. The Committee or its delegate may, in accordance with a uniform and nondiscriminatory policy, direct the Trustee to Ioan a Participant amounts from the portion of his or her Accounts held in the Trust that are attributable to Elective Deferral Contributions, Rollover Contributions and the vested portion of any Employer Contributions. All Ioans shall be in accordance with the terms, conditions, requirements and limitations specified in this Section 8.6 and any separate written document adopted by the Committee and forming part of the Plan. A Ioan pursuant to this Section 8.5 shall only be made for one of the reasons set forth in Section 8.2(a) or for the purchase of the Participant's principal residence as defined in Section 8.5(e). It is intended that all Ioans made to Participants under this Section 8.6 shall meet the requirements of Section 72(p) of the Code.
- (b) <u>Loan Administration</u>. The loan provision of the Plan shall be administered on a uniform and nondiscriminatory basis in accordance with terms and conditions established by the Committee.

The Committee has the authority to review loan requests, execute loan agreements and collect loan payments.

In administering the loan provisions of this Section 8.6, the Committee shall:

- adopt such rules and regulations as it deems necessary for the proper and efficient administration of loans, including, but not limited to, appropriate adjustments in the accounting provisions of the Plan as it deems necessary and advisable to facilitate and account for loans;
- (ii) establish standards that shall be used to determine if a loan request should be approved;

- (iii) determine how the interest rate to be charged on outstanding loans is to be calculated and when the rate to be charged for new loans is to be changed;
- (iv) determine, from time to time, the minimum loan amount;
- (v) employ agents, attorneys, accountants, and other persons to administer the loan provision and to collect outstanding loans; and
- (vi) take all other actions necessary or advisable to carry out the provisions of this Section 8.6.
- (c) Loan Eligibility. Any Participant who is an Eligible Employee may request a loan subject to the terms, conditions and limitations prescribed in this Section 8.6. Neither a Participant who has incurred a Severance from Employment and not been reemployed nor a Beneficiary shall be eligible for a loan under this Section 8.6.
- (d) <u>Loan Request.</u> Each loan request must be made in accordance with procedures prescribed by the Committee.
- (e) <u>Term of Loan and Payment</u>. Each loan shall be evidenced by a legally enforceable agreement in a form approved by the Committee and shall provide for payment of principal and interest based on substantially level amortization payments. All loans shall be subject to a specific repayment schedule with payments to be made not less frequently than quarterly over the term of the loan. The period of repayment for any loan shall in no event exceed 60 months, unless the loan is a principal residence loan, in which case the period of repayment of such loan shall in no event exceed 180 months. A principal residence loan, for this purpose, is a loan used to acquire a dwelling unit that within a reasonable time following acquisition is to be used (determined at the time the loan is made) as the principal residence of the Participant. The Committee shall have the discretion to determine when and under what circumstances a principal loan shall be made and the loan repayment period for such loan.
- (f) <u>Number of Loans.</u> A Participant shall not have more than two (2) loans outstanding at any one time from the Plan.
- (g) <u>Security</u>. A loan to a Participant shall be secured by the Participant's Account. Loan payments shall be required to be made through payroll deductions, and each Participant shall be required to execute an irrevocable authorization directing the Employer to deduct the loan payments from the Participant's wages or salary, which amounts shall be transmitted to the Trustee and applied against the outstanding loan balance. A Participant who terminates with an outstanding loan balance shall be permitted to continue making loan repayments other than through payroll deduction in accordance with the repayment schedule set forth in the promissory note and in accordance with methods prescribed by the Committee. Participants may prepay the entire amount of the remaining unpaid principal balance (and all remaining interest due thereon) at any time without penalty.

- (h) <u>Maximum Loan</u>. Loans to a Participant (when added to the outstanding balance of all other loans from loans from the Plan and any other qualified plan maintained by the Aggregated Employer) shall not be in an amount that exceeds the lesser of:
 - (i) \$50,000, reduced by the excess (if any) of the highest outstanding balance of loans from the Plan during the one-year period ending on the day before the date on which such loan is made, over the outstanding balance of other loans from the Plan on the date the new loan is made, or
 - (ii) fifty percent of the vested portion of the Participant's Account reduced by the then outstanding balance of any other loans that the Participant received from the Plan.
- (i) <u>Interest.</u> Each loan shall bear interest at a rate to be fixed by the Committee and, in determining the interest rate, the Committee shall take into consideration interest rates currently being charged by commercial lending institutions. Interest rates shall be fixed for the terms of the loan at the time the loan is made, and the Committee shall determine periodically the interest rate to be charged on new loans.
- (j) <u>Failure to Repay Loans.</u> The Committee shall establish uniform rules to apply where a Participant fails to repay any portion of a loan made to him or her and accrued interest thereon in accordance with the terms of the loan, or where any portion of a loan and accrued interest thereon remains unpaid on a Participant's Severance from Employment. Such rules shall be consistent with Section 72(p) of the Code and Regulations thereunder. Loan repayments with respect to qualified military service will be suspended as permitted under Section 414(u)(4) of the Code.

Notwithstanding the foregoing provisions of this Section 8.6(j), if a Participant Ioan remains unpaid at the time that a distribution is due the Participant (or his or her Beneficiary) under the Plan, the Committee shall reduce the amount otherwise distributable to the Participant or Beneficiary by the unpaid balance of principal and accrued interest on the Participant's Ioan and distribute (in kind) the promissory note or other agreement evidencing such Ioan in full or partial satisfaction of the obligation to distribute the Participant's vested Account.

The requirement that loan repayments be made on an amortized basis, not less frequently than quarterly, may be suspended for up to one year while a Participant is on a leave of absence approved by the Employer, either without pay or at a rate of pay (after income and employment tax withholding) that is less than the amount of the installment payments required under the terms of the Participant loan. A suspension of regularly scheduled loan payments shall not in any event extend the due date of the Participant loan beyond the due date in effect immediately prior to the leave of absence (or an accelerated due date in the case of the Participant's termination of employment) and the installments due after the leave of absence ends (or, if earlier, after the first year of the leave of absence) must not be less than the installments required under the terms of the original loan.

A default in repayment of a Participant loan shall occur when a Participant fails to make any payment when due under the terms of a Participant loan. Notwithstanding the above, the Committee may establish a grace period, during which Participants may make past due payments without incurring a default. If a grace period is established, it shall in no event continue beyond the last day of the calendar quarter following the calendar quarter in which the required payment was initially due from the Participant. Such grace period, if any, shall be applied in a uniform and nondiscriminatory manner, taking into account commercially reasonable factors concerning whether (or to what extent) a grace period should be allowed based on all of the facts and circumstances of the particular case.

Upon the occurrence of a default, as described above, there shall be a deemed distribution to the Participant equal to the entire outstanding balance of the Participant loan at the time of such default.

(k) <u>Directed Investment.</u> Any loan to a Participant under this Section 8.6 shall be made pro rata from the Investment Funds in which the Participant's individual Account is invested, shall be charged against said Account and shall be treated as a segregated investment of the Participant's Account. Any principal and interest paid on the loan shall be paid in accordance with the Participant's investment elections in effect at the time of the loan repayment. Loan repayments shall be applied first to satisfy accrued loan interest and the remainder shall be applied to principal. If and to the extent required by the Sarbanes- Oxley Act or other applicable law, loans shall not be made to officers of the Employer or other prohibited classifications of Participants.

Article IX - Distributions Upon Severance from Employment

9.1 Eligibility for Distribution

- (a) <u>General Rule.</u> A Participant's vested Account, including a Participant's Prior Employer Account, shall become payable upon Severance from Employment for any reason, including death.
- (b) <u>HEART Act Deemed Severance from Employment.</u> During the period (after more than 30 days) that a Participant is on active duty in the uniformed services (as defined in Chapter 43 of Title 38, United States Code), he or she shall be treated as have incurred a Severance from Employment with the Employer for purposes of receiving a distribution from the Plan under Sections 403(b)(7)(A)(ii) and 403(b)(11)(A) of the Code. If such Participant elects to receive a distribution from the plan under Section 403(b) of the Code, he or she may not be permitted to have Employee contributions as described in Section

4.1 made on his or her behalf or to make any contributions to any other plan maintained by the Aggregated Employer during the six month period following the date of the distribution.

9.2 Forms of Payment

Benefits shall be payable in cash in a single lump sum payment provided, however, that the portion of a Participant's Account invested in the Company Stock Fund may, at the election of the Participant, be made in the form of (a) cash or (b) full shares of Company Stock and cash in lieu of fractional shares.

9.3 Timing of Payment

Benefits that become payable in accordance with Section 9.1 shall be distributed as soon as administratively feasible after the Participant, or the Beneficiary, as the case may be, elects in accordance with procedures established by the Committee to receive a distribution. If the vested value of the Participant's Account is more than \$7,000 at the time benefits become distributable in accordance with Section 9.1, the Participant must consent to the distribution. Such consent must be obtained within the 180-day period ending on the date the Participant elects to receive a distribution.

If the vested value of the Participant's Account is \$7,000 or less at the time benefits become distributable in accordance with Section 9.1, the Participant, or the Beneficiary, as the case may be, shall be required to receive a distribution as soon as administratively practicable of the balance payable in a single lump sum payment. The vested value of the Participant's Account for this purpose shall be determined with regard to that portion of the Account that is attributable to Rollover Contributions (and earnings allocable thereto) within the meaning of Sections 401(c), 403(a)(4), 403(b)(8), and 457(e)(16) of the Code.

If the vested value of the Participant's Account is \$7,000 or less but greater than \$1,000 at the time benefits become distributable in accordance with Section 9.1 and the Participant does not elect to have such distribution paid in the manner described in Section 9.7 or to receive the lump sum directly as described in Section 9.2, the Trustee will pay the distribution in a direct rollover to an individual retirement plan designated by the Committee.

9.4 Minimum Distribution Requirements

The requirements of this Section 9.4 shall take precedence over any inconsistent provisions of the Plan. All distributions required under this Section 9.4 shall be determined and made in accordance with the with the provisions of Section 401(a)(9) of the Code, including the minimum distribution incidental benefits requirement of Section 401(a)(9)(G) of the Code and any applicable Regulations issued thereunder. Effective as of January 1, 2024, the provisions of this Section shall not be applicable to amounts credited to a Participant's Roth Contributions Account.

(a) <u>Time and Manner of Distribution.</u>

- (i) The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's 'Required Beginning Date.' The Required Beginning Date, for purposes of this Section 9.4, means April 1 of the calendar year following the later of (A) the calendar year in which the Participant attains his or her Required Beginning Date Age, or (B) the calendar year in which the Participant retires. A Participant who is a 5% owner as defined in Section 416 of the Code with respect to the Plan Year ending in the calendar year in which the Participant attains his or her Required Beginning Date Age shall have his or her Required Beginning Date determined under (a)(i)(A) above.
- (ii) If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, to the Participant's Beneficiary no later than as follows:
 - (A) If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, then distributions to the surviving Spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained his or her Required Beginning Date Age, if later.
 - (B) If the Participant's surviving Spouse is not the Participant's sole Designated Beneficiary, then distributions to the Designated Beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died.
 - (C) If there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.

(D) If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary and the surviving Spouse dies after the Participant but before distributions to the surviving Spouse begin, this Section 9.4(a)(ii), other than Section 9.4(a)(ii)(A), will apply as if the Surviving Spouse was the Participant.

For purposes of this Section 9.4(a)(ii), and 9.4(c), unless Section 9.4(a)(ii)(D) applies, distributions are considered to begin on the Participant's required beginning date. If Section 9.4(a)(ii)(D) applies distributions are considered to begin on the date distributions are required to begin to the surviving Spouse under Section 9.4(a)(ii)(A).

(b) Required Minimum Distributions During Participant's Lifetime.

- (i) During the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:
 - (A) the quotient obtained by dividing the Participant's account balance (as defined below in Section 9.4(d)(iii)) by the distribution period in the Uniform Lifetime Table set forth in Regulation Section 1.401(a)(9)-9 of the Regulations, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or
 - (B) if the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's Spouse, the quotient obtained by dividing the Participant's account balance by the number in the Joint and Last Survivor Table set forth in Regulation Section 1.401(a)(9)-9 of the Regulations, using the Participant's and Spouse's attained ages as of the Participant's and Spouse's birthdays in the Distribution Calendar Year.
- (ii) Required minimum distributions will be determined under this Section 9.4(b) beginning with the first Distribution Calendar Year and up to and including the Distribution Calendar Year that includes the Participant's date of death.
- (c) <u>Required Minimum Distributions After Participant's Death.</u>
 - (i) If the Participant dies on or after the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participants account balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's Designated Beneficiary, determined as follows:
 - (A) The Participant's remaining life expectancy is calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.

- (B) If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, the remaining life expectancy of the surviving Spouse is calculated for each Distribution Calendar Year after the year of the Participant's death using the surviving Spouse's age as of the Spouse's birthday in that year. For Distribution Calendar Years of after the year of the surviving Spouse's death, the remaining life expectancy of the surviving Spouse is calculated using the age of the surviving Spouse as of the Spouse' birthday in the calendar year of the Spouse's death reduced by one for each subsequent calendar year.
- (C) If the Participant's surviving Spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining life expectancy is calculated using the age of the Designated Beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
- (ii) If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's account balance by the Participant's remaining life expectancy calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- (d) Definitions.

The following definitions apply for purposes of this Section 9.4:

- (i) Designated Beneficiary: The individual who is designated as the beneficiary under Section 3.4 of the Plan and is the designated beneficiary under Section 401(a)(9) of the Code and Regulation Section 1.401(a)(9)-1, Q&A-4.
- (ii) Distribution Calendar Year: A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year which contains the Participant's Required Beginning Date.

For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin under Section 9.4(a)(ii). The required minimum distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Beginning Date. The required minimum distribution for other Distribution Calendar Years, including the required minimum distribution for the Distribution Calendar Year in which the Participant's Required Beginning Date occurs, will be made on or before December 31 of that Distribution Calendar Year.

(iii) Participant's account balance: The account balance as of the last valuation date in the calendar year immediately preceding the Distribution Calendar Year (valuation calendar year) increased by the amount of any contributions made and allocated or forfeitures allocated to the account balance as of dates in the valuation calendar year after the valuation date and decreased by distributions made in the valuation calendar year after the valuation date. The account balance for the valuation calendar year includes any amounts rolled over or transferred to the plan either in the valuation calendar year or in the Distribution Calendar Year if distributed or transferred in the valuation calendar year.

9.5 Special Timing Rules

Unless a Participant elects otherwise, his or her vested Account shall be distributed to him or her no later than 60 days after the close of the Plan Year in which occurs the latest of his or her Normal Retirement Age, the 10th anniversary of the year in which he or she commenced participation in the Plan or the date of his or her Severance from Employment. A Participant must, however, file a claim for benefits before benefits will be distributed to him or her in accordance with this Section 9.5.

The failure of a Participant to consent to a distribution while his or her benefit is immediately distributable within the meaning of Section 411(a)(11) of the Code shall be deemed to be an election to defer commencement of payment of any benefit sufficient to satisfy this Section 9.5.

9.6 Proof of Death

The Committee may require and rely upon such proof of death and such evidence of the right of any Beneficiary or other person to receive the value of the Accounts of a deceased Participant as the Committee may deem proper, and its determination of death and of the right of that Beneficiary or other person to receive payment shall be conclusive.

9.7 Direct Rollovers

(a) <u>In General.</u> Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Distributee's election under this Section 9.7, a Distributee may elect, at the time and in the manner prescribed by the Committee, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Distributee in a Direct Rollover. (b) Eligible Rollover Distribution. An Eligible Rollover Distribution is any distribution of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution does not include: any distribution that is one of a series of substantially equal periodic payments (no less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's designated beneficiary, or for a specified period of 10 years of more; any distribution to the extent such distribution is required under Section 401(a)(9) of the Code; except as otherwise provided below, the portion of any distribution that is not includible in gross income; and, any hardship distribution shall not fail to be an Eligible Rollover Distribution merely because the portion consists of amounts which are not includible in gross income.

However, such portion may be transferred only to an individual retirement account or annuity described in Section 408(a) or (b) of the Code, to a defined contribution plan described in Section 401(a) or 403(a) of the Code, or to an annuity contract described in Section 403(b) of the Code that agrees to separately account for amounts so transferred, including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not includible.

- (c) Eligible Retirement Plan. An Eligible Retirement Plan is an individual retirement account described in Section 408(a) of the Code, an individual retirement annuity described in Section 408(b) of the Code, a simple retirement account described in Section 408(p) of the Code, an annuity plan described in Section 403(a) of the Code, an annuity contract described in Section 403(b) of the Code, an eligible plan under Section 457(b) of the Code which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state and which agrees to separately account for amounts transferred into such plan for the Plan or a qualified trust described in Section 401(a) of the Code, that accepts the Distributee's Eligible Rollover Distribution. An individual retirement plan described in Section 408A of the Code shall be an Eligible Retirement Plan. In the case of a Distributee who is a nonspousal beneficiary of a deceased Participant, an individual retirement plan described in Section 402(c)(8)(B)(i) or (ii) of the Code that is established for the purpose of receiving the distribution on behalf of the nonspousal beneficiary shall be an Eligible Retirement Plan.
- (d) <u>Distributee.</u> A Distributee includes an Employee or former Employee. In addition, the Employee's or former Employee's surviving Spouse and the Employee's or former Employee's Spouse or former Spouse who is the alternate payee under a qualified domestic relations order, as defined in Section 414(p) of the Code, are Distributees with regard to the interest of the Spouse or former Spouse. A Distributee includes an individual who is both: (a) a designated beneficiary as defined in Section 401(a)(9)(E) of the Code of a deceased Participant and (b) not the surviving spouse of the Participant.
- (e) <u>Direct Rollover.</u> A Direct Rollover is a payment by the Plan to the Eligible Retirement Plan specified by the Distributee.

- (f) <u>Waiver of 30 Day Notice.</u> If a distribution is not subject to Sections 401(a)(11) and 417 of the Code, the distribution may be made less than 30 days after the notice required by Regulation Section 1.411(a)-11(c) is given provided:
 - (i) the Committee clearly informs the Participant that the Participant has the right to a period of at least 30 days after receiving the notice to consider the decision whether or not to elect a distribution (and, if applicable, a particular distribution option); and
 - (ii) the Participant, after receiving the notice, affirmatively, elects to receive a distribution.
- (g) <u>Roth Rollover.</u> A Direct Rollover of a distribution from a Participant's Roth Account, Roth Rollover Contribution Account, or Roth In-Plan Conversion Account shall only be made to another Roth Account of an applicable retirement plan described in Section 402A(e)(1) of the Code or to a Roth Individual Retirement Account described in Section 408A of the Code, and only to the extent the rollover is permitted under the rules of Section 402(c) of the Code.

9.8 Ordering Rules for Distributions

The Committee may operationally implement an ordering rule procedure for distributions made in accordance with this Article IX and withdrawals made in accordance with Article VIII from a Participant's Account attributable to Deferral Contributions, Catch-Up Contributions, Roth Contributions, and Roth Catch-Up Contributions. Such ordering rules may specify which type of contributions shall be distributed first. Furthermore, such procedure may permit the Participant to elect which type of contributions shall be distributed first.

Article X - Top Heavy Provisions

10.1 When Applicable

If the Plan is determined to be "Top Heavy" for any Plan Year, the provisions of this Article shall supersede any conflicting provisions of the Plan.

10.2 Top Heavy Determination

(a) The Plan shall be Top Heavy with respect to any Plan Year in which, as of the "Determination Date", the ratio of the present value of accrued benefits under all defined benefit plans in the "Aggregation Group" for "Key Employees" plus all Account balances attributable to Employer and Employee contributions (except as otherwise noted below) under the Plan and all other defined contribution plans in the Aggregation Group, exceeds 60% of such present value of accrued benefits and such Account balances for all Key Employees and Non-Key Employees under all plans in the Aggregation Group. The present values of accrued benefits and the amounts of account balances of an Employee as of the determination date shall be increased by the distributions made with respect to the Employee under the Plan and any plan aggregated with the Plan under Section 416(g)(2) of the Code during the one year period ending on the determination date. The preceding sentence shall also apply to distributions under a terminated plan which, had it not terminated, would have been aggregated with the Plan under Section 416(g)(2)(A)(i) of the Code. In the case of a distribution for a reason other than severance from employment, death or disability, this provision shall be applied by substituting a five year period for the one year period. The accrued benefits and accounts of any individual who has not performed services for the Employer during the one year period ending on the determination date shall not be taken into account.

The accrued benefits and account balances of any individual who is not a Key Employee but who was a Key Employee in a prior year will be disregarded. In any event, the calculation of the Top Heavy ratio and the extent to which distributions, tax deductible qualified employee contributions, rollovers and transfers are taken into account shall be in accordance with Section 416 of the Code and the Regulations thereunder. When aggregating plans, accrued benefits and account balances under other plans will be calculated as of determination dates that are within the same calendar year.

- (b) In determining the cumulative accrued benefits under a defined benefit plan for purposes of this Section 10.2, the actuarial assumptions specified by the defined benefit plan for this purpose shall be utilized. If differing actuarial assumptions are specified for two or more defined benefit plans, the actuarial assumptions for the defined benefit plan including the largest number of employees in the first year any defined benefit plan is included within the Aggregation Group shall be utilized. Solely for the purpose of determining if the Plan or any other plan in a required aggregation group of which the Plan is a part is a Top Heavy Plan, the accrued benefit of an Employee other than a Key Employee shall be determined (a) under the method, if any, that uniformly applies for accrual purposes under all plans maintained by the Employer, or (b) if there is no such method, as if such benefit accrued not more rapidly than the slowest accrual rate permitted under the fractional accrual rate of Section 411(b)(1)(C) of the Code.
- (c) The term Top Heavy shall not include a plan which consists solely of (a) a cash or deferred arrangement which meets the requirements of Section 401(k)(12) of the Code and (b) matching contributions with respect to which the requirements of Section 401(m)(11) of the Code are met. If, but for the foregoing, a plan would be treated as a Top Heavy plan because it is a member of an aggregation group which is a Top Heavy group, contributions under the plan may be taken into account in determining whether any other plan in the group meets the requirements of Section 416(c)(2) of the Code.

10.3 Minimum Contribution

For each year the Plan is Top Heavy the Employer shall contribute to the Plan and allocate to the Matching Contributions Account of each Participant who is a Non-Key Employee (including such an individual who is eligible to participate but has not elected to do so in accordance with Article III) an amount that is not less in total than the lesser of 3% of the Non-Key Employee's Compensation for the Plan Year or the greatest amount (expressed as a percentage of Compensation) allocated to the Account of any Key Employee for that year. Compensation for purposes of this Article X shall mean compensation within the meaning of Section 415(c)(3) of the Code. This minimum allocation shall be made even though, under other Plan provisions, the Eligible Employee would not otherwise be entitled to receive an allocation or would have received a lesser allocation for the year because of (a) his or her failure to be employed on a specified date such as the last day of the Plan Year, (b) his or her failure to make mandatory contributions, if any, to the Plan, or (c) his or her Compensation being less than a stated amount.

This requirement shall not apply to the extent the Participant is covered under any other plan or plans of the Employer and such Employer has provided that the minimum benefit or minimum allocation requirements applicable to Top Heavy Plans will be satisfied in the other plan or plans. Matching Contributions described in Article IV shall be taken into account for purposes of satisfying the minimum allocation requirements of Section 416(c)(2) of the Code and the Plan. Matching Contributions that are used to satisfy the minimum allocation requirements shall be treated as Matching Contributions of purposes of the actual contribution percentage test and other requirements of Section 401(m) of the Code.

10.4 Vesting Rules

For any Plan Year in which the Plan is Top-Heavy, the vesting schedule set forth in Section 7.2 shall apply.

10.5 Dual Plan Special Limitations

If a Key Employee participates in both the Plan and a defined benefit plan maintained by the Employer, then for all years that the Plan and the defined benefit plan are Top Heavy and the Top Heavy ratio referred to in Section 10.6(b) does not exceed 90%, the minimum benefit described in Section 416(h)(2)(A) of the Code shall be provided under the defined benefit plan for each Non-Key Employee.

10.6 Aggregation Groups

- (a) A Required Aggregation Group is defined as (a) each qualified plan of the Aggregated Employer in which at least one Key Employee participates or participated during the Determination Period (regardless of whether the plan has terminated) and (b) any other qualified plan of the Aggregated Employer which enables a plan described in (a) to meet the requirements of Section 401(a)(4) or 410(b) of the Code. The Determination Period, for this purpose, is the Plan Year containing the Determination Date or any of the four preceding Plan Years. A Permissive Aggregation Group is defined as a Required Aggregation Group plus any other plan or plans of the Employer which, when considered as a group with the Required Aggregation Group, would continue to satisfy the requirements of Sections 401(a)(4) and 410(b) of the Code.
- (b) Each plan of the Employer required to be included in an Aggregation Group shall be treated as a Top Heavy Plan if such group is a Top Heavy group. A required aggregation will be considered a Top Heavy group if the present value of the cumulative accrued benefits for Key Employees under all defined benefit plans included in such group and the aggregate of the accounts of Key Employees under all defined contribution plans included in such group exceed 60% of a similar sum determined for all Employees.

10.7 Key Employee Defined

- (a) A Key Employee means any Employee or former Employee (including any deceased Employee) who at any time during the Plan Year that includes the determination date was an officer of the Employer having annual Compensation greater than \$220,000 (as adjusted under Section 416(i)(1) of the Code), a 5% owner of the Employer, or a 1% owner of the Employer having annual compensation of more than \$150,000. For this purpose, annual Compensation means compensation within the meaning of Section 415(c)(3) of the Code. The determination of who is a Key Employee will be made in accordance with Section 416(i)(1) of the Code and the applicable Regulations of general applicability issued thereunder.
- (b) A Non-Key Employee is an Employee who is not a Key Employee.

10.8 Determination Date

Determination Date means with respect to the initial Plan Year, the last day of the first Plan Year and, for each other Plan Year, the last day of the preceding Plan Year.

Article XI - Administration of Plan

11.1 Committee

The Committee shall act as the Plan administrator and be responsible for the day-to-day operations of the Plan. The Committee shall be appointed by the Board of Directors and shall consist of one or more persons. The Board of Directors may remove any member of the Committee at any time with or without cause. The Board of Directors will fill vacancies in the Committee as soon as it is reasonably possible after the vacancy occurs. Until a new appointment is made, the remaining member or members of the Committee shall have full authority to act. The Board of Directors is responsible for transmitting to the Trustee the names and authorized signatures of the Committee, the name and signature of new members. Any member of the Committee may resign by delivering his or her written resignation to the Board of Directors, the Trustee and Committee. Such resignation shall become effective upon its receipt by the Board of Directors or on such other date as is agreed to by the Board of Directors and the resigning member.

11.2 Powers and Duties

The Committee shall keep a record of all its proceedings and acts with respect to its administration of the Plan and shall maintain all such books of accounts, records and other data as may be necessary for the proper administration of the Plan. The Committee shall have the maximum discretionary authority to interpret the provisions of the Plan. Any action taken by the Committee which is authorized, permitted, or required by the Plan, is final and binding on the Committee, all persons who have or claim an interest under the Plan and all third parties dealing with the Committee. The Committee shall notify the Trustee of any action taken by the Committee affecting the Trustee and its obligations or rights regarding the Plan and, when required, shall notify any other interested person or persons.

The Committee has the following powers and duties:

- (a) To determine the rights of eligibility of an Employee to participate in the Plan, the value of a Participant's Account and the nonforfeitable percentage of each Participant's Account;
- (b) To adopt rules of procedure and regulations necessary for the proper and efficient administration of the Plan provided the rules are not inconsistent with the terms of the Plan;
- (c) To establish and modify the method of accounting for the Plan or the Trust;
- (d) To construe and enforce the terms of the Plan and the rules and regulations it adopts, including interpretation of the Plan documents and documents related to the Plan's operation. If the terms of the Plan are unclear, the Committee may interpret the Plan, provided such interpretation is consistent with the provisions of Section 401(a) of the Code and is performed in a uniform and nondiscriminatory manner;

- (e) To direct the Trustee as respects the crediting and distribution of the Trust Fund;
- (f) To review and render decisions respecting a claim for (or denial of a claim for) a benefit under the Plan;
- (g) To furnish the Employer with information which the Employer may require for tax or other purposes;
- (h) To engage the service of agents whom it may deem advisable to assist it with the performance of its duties, including counsel, accountants and other consultants to aid it in exercising its powers and carrying out its duties;
- (i) To appoint, at the direction of the Company, an Investment Manager (as defined in Section 3(38) of ERISA), who shall have responsibility for investment of the Trust Fund;
- (j) To establish, in its sole discretion, a nondiscriminatory policy which the Trustee must observe in making loans, if any, to Participants;
- (k) To establish procedures, in its sole discretion, to assess each Participant's and Beneficiaries' Account, a per capita annual administrative fee, to be assessed to each Participant's Account at the discretion of the Committee;
- To allocate, in its sole discretion, amongst eligible Participants and Beneficiaries, no more frequently than once per calendar quarter, any Revenue Credits as defined in the Service Agreement between the Company and the Committee;
- (m) To delegate administrative authority and to establish rules of procedure for such Committee, including rules regarding how such Committee is to act, the vote required for action by the Committee and other procedures for the operation of the Committee as deemed appropriate by the Committee; and
- (n) To perform any other acts necessary and proper for the administration of the Plan, except for such acts that are to be performed by the Company, the Fund Manager or the Trustee.

All rules, procedures and decisions of the Committee shall be uniformly and consistently applied to all Participants in similar circumstances. Such rules, procedures and decisions so made shall be conclusive and binding on all persons having an interest in the Plan.

The Committee shall be a named fiduciary as defined in ERISA Section 402(a)(2). The Committee may, however, establish procedures for (a) the allocation of fiduciary responsibilities (other than "trustee responsibilities" as defined in Section 405(c) of ERISA) under the Plan among its members, and (b) the designation of persons other than named fiduciaries to carry out fiduciary responsibilities (other than trustee responsibilities) under the Plan. If any fiduciary responsibility is allocated or delegated to any person, no named fiduciary shall be liable for any act or omission of such person, except as provided in Section 405(c) of ERISA.

The Committee and any representative that the Committee chooses to assist it to carry out its responsibilities under the Plan shall have the maximum discretionary authority permitted by the law to interpret, construe and administer the Plan, to make determinations regarding Plan participation, enrollment and eligibility for benefits, to evaluate and determine the validity of benefit claims, and to resolve any and all claims and disputes regarding the rights and entitlements of individuals to participate in the Plan and to receive benefits and payments pursuant to the Plan. The decisions of the Committee and its representatives shall be given the maximum deference permitted by law.

11.3 Compensation and Expenses

The Company shall have discretion to pay or reimburse (or to direct an Employer to pay or reimburse) any reasonable costs and expenses of the Plan, including but not limited to any operational and administrative expenses of the Plan as well as compensation of fiduciaries and costs and expenses incurred by the Committee as a result of the performance of its duties and responsibilities hereunder or as may be required to operate and administer the Plan. Operational and administrative costs may include, but are not limited to legal fees, accounting fees, consulting fees, Employee and Participant communication fees and third-party administrator fees. Costs and expenses of the Committee may include, but are not limited to fees to accountants, consultants, counsel, specialists, and other persons employed or appointed to operate and administer the Plan or to help the Committee or Employers perform their duties and responsibilities hereunder. Any reasonable costs and expenses, including any operational and administrative costs and expenses, of the Plan not so paid or reimbursed shall be paid from the assets of the Plan. Any person who receives full-time pay for other services to an Employer shall not be entitled to any compensation for services to the Plan. The members of the Committee shall not receive compensation with respect to their services for the Committee but may be reimbursed for their expenses properly and actually incurred as members of the Committee.

11.4 Company Approval of Equity Builder Contributions

Any Matching Contribution formula and the amount of any Equity Builder Contribution shall be subject to the approval of the Board of Directors or its delegate.

11.5 Claims Procedure

- (a) <u>Definitions.</u> For purposes of this Section 11.5, the following words or phrases in quotes when capitalized will have the meaning set forth below:
 - (i) "Adverse Benefit Determination" means a denial, reduction, or the termination of, or a failure to provide or make payment (in whole or in part) with respect to a Claim for a benefit, including any such denial, reduction, termination, or failure to provide or make payment that is based on a determination of a Participant's or Beneficiary's eligibility to participate in the Plan.

- (ii) "Claim" means a request for a benefit or eligibility to participate in the Plan, made by a Claimant in accordance with the Plan's procedures for filing Claims, as described in this Section 11.5.
- (iii) "Claimant" is defined in Section 11.5(b)(ii).
- (iv) "Claims Administrator" means the Committee, or such other party designated by the Committee to review Claims.
- (v) "Notice" or "Notification" means the delivery or furnishing of information to an individual in a manner that would satisfies applicable Department of Labor regulations with respect to material required to be furnished or made available to an individual.
- (vi) "Relevant Documents" include documents, records, or other information with respect to a Claim that:
 - (A) were relied upon by the Claims Administrator in making the benefit determination;
 - (B) were submitted to, considered by, or generated for, the Claims Administrator in the course of making the benefit determination, without regard to whether such documents, records or other information were relied upon by the Claims Administrator in making the benefit determination;
 - (C) demonstrate compliance with administrative processes and safeguards required in making the benefit determination; or
 - (D) constitute a statement of policy or guidance with respect to the Plan concerning the denied benefit for the Participant's circumstances, without regard to whether such advice was relied upon by the Claims Administrator in making the benefit determination.
- (b) <u>Procedure for Filing a Claim.</u> In order for a communication from a Claimant to constitute a valid Claim, it must satisfy the following paragraphs (i) and (ii) of this paragraph (b).
 - (i) Any Claim submitted by a Claimant must be in writing on the appropriate Claim form (or in such other manner acceptable to the Claims Administrator) and delivered, along with any supporting comments, documents, records, and other information, to the Claims Administrator in person, or by mail postage paid, to the address for the Claims Administrator provided in the Plan's Summary Plan Description.
 - (ii) Claims and appeals of denied Claims may be pursued by a Participant or an authorized representative of the Participant (each of whom will be referred to in this Section as a "Claimant"). However, the Claims Administrator may establish reasonable procedures for determining whether an individual has been authorized to act on behalf of a Participant.

- (c) <u>Initial Claim Review.</u> The initial Claim review will be conducted by the Claims Administrator, with or without the presence of the Claimant, as determined by the Claims Administrator in its discretion. The Claims Administrator will consider the applicable terms and provisions of the Plan and amendments to the Plan, information and evidence that is presented by the Claimant and any other information it deems relevant. In reviewing the Claim, the Claims Administrator will also consider and be consistent with prior determinations of Claims from other Claimants who were similarly situated and which have been processed through the Plan's claims and appeals procedures within the past 24 months.
- (d) Initial Benefit Determination.
 - (i) The Claims Administrator will notify the Claimant of the Claim Administrator's determination within a reasonable period of time, but in any event (except as described in paragraph (ii) below) within 90 days after receipt of the Claim by the Claims Administrator.
 - (ii) The Claims Administrator may extend the period for making the benefit determination by 90 days if it determines that such an extension is necessary due to matters beyond the control of the Plan and if it notifies the Claimant, prior to the expiration of the initial 90 day period, of circumstances requiring the extension of time and the date by which the Claims Administrator expects to render a decision.
- (e) Manner and Content of Notification of Adverse Benefit Determination.
 - (i) The Claims Administrator will provide a Claimant with written or electronic Notice of any Adverse Benefit Determination that would satisfy Department of Labor regulations.
 - (ii) The Notification will set forth in a manner calculated to be understood by the Claimant:
 - (A) The specific reason or reasons for the Adverse Benefit Determination;
 - (B) Reference to the specific provision(s) of the Plan on which the determination is based;
 - (C) Description of any additional material or information necessary for the Claimant to perfect the Claim and an explanation of why such material or information is necessary; and
 - (D) A description of the Plan's review procedures and the time limits applicable to such procedures, including a statement of the Claimant's right to bring a civil action under Section 502(a) of ERISA following an Adverse Benefit Determination or review.

(f) <u>Procedure for Filing a Review of an Adverse Benefit Determination.</u>

(i) Any appeal of an Adverse Benefit Determination by a Claimant must be brought to the Claims Administrator within 60 days after receipt of the Notice of the Adverse Benefit Determination. Failure to appeal within such 60-day period will be deemed to be a failure to exhaust all administrative remedies under the Plan.

The appeal must be in writing utilizing the appropriate form provided by the Claims Administrator (or in such other manner acceptable to the Claims Administrator); provided, however, that if the Claims Administrator does not provide the appropriate form, no particular form is required to be utilized by the Participant. The appeal must be filed with the Claims Administrator at the address listed in the Plans' Summary Plan Description.

- (ii) A Claimant will have the opportunity to submit written comments, documents, records, and other information relating to the Claim.
- (g) <u>Review Procedures for Adverse Benefit Determinations.</u>
 - (i) The Claims Administrator will provide a review that takes into account all comments, documents, records, and other information submitted by the Claimant without regard to whether such information was submitted or considered in the initial benefit determination.
 - (ii) The review procedure may not require more than two levels of appeals of an Adverse Benefit Determination.
- (h) Timing and Notification of Benefit Determination on Review. The Committee will make a determination on the appeal of the Adverse Benefit Determination no later than the date of the meeting of the Committee that immediately follows the receipt of the appeal of the Adverse Benefit Determination, unless the appeal is filed within 30 days preceding the date of such meeting. In such case, a benefit determination will be made by no later than the date of the second meeting of the Committee following the receipt of the appeal of the Adverse Benefit Determination. If the Committee determines that special circumstances require an extension of time for processing the appeal of the Adverse Benefit Determination, a benefit determination shall be rendered not later than the third meeting of the Committee following receipt of the appeal of the Adverse Benefit Determination. If the Committee determines that an extension is required, the Committee shall provide the claimant with written Notice of the extension, describing the special circumstances and the date as of which the benefit determination will be made, prior to the commencement of the extension. The Committee shall notify the claimant, in accordance with (i) of this Section 11.5 as soon as possible, but not later than five days after the benefit determination is made.

If such an extension is necessary due to a failure of the Claimant to submit the information necessary to decide the Claim, the period in which the Committee is required to make a decision will be tolled from the date on which the notification is sent to the Claimant until the Claimant adequately responds to the request for additional information.

- (i) Manner and Content of Notification of Benefit Determination on Review.
 - (i) The Claims Administrator will provide a written or electronic Notice of the Plan's benefit determination on review, in accordance with applicable Department of Labor regulations.
 - (ii) The Notification will set forth:
 - (A) The specific reason or reasons for the Adverse Benefit Determination;
 - (B) Reference to the specific provision(s) of the Plan on which the determination is based;
 - (C) A statement that the Claimant is entitled to receive, upon request and free of charge, reasonable access to and copies of all Relevant Documents; and
 - (D) A statement of the Claimant's right to bring a civil action under Section 502(a) of ERISA following an Adverse Benefit Determination.
- (j) <u>Exhaustion of Administrative Remedies.</u> The claims procedures set forth in this Section 11.5 shall be strictly adhered to by each Participant or Beneficiary under the Plan and no judicial or arbitration proceedings with respect to any claim for Plan benefits hereunder shall be commenced by any such Participant or Beneficiary until the proceedings set forth herein have been exhausted in full.
- (k) <u>Statute of Limitations.</u> No cause of action may be brought by a claimant who has received an Adverse Benefit Determination later than two years following the date of such Adverse Benefit Determination.

11.6 Bonding of Fiduciaries

The Committee shall be responsible for seeing that every fiduciary of the Plan and every person who handles Plan assets shall be covered by a fidelity bond of similar policy of insurance to the extent required by Section 412 of ERISA. The costs of such coverage shall be paid out of the Plan assets, upon direction of the Committee, if the cost thereof shall not be timely paid by the Employer.

11.7 Standard of Conduct

The Committee and any other person to whom any fiduciary responsibility with respect to the Plan is allocated or delegated shall discharge his or her duties and responsibilities with respect to the Plan in accordance with the standards set forth in Section 404(a)(1) of ERISA, which provides, subject to the Sections 403(c) and (d), 4042, and 4044 of ERISA, that a fiduciary shall discharge his or her duties with respect to a plan solely in the interest of the participants and beneficiaries and

- (a) For the exclusive purpose of:
 - (i) Providing benefits to participants and their beneficiaries; and
 - (ii) Defraying reasonable expenses of administering the plan;
- (b) With the care, skill, prudence and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims;
- (c) By diversifying the investments of the plan so as to minimize the risk of large losses, unless under the circumstances it is clearly prudent not to do so; and
- (d) In accordance with the documents and instruments governing the plan insofar as such documents and instruments are consistent with the provisions of this title.

Article XII - Management of Funds

12.1 Appointment of Trustees

Subject to the provisions of Section 12.4, the Company shall appoint one or more Trustees to receive and hold in trust all contributions paid into the Trust Fund. Such Trustee or Trustees shall serve at the pleasure of the Board of Directors and shall have such rights, powers, and duties as the Board of Directors shall from time to time determine including but not limited to those stated below.

12.2 Investment of Trust Fund by Trustees

All contributions made to the Trust Fund pursuant to the Plan shall be paid to the Trustees and, except as herein otherwise provided, shall be held, invested, and reinvested by the Trustees without distinction between principal and income in such securities or such other property, real or personal, wherever situated, as the Trustees shall deem advisable, including, but not limited to, shares of stock, common or preferred, whether or not listed on any exchange, participations in mutual investment funds, bonds and mortgages, and other evidences of indebtedness or ownership, or in loans to Participants (consistent with other provisions hereof), and participations in any common trust fund established or maintained by the Trustees for the collective investment of fiduciary funds and shall not be limited by any state statute or judicial decision prescribing or limiting investments appropriate for trustees. The Trustees shall hold and retain all the property and assets of the Trust Fund including income from investments and from all other sources, for the exclusive benefit of the Participants and their Beneficiaries, as provided herein, and for paying the costs and expenses of administering the Plan or Trust Fund, to the extent that the same are not paid by any Employer. Reasonable expenses attendant to qualified domestic relations order determinations shall be allocated to the Account of the Participant or Beneficiary seeking the determination.

12.3 Investment of Trust Fund by Investment Manager

The Committee may enter into one or more agreements for the appointment of one or more Investment Managers to supervise and direct all the investment and reinvestment of a portion or all of the Trust Fund in accordance with the provisions of the Plan in the same manner and with the same powers, duties, obligations, responsibilities, and limitations as apply to the Trustees. As a condition to its appointment, an Investment Manager shall acknowledge in writing that it is a fiduciary with respect to the Trust Fund.

An Investment Manager so appointed shall be an investment advisor registered under the Investment Advisors Act of 1940, a bank as defined in such Act or an insurance company that is qualified to manage the assets of employee benefit plans pursuant to the laws of more than one state. The Trustees shall be bound by the supervision and direction of the Investment Manager, unless and until the Company amends or revokes the appointment or authority of the Investment Manager.

The Company may furnish an Investment Manager with written investment guidelines for investment of the Trust Fund assets, which guidelines may include directions with respect to diversification of the investments. Any Investment Manager shall receive such reasonable compensation chargeable against the Trust Fund or payable by each Employer as shall be agreed upon by the Company. The Company may revoke any agreement with the Investment Manager at any time by 30 days' written notice to the Investment Manager. Any Investment Manager may resign by 30 days' written notice to the Company.

12.4 Exclusive Benefit Rule

Except as otherwise provided in the Plan, no part of the corpus or income of the assets of the Plan shall be used for, or diverted to, purposes other than for the exclusive benefit of Participants and other persons entitled to benefits under the Plan. No person shall have any interest in or right to any part of the earnings of the assets of the Plan, or any right in, or to, any part of the assets held under the Plan, except as and to the extent expressly provided in the Plan.

Article XIII - Amendment, Merger, Termination of Plan

13.1 Amendment of Plan

Subject to the provisions of Section 13.4, HEICO Corporation, by action of its Board of Directors, or by the Committee to the extent that such power is delegated by the Board of Directors to the Committee, shall have the right at any time to amend the Plan, and retroactively if deemed necessary or appropriate, except that no such amendment shall make it possible for any part of the assets of the Plan to be used for, or diverted to, purposes other than for the exclusive benefit of persons entitled to benefits under the Plan. No amendment shall be made which has the effect of decreasing the balance of the Accounts of any Participant or of reducing the nonforfeitable percentage computed under the Plan as in effect on the date on which the amendment is adopted or, if later, the date on which the amendment becomes effective.

All such amendments shall be as set forth in an instrument in writing executed by an officer of the Company if adopted by the Company, or by a designated member of the Committee if adopted by the Committee. Any amendment may be current, retroactive, or prospective, in each case as provided therein.

13.2 Merger or Consolidation

The Plan may not be merged or consolidated with, and its assets or liabilities may not be transferred to, any other plan unless each person entitled to benefits under the Plan would, if the resulting plan were then terminated, receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit he or she would have been entitled to receive immediately before the merger, consolidation, or transfer if the Plan had then terminated.

13.3 Participating Employers

- (a) <u>Additional Participating Employers.</u> If any organization which is an Affiliate of HEICO Corporation, such organization may become an Employer upon appropriate resolutions by that organization's board of directors, other governing body, or its delegate. Upon receipt and acceptance of the resolutions, the Committee or its delegate shall approve adoption of the Plan by the Affiliate.
- (b) <u>Delegation of Authority.</u> Adoption of the Plan by an Affiliate in accordance with this Section shall constitute the delegation by the organization to the Board of Directors, the Trustees, and the Committee as provided in Article XIII, of full authority to amend the Plan. Such adoption shall constitute the Employer's delegation to the Board of Directors of the authority to appoint the Committee as Plan Administrator and on behalf of the Affiliate to enter into any contracts related to this Plan.

- (c) <u>Duties of Participating Employers Generally.</u> Each Participating Employer shall supply full and timely information to the Committee, or its delegate, on all matters relating to its Employees' Compensation, employment, retirement, Disability, transfer, death, or other termination of their status as Participants or Employees and such additional pertinent facts as are necessary to enable the Committee to perform its functions. Each Participating Employer shall have the sole responsibility for making the contributions, if any, provided for in Article IV with respect to Participants who are its Employees, subject to the direction of the Committee.
- (d) <u>Right to Withdraw.</u> Each Employer reserves to itself the right to terminate its status as an Employer under the Plan and withdraw from the Plan as provided in this Section. The Committee, or its delegate, under guidelines established by the Company, shall determine whether or not to consent to the withdrawal of an Employer. In the event of withdrawal of participation in the Plan by an Employer, the funds held for the benefit of the Participants or Beneficiaries of the withdrawing Employer shall be segregated. That portion of the fund applicable to any Employer which has not been withdrawn shall be unaffected. Notwithstanding the foregoing, in the event of withdrawal from participation in the Plan by an Employer, if the Employer and the Committee agree, the assets so segregated shall be transferred to a new Trustee as specified by the withdrawing Employer, subject to the approval of the Committee.
- (e) <u>Withdrawal Directed By Committee.</u> The Committee reserves to itself the right to terminate an Employer's status under the Plan if such withdrawal is necessary or appropriate to maintain compliance with the applicable provisions of Section 401(a) of the Code and require that such Employer withdraw from the Plan or to discontinue contributions to the Plan with respect to its Employees. In the event of withdrawal of participation in the Plan by an Employer, the funds held for the benefit of the Participants or Beneficiaries of the withdrawing Employer shall be segregated. That portion of the fund applicable to any Employer which has not been withdrawal from participation in the Plan by an Employer, if the Employer and the Committee agree, the assets so segregated shall be transferred to a new Trustee as specified by the withdrawing Employer, subject to the approval of the Committee.

13.4 Termination of Plan

- (a) The Board of Directors, or its delegate, may terminate the Plan or completely discontinue contributions under the Plan for any reason at any time. In the case of the termination or partial termination of the Plan, or of the complete discontinuance of Employer contributions to the Plan, affected Participants shall be 100% vested in and have a nonforfeitable right to the total amount in all of their Accounts under the Plan as of the date of the termination or discontinuance. The total amount in each Participant's Account shall be distributed, as the Committee shall direct, to him or her or for his or her benefit or continued in trust for his or her benefit.
- (b) The Plan will be deemed terminated (a) if and when the Company is judicially declared bankrupt or executes a general assignment to or for the benefit of its creditors, (b) if and when the Company is a party to a merger in which it is not the surviving organization unless the surviving organization adopts the Plan within 60 days after the merger, or (c) upon dissolution of the Company.

Article XIV - Miscellaneous Provisions

14.1 Limitation of Liability

Neither the Company, any Employer, the Committee, or any of their respective directors, members, officers, and employees, shall incur any liability for any act or failure to act unless such act or failure to act constitutes willful misconduct or gross negligence in relation to the Plan.

14.2 Indemnification

The Company indemnifies and saves harmless the Committee from and against any and all loss resulting from liability to which the Committee may be subjected by reason of any act or conduct (except willful misconduct or gross negligence) in the Committee's official capacity in the administration of the Plan, including all expenses reasonably incurred in the Committee's defense, in case the Employer fails to provide such defense. The indemnification provisions of this Section 14.2 do not relieve the Committee from any liability under ERISA for breach of a fiduciary duty. Furthermore, the Committee and the Employer may execute a letter agreement further delineating the indemnification agreement of this Section 14.2, provided the letter agreement is consistent with and does not violate ERISA.

The Plan may purchase insurance for its fiduciaries or for itself to cover liability or losses occurring by reason of the act or omission of a fiduciary, if such insurance permits recourse by the insurer against the fiduciary in the case of a breach of a fiduciary obligation by such fiduciary. A fiduciary may purchase insurance to cover any potential fiduciary for his or her own account. An Employer or an employee organization may purchase insurance to cover potential liability of one or more persons who serve in a fiduciary capacity with regard to the Plan.

14.3 Compliance with ERISA

Anything herein to the contrary notwithstanding, nothing above or any other provision contained elsewhere in the Plan shall relieve a fiduciary or other person of any responsibility or liability for any responsibility, obligation, or duty imposed upon him or her pursuant to the Title I, Part 4 of ERISA. Furthermore, anything in the Plan to the contrary notwithstanding, if any provision of the Plan is voided by Sections 410 and 411 of ERISA, such provision shall be of no force and effort only to the extent that it is voided by such Section.

14.4 Nonalienation of Benefits

- (a) None of the payments, benefits, or rights of any Participant shall be subject to any claim of any creditor of such Participant and, in particular, shall be free from attachment, garnishment, trustee's process, or any other legal or equitable process available to any creditor of such Participant. No Participant shall have the right to alienate, commute, pledge, encumber or assign any of the benefits or payments which he or she may expect to receive, contingently or otherwise, under the Plan, except the right to designate a Beneficiary or Beneficiaries as hereinbefore provided.
- (b) Section 14.4(a) also shall apply to the creation, assignment or recognition of a right to any benefit payable with respect to a Participant pursuant to a domestic relations order, unless such order is a qualified domestic relations order (QDRO) as defined in Section 414(p) of the Code. For this purpose, on or after August 17, 2007, a domestic relations order that otherwise satisfies the requirements of a QDRO as defined in Section 414(p) of the Code will not fail to be treated as a QDRO solely because the order is issued after, or revises another domestic relations order or QDRO or solely because of the time at which the order is issued, including issuance after the annuity starting date or after the Participant's death.

Any fees associated with the review, processing, and administration of a QDRO shall be charged against the Account of the affected Participant and the account of the affected Alternate Payee. The Plan's QDRO procedures are set forth in a separate document, which are incorporated herein as if its terms were fully set forth in this document.

(c) The Plan may offset against the Account of any Participant, any amount that the Participant is ordered or required to pay under a judgment, order, decree, or settlement described in ERISA Section 206(a)(4) and Section 401(a)(13)(C) of the Code.

14.5 Employment Not Guaranteed By Plan

Neither the establishment of the Plan nor its amendment nor the granting of a benefit pursuant to the Plan shall be construed as giving any Participant the right to continue as an Employee of an Employer, as limiting the rights of such Employer to dismiss or impose penalties upon the Participant or as modifying in any other way the terms of employment of any Participant.

14.6 Form of Communication

Any election, application, claim, notice, or other communication required or permitted to be made by or to a Participant, the Committee, the Company, or an Employer in writing shall be made in such form as the Committee, the Company, the Employer, or the Committee, as the case may be, shall prescribe.

Such communication shall be effective upon mailing if sent first class, postage prepaid and addressed to the addressee at its principal office, or to the Participant at his or her last known address, or upon personal delivery, if delivered to an officer of the addressee or to the Participant, as the case may be.

Notwithstanding anything in the Plan to the contrary, any notice, form or other communication hereunder shall be made in the manner prescribed by the Committee in accordance with applicable law, which may include, in appropriate circumstances, communication by telephone or by electronic or other means.

14.7 Facility of Payment

If a Participant or Beneficiary entitled to receive payments hereunder is unable to care for his or her affairs because of illness, accident, or disability, and a duly qualified guardian or legal representative is appointed for such Participant or Beneficiary, the Committee shall direct the Trustee to pay any amount to which the Participant or Beneficiary is entitled to such duly qualified guardian or legal representative upon claim of such guardian or legal representative.

If a duly qualified guardian or legal representative is not appointed for such Participant or Beneficiary, the Committee shall direct the Trustee to pay any amount to which the Participant or Beneficiary is entitled to such person's Spouse, child, grandchild, parent, brother or sister, or to a person deemed by the Committee to have incurred expense for such person entitled to payment. Any payment made pursuant to this Section 14.7 in good faith shall be a payment for the account of the Participant or Beneficiary and shall be a complete discharge from any liability of the Plan and the Committee.

14.8 Reduction for Overpayment

The Committee will, whenever it determines that a person has received a benefit payment under the Plan in excess of the amount to which the person is entitled under the terms of the Plan, make a reasonable attempt to collect such overpayment from the person. The amount of any overpayment may be set off against further amounts payable to or on account of the person who received the overpayment.

14.9 Unclaimed Benefits

If the Committee cannot ascertain the whereabouts of any person to whom a payment is due under the Plan, and if, after five years from the date such payment is due, a notice of such payment due is mailed to the last known address of such person, as shown on the records of the Employer and within three months after such mailing such person has not made written claim therefore, the Committee, if it so elects, may direct that such payment and all remaining payments otherwise due to such person be cancelled on the records of the Plan and the amount thereof applied in any manner permitted by Section 7.3, and upon such cancellation, the Plan shall have no further liability therefore except that, in the event such person later notifies the Committee of his or her whereabouts and requests the payment or payments due to him or her, the amount so applied shall be paid to him or her as provided in Article IX without adjustment for gains and losses.

14.10 Payments to Minors and Incompetents

If the Committee receives evidence satisfactory to it that any person entitled to receive any benefit under the Plan is, at the time when such benefit is payable, a minor, or physically or mentally incompetent to receive such benefit and to give a valid release therefore, and that another individual or institution is then maintaining or has custody of such person, and that no guardian or other representative of the estate of such person has been duly appointed, the Committee may authorize payment of such benefit otherwise payable to such person to such other individual or institution, and the release of such other individual or institution will be valid and a complete discharge for the purpose of such benefit.

14.11 Reliance on Information Provided to the Plan

Notwithstanding anything contained herein to the contrary, if an individual is provided a statement in confirmation of any election or information provided to the Plan by such individual hereunder, the election or information reflected on such confirmation statement will be deemed to be accurate and may be conclusively relied upon for all purposes hereunder unless the individual timely demonstrates to the Committee, in the form and manner established by the Committee, that the election or information reflected on the confirmation statement is not what the individual originally delivered to the Committee.

14.12 Service in More Than One Fiduciary Capacity

Any individual, entity, or group of persons may serve in more than one fiduciary capacity with respect to the Plan.

14.13 Binding Effect of Company's Actions

Each Employer shall be bound by any and all decisions and actions taken by the Company hereunder.

14.14 Military Service

Notwithstanding any other provision of the Plan to the contrary, service credit and contributions with respect to qualified Military Service will be provided in accordance with Section 414(u) of the Code.

14.15 Limitation of Rights

Neither the Employer nor the Committee represents or guarantees that the value of a Participant's Accounts shall at any time equal or exceed the amount previously contributed thereto. Neither the establishment of the Plan nor any modification thereof, nor the creating of any account, nor the payment of any benefits shall be construed as giving to any Participant or other person any legal or equitable right against the Employer except as provided in the Plan.

14.16 Limitation of Third-Party Rights

Nothing expressed or implied in the Plan is intended or will be construed to confer upon or give to any person, firm, or association other than the Employer, the Participants, or Beneficiaries, and their successors in interest, any right, remedy, or claim under or by reason of this Plan except pursuant to a QDRO.

14.17 Invalid Provisions

In case any provision of this Plan is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Plan. The Plan will be construed and enforced as if the illegal and invalid provisions had never been included.

14.18 One Plan

This Plan may be executed in any number of counterparts, each of which will be deemed an original and the counterparts will constitute one and the same instrument and may be sufficiently evidenced by any one counterpart.

14.19 Use and Form of Words

Whenever any words are used herein in the masculine gender, they will be construed as though they were also used in the feminine gender in all cases where that gender would apply, and vice versa. Whenever any words are used herein in the singular form, they will be construed as though they were also used in the plural form in all cases where the plural form would apply, and vice versa.

14.20 Headings

Headings to Articles and Sections are inserted solely for convenience and reference, and in the case of any conflict, the text, rather than the headings, shall control.

14.21 Authorization

Whenever the Employer under the terms of the Plan is permitted or required to do or perform any act or matter or thing, it shall be done and performed by any officer thereunto authorized by the Employer.

14.22 Governing Law

The provisions of this Plan, and the rights and obligations herein created, will be governed by, and construed according to the federal laws governing employee benefit plans under the Code and ERISA and to the extent not preempted by Federal law, shall be interpreted according to the substantive law of the State of Florida.

14.23 Protected Benefits

All benefits which are protected by the terms of Section 411(d)(6) of the Code and Section 204(g) of ERISA, which cannot be eliminated without adversely affecting the qualified status of the Plan on and after the Effective Date, will be provided under the Plan to Participants for whom such benefits are protected.

14.24 Receipt and Release

Subject to the provisions of ERISA and to the extent permitted by ERISA, any final payments or distribution to any Participant, his or her Beneficiary, or his or her legal representative in accordance with the Plan shall be in full satisfaction of all claims against the Committee and the Employer.

14.25 Forum and Venue

Any judicial action related to the provisions of this Plan, and the rights and obligations herein created shall be brought in a court, either federal or state, as may be applicable, located in Broward County, Florida.

IN WITNESS WHEREOF, and as evidence of the adoption of the Plan, the undersigned officer duly authorized has appended his or her signature this 25th day of March 2024.

HEICO Corporation

By: /s/ CARLOS L. MACAU, JR.

Title: Treasurer

Exhibit 1

Participating Employers

Participating Employer	EIN	Effective Date
HEICO Corporation	65-0341002	January 1, 1985
Jet Avion Corporation	59-2699611	July 1, 1985
LPI Industries Corporation	65-0054782	April 1, 1989
HEICO Aerospace Corporation	59-0791770	April 27, 1993
Aircraft Technology, Inc.	65-0233725	October 1, 1998
Radiant Power Corp.	65-0892651	February 1, 1999
Turbine Kinetics, Inc.	65-0845883	October 1, 1999
Thermal Structures, Inc.	95-3575611	August 1, 1999
Santa Barbara Infrared, Inc.	77-0111325	January 1, 2000
Northwings Accessories Corp.	65-0312802	January 1, 2000
Leader Tech, Inc.	04-2667972	April 1, 2000
Future Aviation, Inc.	65-1011336	April 1, 2001
Analog Modules, Inc.	59-2074349	May 1, 2001
Jetseal, Inc.	91-1433851	January 1, 2002
Inertial Airline Services, Inc.	34-1823836	January 1, 2002
HEICO Aerospace Parts Corp.	65-1146790	July 1, 2002
AeroDesign, Inc.	62-1858631	January 1, 2003
Niacc-Avitech Technologies Inc.	51-0453669	July 1, 2003
Sierra Microwave Technology, LLC	37-1480034	January 1, 2004
Connectronics Corp.	20-1971140	December 1, 2004
Lumina Power, Inc.	20-2350926	May 1, 2005
Engineering Design Team, Inc.	93-0964386	January 1, 2006
Prime Air, LLC	20-5545289	January 1, 2007
Sunshine Avionics LLC	26-1913893	July 1, 2008

Participating Employer	EIN	Effective Date
HEICO Parts Group, Inc.	26-3082967	January 1, 2009
VPT, Inc.	54-1684156	June 1, 2009
dB Control Corp.	27-1784894	July 1, 2010
3D Plus U.S.A., Inc.	74-2947071	January 1, 2012
Switchcraft, Inc.	36-2051512	January 1, 2012
Conxall Corporation	36-2944789	January 1, 2012
Ramona Research, Inc.	45-4786673	July 1, 2012
CSI Aerospace, Inc.	45-5531151	October 1, 2012
Action Research Company	46-1086859	January 1, 2013
Reinhold Industries, Inc.	13-2596288	September 1, 2013
Lucix Corporation	77-0504129	November 29, 2013
SI-Rel Inc.	46-3801015	January 1, 2014
Seal Dynamics LLC	20-3713644	January 1, 2015
Harter Aerospace, LLC	47-2458702	April 1, 2015
Thermal Energy Products, Inc.	33-0238469	October 1, 2015
Astroseal Products Mfg. Corporation	06-0808406	January 1, 2016
Midwest Microwave Solutions, Inc.	27-0124198	January 1, 2016
Robertson Fuel Systems, L.L.C.	86-0775241	April 1, 2016
Aerospace & Commercial Technologies, LLC	47-4759100	January 1, 2017
Air Cost Control US, LLC	20-5059785	July 1, 2017
Carbon By Design	20-1043190	January 1, 2018
AeroAntenna Technology, Inc.	95-4322670	January 1, 2018
Sensor Technology Engineering, LLC	82-4199356	July 1, 2018
HEICO Aerospace Parts Corp.	65-1146790	July 1, 2018
Optical Display Engineering, LLC	46-1634579	January 1, 2019

Participating Employer	EIN	Effective Date
Dielectric Sciences, Inc.	04-2603559	January 1, 2019
Blue Aerospace LLC	27-4313205	January 1, 2019
Specialty Silicone Products, Inc.	14-1725435	January 1, 2019
Apex Microtechnology, Inc.	46-0624696	April 1, 2019
Solid Sealing Technology, Inc.	20-0539521	May 1, 2019
Decavo, L.L.C.	26-4673544	January 1, 2020
Research Electronics International, LLC	62-1597816	January 1, 2020
IRCameras LLC	27-4686945	January 1, 2020
HEICO Repair Group Aerostructures, LLC	82-2700627	January 1, 2020
Rocky Mountain Hydrostatics, LLC	84-1168334	October 1, 2020
Quell Corporation	46-0507356	January 1, 2021
Intelligent Devices, LLC.	52-1943471	January 1, 2021
Transformational Security, LLC	20-1891706	January 1, 2021
Pyramid Semiconductor Corp.	86-2249124	July 1, 2021
Camtronics, LLC	86-3999996	October 1, 2021
Connect Tech Inc.	98-0187158	January 1, 2022
Ridge Engineering, LLC	52-0908121	January 1, 2022
R. H. Laboratories, Inc.	02-0527375	January 1, 2022
The Bechdon Company, LLC	52-0895348	January 1, 2022
Pioneer Industries, Inc.	11-1746409	July 1, 2022
Accurate Metal Machining, Inc.	34-1193166	October 1, 2022
Charter Engineering, Inc.	59-3400419	October 1, 2022
Sensor Systems, Inc.	95-2127229	October 1, 2022
Ironwood Electronics, Inc.	47-4956282	November 1, 2022
Exxelia USA, Inc.	45-2894705	April 1, 2023

Participating Employer	EIN	Effective Date
Micropen Technologies Corporation	16-1188497	April 1, 2023
Wencor Group, LLC	27-1732702	August 4, 2023

* As of January 1, 2024

RULE 13a-14(a)/15d-14(a) CERTIFICATION

I, Laurans A. Mendelson, certify that:

- (1) I have reviewed this Quarterly Report on Form 10-Q of HEICO Corporation;
- (2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- (3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- (4) The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- (5) The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 30, 2024

/s/ LAURANS A. MENDELSON

Laurans A. Mendelson Chief Executive Officer (Principal Executive Officer)

RULE 13a-14(a)/15d-14(a) CERTIFICATION

I, Carlos L. Macau, Jr., certify that:

- (1) I have reviewed this Quarterly Report on Form 10-Q of HEICO Corporation;
- (2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- (3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- (4) The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- (5) The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 30, 2024

/s/ CARLOS L. MACAU, JR.

Carlos L. Macau, Jr. Chief Financial Officer (Principal Financial Officer)

SECTION 1350 CERTIFICATION

In connection with the Quarterly Report of HEICO Corporation (the "Company") on Form 10-Q for the period ended April 30, 2024 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Laurans A. Mendelson, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 30, 2024

<u>/s/ LAURANS A. MENDELSON</u> Laurans A. Mendelson Chief Executive Officer (Principal Executive Officer)

SECTION 1350 CERTIFICATION

In connection with the Quarterly Report of HEICO Corporation (the "Company") on Form 10-Q for the period ended April 30, 2024 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Carlos L. Macau, Jr., certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 30, 2024

<u>/s/ CARLOS L. MACAU, JR.</u> Carlos L. Macau, Jr. Chief Financial Officer (Principal Financial Officer)